

### **University of Huddersfield Repository**

Gao, Lanlan

An Exploratory Study: Games Promoting the Development of Virtual Inquiry-Based Learning Community

#### **Original Citation**

Gao, Lanlan (2020) An Exploratory Study: Games Promoting the Development of Virtual Inquiry-Based Learning Community. Doctoral thesis, University of Huddersfield.

This version is available at http://eprints.hud.ac.uk/id/eprint/35558/

The University Repository is a digital collection of the research output of the University, available on Open Access. Copyright and Moral Rights for the items on this site are retained by the individual author and/or other copyright owners. Users may access full items free of charge; copies of full text items generally can be reproduced, displayed or performed and given to third parties in any format or medium for personal research or study, educational or not-for-profit purposes without prior permission or charge, provided:

- The authors, title and full bibliographic details is credited in any copy;
- A hyperlink and/or URL is included for the original metadata page; and
- The content is not changed in any way.

For more information, including our policy and submission procedure, please contact the Repository Team at: E.mailbox@hud.ac.uk.

http://eprints.hud.ac.uk/

# AN EXPLORATORY STUDY: GAMES PROMOTING THE DEVELOPMENT OF VIRTUAL INQUIRY-BASED LEARNING COMMUNITY

LANLAN GAO

A thesis submitted to the University of Huddersfield in partial fulfilment of the requirements for the degree of Doctor of Philosophy

The University of Huddersfield

September 2020

## **Copyright statement**

- The author of this thesis owns any copyright in the thesis (including any appendices, tables and/or graphs/ figures). And she has given the University of Huddersfield the right of such copyright for utilising in any educational and/or teaching, administrative and promotional purposes.
- ii. When people copies of this thesis in full or in extracts, they may make it only according to the regulations of the University Library that may be obtained details from the Librarian. This page must be listed as a part of any such copies used.

## Acknowledgements

If I do not have the support of many great people who always help me for many years, my doctoral research would not have been possible to complete. I need to deeply express my thanks to them.

First of all, I would like to express my deepest gratitude to my supervisors, Professor Rupert Ward and Professor Carlo Fabricatore. Thank Carlo for offering me the opportunity to research in the field of game-based learning in my study earlier. He guided me to have consciousness of doing academic research. In the final year of my study, thank Professor Rupert Ward for his support and help. When I faced many challenges and felt confused at that moment, he guided me to overcome these difficulties to finish my thesis smoothly. Their guidance, care, encouragement and valuable suggestions have a great help for my study life. And the two professors' diligent work and rigorous academic attitude positively impact my personal growth.

I would also like to express my sincere gratitude to my co-supervisor, Professor Maria Lopez-Campino, we were together to participate in many inspiring discussions and she gave many useful suggestions for my research project. Her wisdom and humour always encouraged me to move forward.

I would like to thank two tutors Diane Harper and Tatyana Karpenko-Seccombe. They helped me to improve my English skills, including revising papers and practising my speaking. Their payment makes me get more confident to solve the questions of writing the thesis and the final presentation.

I would like to thank all other friends in this office, including Yiheng Hu, Jing Li, SiYue Song, and other supported colleagues. These talented and helpful people give me wonderful experience when working with them.

I would like to thank the three teachers for their help. Dongsheng Fu, Kecheng Xiao and Junwang Lin, the three teachers positively organized testers to participate in the tests of the games and ensured testers after playing related games to complete online questionnaires. If I do not have the support of these talented and helpful people, my work on the test could not be delivered smoothly. Thanks very much.

Finally, I would like to express my greatest gratitude to all my family members, for their endless support and deep concern. I use this article as a gift to my dear families, thank them for their support and care to me. Studying for 23 years, even though suffering many difficulties, I come here to complete the final step. It will be a perfect and memorable experience for my life when I memory these times. So thanks to all supervisors, friends, colleagues and family members again.

### Abstract

The thesis summarises an exploratory study of digital games being used to promote the development of a virtual inquiry-based learning community (VILC). VILC has been shown to provide an important framework for enhancing students' problem-solving skills. However, little attention has been paid to the promotion of the value in social learning generated by VILC. There exists, a relationship between games and learning values, with learning value generated in social learning, and the use of games supporting social learning. To improve the development of the learning value of VILC, this thesis considers using digital games as specific communities providing the opportunity to create value. A value creation framework is utilised to evaluate social learning in games/communities. As the community of inquiry (COI) is related to the nature of knowledge construction in inquiry-based learning (IBL), VILC involves both COI and IBL. According to this, this research carries out investigations involving the use of game characteristics to support IBL and to promote learning values of COI.

The research onion had been utilised throughout the methodology of this thesis as an effective tool to help the organisation of the research and development of the research design according to the layers of the research onion (i.e. research philosophy, research methods, research strategies, time horizons, and data collection methods).

The first step of this research identifies game characteristics used in the game and game-based strategy that is designed to promote IBL. The analysis revealed that the game genre of roleplaying games (RPG) is often used, given its high suitability to support IBL. A wide variety of game elements were used including storylines, interactive artefacts, gameplay-context coherence and task assessment mechanics. The study suggests that these game features can be designed to provide different levels of support to students, and structure for their learning progression according to different theoretical models and the creation of diverse environments to support making sense of knowledge in context.

The second step reviews activities of COI as the key link which connects COI and value creation. These activities were evaluated to explore related values generated by activities. The results showed that value creation was impacted by teaching presence (TP), social presence (SP) and cognitive presence (CP). In particular, reframing value was not positively promoted by COI. This result provides important evidence by which to explore the third step of this research.

The third step presents a model of game features promoting reframing value to clarify the relationship between digital RPGs and reframing value. The reliability of the model was evaluated by 39 experts according to their teaching experience. They gave a high score to

acknowledge the reliability of the model. The result of the factor analysis of 106 participants (teachers and students) was consistent with the model. This ensured the validity of the model. Participants of tests agreed that 10 game features such as multiple endings of the game, game stories, were inductively extracted from players' comments of RPGs. These features were oriented at promoting four criteria of reframing value (i.e. *promoting reflection, the change of evaluation, generating new assessments for other people and suggestions for community development*). The relationship between digital RPGs and reframing value can support reflection of multiple dimensions (i.e. four criteria of reframing value ) in the process of social learning, create related environments assisting students to understand the knowledge and provide a reference for related stakeholders (e.g. teacher educators) who can use the model to select or design games aiming at promoting reframing value. A guideline was attached in order to provide support to related users of the model.

## **Table of Contents**

### Contents

Copyright statement2
Acknowledgements
Abstract4
Table of Contents 6
List of Tables11
List of Figures
List of abbreviations16
1 Introduction18
1.1 Research background
1.1.1 Games
1.1.2 Inquiry-based learning (IBL)
1.1.3 Community of inquiry (COI)
1.1.4 Virtual inquiry-based learning community (VILC)21
1.1.5 Value creation
1.2 Aims and objectives
1.3 Contributions to the knowledge
1.3.1 Exploring serious games supporting IBL26
1.3.2 The value creation in COI

1.3.3 A model of game features promoting reframing value	27
1.4 Thesis structure	29
1.5 Publications	31
2 Literature review	32
2.1 Serious games	32
2.1.1 Benefits of SGs and computer games in education	33
2.2 Exploring serious games supporting IBL	37
2.2.1 Three core characteristics of IBL	38
2.2.2 Game-based IBL	41
2.2.3 Research question	42
2.3 The value creation in COI	43
2.3.1 Theory background	45
2.3.2 Research question	49
2.4 A model of game features promoting reframing value	50
2.4.1 Theory background for constructing the model	50
2.4.2 Research questions for constructing the model	57
2.4.3 Theory background for the reliability of the model	57
2.4.4 Research questions for the reliability of the model	58
2.4.5 Theory background for the validity of the model	58
2.4.6 Research questions for the validity of the model	60

2.5 Theory background for the research design
2.5.1 Positivism (research philosophy)61
2.5.2 Research methods
2.5.3 Case study (research strategy)
2.6 Summary
3 Methodology
3.1 Exploring serious games supporting inquiry-based learning
3.2 The value creation in communities of inquiry
3.2.1 Data selection
3.2.2 Data analysis
3.2.3 The process of coding
3.3 A model of game features promoting reframing value72
3.3.1 Sample selection73
3.3.2 Data analysis
3.3.3 An instrument for evaluating the reliability of the model
3.3.4 The validity of the model
3.4 Summary
4 Research results
4.1 Exploring serious games supporting inquiry-based learning
4.1.1 What are the characteristics of a game-based inquiry-based learning strategy?88

4.1.2 What game features have been used to support IBL?
4.1.3 What aspects of IBL are being supported by game features?
4.1.4 What are the reported impacts on students?
4.2 The value creation in communities of inquiry104
4.2.1 The actual values that COIs can contribute to improving learning experiences in
virtual environments (answering research question RQ1)
4.2.2 TP, SP and CP represent COI aspects determining the value generation process (answering RQ2)
4.3 A model of game features promoting reframing value
4.3.1 Which game features may influence reframing?
4.3.2 Which evaluative criteria of reframing value are affected by game features?109
4.3.3 How game features promote these criteria of reframing value?
4.3.4 An instrument for evaluating the reliability of the model
4.3.5 The validity of the model
4.4 Summary
5 Discussion
5.1 Exploring serious games supporting IBL
5.2 The value creation in communities of inquiry
5.2.1 Discussion
5.2.2 The importance of promoting reframing value
5.3 A model of game features promoting reframing value

5.3.1 An instrument for evaluating the reliability of the model	
5.4 Summary	146
6 Conclusions and Future work	147
6.1 Conclusions	147
6.1.1 Exploring serious games supporting IBL	148
6.1.2 The value creation in COI	149
6.1.3 A model of game features promoting reframing value	150
6.2 Future work	152
Publications	153
References	154
Appendix A Feasibility of Using Serious Games for Teachers' H	Professional
Development	
Appendix B The questions of the five-cycle value	194
Appendix C Questionnaires	
C.1 A questionnaire for testing the reliability of the model	197
C.2 The questionnaires for testing the validity of the model	100

## List of Tables

Table 2.1: The possible IBL structures have been proposed by scholars 39
Table 2.2: Four distinct levels of inquiry (Banchi and Bell, 2008) 40
Table 2.3: COI categories and indicators (Chanprasitchai and Khlaisang, 2016, Shields, 2003)
Table 2.4: The theory of five-cycle value creation (Wenger et al., 2011)
Table 2.5: The indicators for changes of reflection (Wenger et al., 2011) 56
Table 2.6: The assessment criteria of reframing value (Wenger et al., 2011) 56
Table 3.1: The table for the details of participants' comments
Table 4.1: Characteristics of a game-based IBL strategy 89
Table 4.2: The impacts of serious games used on IBL 99
Table 4.3: Activities of COI
Table 4.4: The statistics form that the appearing numbers of activities in three presences. 108
Table 4.5: Participants' feedback 126
Table 4.6: Limbo- Results of One-way ANOVA Analysis
Table 4.7: World of Warcraft- Results of One-way ANOVA Analysis 134
Table 4.8: Chinese Paladin: Sword and Fairy 3- Results of One-way ANOVA Analysis 134

Table 5.1: The activities of COI involving some questions of value creation
Table 5.2: The values and formal learning environment
Table A.1: Context of use of SGs 180
Table A.2: Impacts of SGs 185
Table B.1: The value creation of the five-cycle (Wenger et al., 2011)
Table C.1: A form for testing the reliability of the model 197
Table C.2: The questionnaires for the game experience test (Chinese Paladin: Sword and Fairy
3)
Table C.3: The questionnaires for the game experience test ( <i>Limbo</i> ) 202
Table C.4: The questionnaires for the game experience test (World of Warcraft)

# List of Figures

Figure 1.1 The triangular relationship between games and VILC
Figure 1.2 The research questions of the first step
Figure 1.3 The research questions of the second step
Figure 1.4 The research questions of the third step
Figure 3.1 Methodology of exploring serious games supporting IBL adopted from Saunders
<i>Research Onion</i> (2003)
Figure 3.2 Methodology of the value creation in COI adopted from Saunders' Research Onion
(2003)
Figure 3.3 Codebook excerpt
Figure 3.4 Methodology of the model adopted from Saunders' Research Onion (2003)72
Figure 3.5 The rank of role-play games in CGWR for iPhone74
Figure 3.6 The rank of role-play games in CGWR for iPad75
Figure 3.7 Codebook excerpt (the number of the themes was collected in Files; the number of
players' comments was collected in References)
Figure 3.8 Methodology for measuring the reliability of the model adopted from Saunders
Research Onion (2003)

Figure 3.9 Methodology for measuring the validity of the model adopted from Saunders'
<i>Research Onion</i> (2003)
Figure 4.1 Game features supporting one core IBL characteristic
Figure 4.2 Game features supporting two core IBL characteristics
Figure 4.3 Game features supporting three core IBL characteristics
Figure 4.4 The model of game features promoting one criterion of reframing value
Figure 4.5 The model of game features promoting two criteria of reframing value
Figure 4.6 The model of game features promoting three criteria of reframing value
Figure 4.7 The relationship between game features and reframing value
Figure 4.8 The frequency of the response ( <i>Chinese Paladin: Sword and Fairy 3</i> )127
Figure 4.9 The frequency of the response ( <i>Limbo</i> )127
Figure 4.10 The frequency of the response (World of Warcraft)
Figure 4.11 Factor analysis (factor loading) with varimax rotation of measures of the
relationship between game features and criteria of reframing value (World of Warcraft)129
Figure 4.12 Factor analysis (factor loading) with varimax rotation of measures of the
relationship between game features and criteria of reframing value ( <i>Limbo</i> )

Figure 4.13 Factor analysis (factor loading) with varimax rotation of measures of the
relationship between game features and criteria of reframing value (Chinese Paladin: Sword
<i>and Fairy</i> 3)132
Figure A.1 Key characteristics of SGs demonstrated effectiveness in mediating teacher
education
Figure A.2 A model of SGs-mediated teacher education outcomes

# List of abbreviations

AR	Augmented reality
ANOVA	Analysis of variance
CG	Control group
COI	Communities of inquiry
СР	Cognitive presence
CGWR	China games weight rank
DCENT	Digital Creativity Enhanced in Teacher Education
DEG	Digital education game
DGBL	Digital game-based learning
EG	Experimental group
FPS	First-person shooting
GBL	Game-based learning
GET	Game experience test
IBL	Inquiry-based learning
ICT	Information and communication technology

IT	Information technology
NPC	No-play character
PC	Personal computer
PR	Popularity rate
RPG	Role-playing game
RQ	Research question
RR	Response rate
SDT	Self-determination theory
SP	Social presence
STIMULATE	Science Training Immersive Modules for University Learning Around Teacher Education
TP	Teaching presence
VET	Virtual experience test
VLC	Virtual learning community
VILC	Virtual inquiry-based learning community

### **1** Introduction

This chapter gives a brief introduction, including the research background, aims and objectives of the research and contributions of the research.

Games are playing an increasingly important role in education (e.g. Fabricatore et al., 2019; Bressler et al., 2021; Hwang and Chen, 2017; Erickson et al., 2021; Barr, 2017), especially by improving social learning (Bakhanova et al., 2020, Medema et al., 2016, Den Haan and Van der Voort, 2018).

There exists, a relationship between games and learning values, with learning value generated in social learning (Wenger et al., 2011), and the use of games supporting social learning (Bakhanova et al., 2020, Medema et al., 2016, Den Haan and Van der Voort, 2018).

Because the game is seen as a specific community where interaction can be generated (Bolter et al., 2000), a value creation framework i.e. Wenger et al.'s framework (2011) can evaluate social learning in games.

The virtual inquiry-based learning community (VILC) has been shown to be an important framework given that it enhances students' problem-solving skills (e.g. Eteokleous and Ktoridou, 2012; Setiani and MacKinnon, 2015; Chanprasitchai and Khlaisang, 2016; Suarez et al., 2016). As VILC is a specific type of community, it can highlight values that other communities can potentially create. However, little attention has been paid to how a

VILC can promote the value in social learning. Hence, the thesis investigates how digital games can be used to promote the development ,and improve the learning value, of VILC.

As the community of inquiry (COI) is related to the nature of knowledge construction in inquiry-based learning (IBL), VILC involves both COI and IBL. Utilising games to support IBL and COI presents considerable challenges in terms of the related issues of identifying game features and the extent to which they promote IBL and COI. This research will therefore carry out investigations involving the use of game characteristics to support IBL and to promote the learning values of COI. The research onion (Saunders et al., 2009) will be utilised throughout to explain the methodology of this thesis, providing an effective structure for this research, particularly through the layers of the research onion (i.e. research philosophy, research methods, research strategies, time horizons, and data collection methods).

#### 1.1 Research background

#### 1.1.1 Games

Both digital and non-digital games have been leveraged for over three decades to foster learning at all stages of formal education (e.g. Connolly et al., 201; Smith and Golding, 2018), in professional development contexts (e.g. Lameras et al., 2014; Bochennek et al., 2007; Fu et al., 2016), and in informal environments (e.g. Matijevic and Topolovcan, 2019; Bayeck, 2020). Research has amply theorised and demonstrated that games can effectively enhance learners' motivation, engagement, and learning outcomes in multiple domains (Boyle et al., 2016, Mayer, 2019, Fabricatore et al., 2019, De Freitas, 2006, Bayeck, 2020).

#### **1.1.2 Inquiry-based learning (IBL)**

Inquiry-based learning (IBL) is a student-centred pedagogical approach (Kember, 1997), in which students find the answers to problems through self-directed inquiry, driving their own learning (Prince and Felder, 2006). Students solve problems, build explanations and construct knowledge by actively exploring, observing, gathering and reflecting on information present in a rich and meaningful context (Barron and Darling-Hammond, 2008, Condliffe et al., 2016, Kuhlthau et al., 2015). In contrast to traditional instructional learning, teachers play a guiding role, supporting students to develop their inquiry and critical thinking abilities so they can learn by themselves (Bennett, 2015). IBL is an iterative process wherein students must evaluate the strengths and weaknesses of their own results, determine their value and decide whether or not they should gather more information to elaborate on the problem being addressed. As a result, inquiry can be seen as a constant trial and error process. Research indicates that IBL can effectively support learning (Hwang et al., 2012, Lim, 2004, Pedaste et al., 2015), improving students' conceptual understanding (Gillies and Nichols, 2015, Wilson et al., 2010), and their higher-order thinking skills (Raes et al., 2012).

#### **1.1.3 Community of inquiry (COI)**

The concept of Community Of Inquiry (COI) refers to a group of individuals collaboratively shared knowledge and technology, engaged in critical dialogue and

rethinking to construct personal meaning, and confirmed mutual understanding (Garrison et al., 1999). The COI theoretical framework supports the creation of online collaborative learning experiences. COI designs and evaluates virtual learning environments through reciprocal interaction of activities among multiple presences, such as Cognitive Presence (CP), Social Presence (SP), Teaching Presence (TP) (Garrison, 2011, Kozan and Richardson, 2014), Emotional Presence (EP) (Cleveland-Innes and Campbell, 2012, Stenbom et al., 2016a) and Learning Presence (LP) (Shea et al., 2013, Hayes et al., 2015).

#### 1.1.4 Virtual inquiry-based learning community (VILC)

COI framework is related to the nature of knowledge construction in IBL (Suárez et al., 2016) and was first proposed by Garrison et al. (2011) based on the model of critical thinking and practical inquiry. Given the relationship between COI and IBL, the virtual inquiry-based learning community (VILC) integrates COI and IBL as a novel approach to learning (e.g. Eteokleous and Ktoridou, 2012; Setiani and MacKinnon, 2015; Chanprasitchai and Khlaisang, 2016; Suarez et al., 2016) to enhance students' question-solving skills (Chanprasitchai and Khlaisang, 2016) and co-create knowledge (Suarez et al., 2016). For example, integrating the IBL approach optimizes the design of activities in COI (Chanprasitchai and Khlaisang, 2016).

#### 1.1.5 Value creation

Value creation refers to the values of social learning which occur through interaction and communication in communities and networks (Reed et al., 2010, Wenger et al., 2011). Social learning is seen as one of the most influential learning theories (e.g. behaviourism

and cognitivism) (Pinho et al., 2020) explaining human behaviour through using continuous interaction among cognition, behaviour and environmental influences (Bandura and Walters, 1977, Prestridge, 2019).

Given its nature, Wenger et al.'s (2011) value creation framework (e.g. creating immediate value, potential value, applied value, realized value and reframing value) represents a suitable tool to evaluate value creation. Social learning is seen as the practical process and results of value creation in the networks (van Amersfoort et al., 2012). In order to explore the nature of social learning (Wenger et al., 2011), the framework can be used as an effective way of seeing implicit learning outcomes (van Amersfoort et al., 2012). For example, the framework is utilised broadly to capture knowledge cocreation (Gidhagen et al., 2011) and understand what community members achieved from their participation (Dingyloudi and Strijbos, 2015). Hence, many studies reiterate the importance of Wenger et al.'s framework in the value evaluation of virtual communities and networks (e.g. van Amersfoort et al., 2012; Smith et al., 2017; Booth and Kellogg, 2015; Dingyloudi and Strijbos, 2015).

#### 1.2 Aims and objectives

The thesis aims to investigate an exploratory study that digital games are used to promote the development of a virtual inquiry-based learning community (VILC).

VILC has been shown to be an important framework given that it enhances students' problem-solving skills (e.g. Eteokleous and Ktoridou, 2012; Setiani and MacKinnon, 2015; Chanprasitchai and Khlaisang, 2016; Suarez et al., 2016). However, little attention has been paid to the promotion of social learning value through VILC. To improve the development of the learning value of VILC, this thesis considers digital games that are used as an educational strategy providing the opportunity to create value.

As the framework of Community Of Inquiry (COI) is related to the nature of knowledge construction in inquiry-based learning (IBL), VILC involves both COI and IBL. This research includes three steps involving the use of game characteristics to support IBL and promote the learning values of COI. This includes the following sub-research: exploring game characteristics supporting three core features of IBL (the first step of this research), investigating actual values created by COI and weaknesses generated during the process of value creation in COI (the second step of this research), and identifying game features that promote related value in order to rectify the weaknesses (the third step of this research), see Figure 1.1.



Figure 1.1 The triangular relationship between games and VILC

In order to achieve the aforementioned aim, the following sub-aims will be pursued:

(1) To investigate how game features have been used to support the inquiry processes, and

how game features may be related to impact on achievement and engagement

For this, the research questions require to be addressed:



Figure 1.2 The research questions of the first step

(2) To investigate the values created by COI and analyse the possible existing weaknesses from the perspective of value creation

For this, the research questions require to be addressed:

What are the actual values that COIs can contribute to improve learning experiences in virtual environments?

To what extent do TP, SP and CP represent COI aspects determining the value generation process?

#### Figure 1.3 The research questions of the second step

(3) To rectify the weaknesses generated by COI

- To establish an empirical model that describes the relationship between reframing value and game features, involving game features helped to deploy reframing functions, and the approaches
- (ii) To evaluate the model through the feedbacks of experts, teachers and students, in order to ascertain whether the identified features can help COI to generate reframing value as represented by them

For this, the research questions require to be addressed:

C	Which game features may influence reframing?
(	Which evaluative criteria of reframing value are affected by game features?
(	How game features promote these criteria of reframing value?
(	What is the reliability of the model?
$\mathcal{L}$	What is the validity of the model?

#### Figure 1.4 The research questions of the third step

#### **1.3 Contributions to the knowledge**

The research work in this thesis includes the following contributions to knowledge:

#### **1.3.1 Exploring serious games supporting IBL**

Several game features (e.g. storylines, interactive artefacts, gameplay-context coherence and task assessment mechanics) could be used to support inquiry processes. The study suggests that these game features can be designed to provide different levels of support to students, structure the learning progression according to different theoretical models and create diverse environments to support making sense of knowledge in context. Games present these intrinsic features (e.g. storylines, interactive artefacts) that can naturally support the core aspects of IBL: the provision of progressive and context-sensitive guidance, the hierarchical structure of the learning process and the creation of a meaningful environment to solve problems. This means that students who play IBL games can take active control of the pace of their progress, exert autonomy for decision-making and make timely observations regarding the relationships between actions, concepts and context. There is much space for these features to be explored and used by games aiming at supporting IBL.

#### **1.3.2** The value creation in COI

Community of Inquiry (COI) has become increasingly popular as an effective framework of social learning that promotes critical thinking and improves learning skills in online settings. The framework has been used to design and evaluate the interaction of virtual learning environments through multiple presences (teaching and social). However, little is known about how COI contributes value by improving learning experiences in virtual environments and what determines the generation of this value. Through reviewing activities of COI that connect COI and value creation and evaluating these activities, this study found the related values created by COI and the extent to which Teaching Presence (TP), Social Presence (SP) and Cognitive Presence (CP) represent COI aspects determining the value generation process. Hence, this study can provide an important means to explore the nature of social learning; being utilised as an effective way of seeing implicit learning outcomes of COI; enabling stakeholders (e.g. participants of a virtual learning community, the organizations of a community and their sponsors) to achieve their expected outcomes; and optimizing the design of related activities to maximize learning value.

#### 1.3.3 A model of game features promoting reframing value

The second step of this research result showed that reframing value was not positively promoted by COI. Promoting reframing value is similar to promoting critical reflection, this research proposes that games are utilised as an educational strategy providing opportunities to create reframing value.

This study presents a model of using role-playing games (RPG) to promote reframing value in order to clarify the relationship between game features and four criteria of reframing value. This is the most important contribution because game designers, related educators and researchers who use games aiming at promoting reframing value or regard the need for a model to link games features with learning value will benefit. The relationship between digital RPGs and reframing value can support reflection of multiple dimensions (i.e. four criteria of reframing value ) in the process of social learning, create related environments assisting students to understand the knowledge and provide a reference for selecting and designing games oriented at promoting reframing value. A guideline was attached in order to provide support to related users of the model.

#### **1.4 Thesis structure**

With the aim that investigating games is used as a robust tool for promoting IBL and COI, this research discusses three aspects including exploring serious games supporting IBL, the value creation in communities of inquiry and a model of game features promoting reframing value. According to the three aspects of this research, the thesis is organized into six chapters and three appendixes to clarify research background, methodology, results, discussions and conclusions.

Chapter 1 gives a brief introduction, including the research background, aims and objectives of the research and the thesis structure.

Chapter 2 reviews the current literature to describe what has been done before and which research questions still need to be solved. The topic involves the theory backgrounds for using serious games supporting IBL, value creation in COI and constructing a model of using RPGs to promote reframing value.

Chapter 3 clarifies the research approaches. This includes related research processes, data capture, data analysis, sample size, calculation of inter-coder consistency and testing the reliability and validity for the model of using RPGs to promote reframing value.

Chapter 4 proposes related research results such as an overview of characteristics of a game-based IBL strategy; the actual values that COIs can contribute to improving learning

experiences in virtual environments; game features that may influence reframing value have been extracted.

Chapter 5 gives discussions. For example, the suitability of RPGs for supporting IBL; the importance of promoting reframing value; some special game features that only appear in stand-alone or online games to promote reframing value.

Chapter 6 presents an overall conclusion of this study and according to related limitations in current research proposes suggestions for future work.

The first appendix presents expanded research i.e. a few studies involve utilising IBLserious games to support teacher training. This research, from a broader perspective, investigates the impact of serious games (SG) on teachers' professional development.

The other two appendixes include a form for the questions of the five-cycle value and questionnaires that test the reliability and validity of the model when using RPGs to promote reframing value.

#### **1.5 Publications**

The work in this thesis has produced four papers, including three papers that have been published in related conference proceedings, and one paper (i.e. Construction of a Model that Uses Digital Role-playing Games Promoting Reframing Value) that is being reviewed by the *Journal of Educational Technology Research and Development*. At the end of this thesis (Publications section), the full publication list can be shown.

## 2 Literature review

This chapter reviews the current literature to describe what has been done before and which research gaps still need to be addressed. The topic involves serious games, educational research in games and serious games, related research questions and theory backgrounds for the three steps of this study: using serious games supporting IBL, the value creation in COI and constructing a model of game features promoting reframing value.

#### 2.1 Serious games

Based on the research of Susi et al. (2007), serious games initiated from the "serious games initiative" in 2002 (seriousgames.org). "The Serious Games Initiative is focused on uses for games in exploring management and leadership challenges facing the public sector. Part of its overall charter is to help forge productive links between the electronic game industry and projects involving the use of games in education, training, health, and public policy" (Susi, et al., 2007 p.3, seriousgames.org).

With the term "serious game" becoming widely used, there seem to now be many different definitions available for the concept. For example, Susi et al. (2007) proposed that the definition of serious games (SG) should be studying the relationship between the game aspect and the application of gaming technology. They suggest that Serious games are adopted to game technology, process and design in solving problems faced by enterprises and other institutions. It can promote the transfer and cross-application of game development knowledge in some non-game markets (i.e training, sales) (Susi et al., 2007).

From the perspective of the application of SGs, the entertainment dimension (Michaud et al., 2008, Jantke, 2010, Zyda, 2005), the user's experience (De Freitas and Liarokapis, 2011, Arnab et al., 2011) and the combination of media (Lin et al., 2006, Arnab et al., 2011), Laamarti et al. (2014) all contribute to a definition of SGs: the SG becomes an application with three components (e.g. experience, entertainment, and multimedia). Jacobs (2021), considering the theoretical perspectives on player choice in SGs, therefore proposed that SGs can be seen as a form of promotional communication, media experience and technical innovations.

However, whilst there is no absolute agreement on a definition, most scholars agree on a core meaning when the primary purpose of SGs is within education (Bontchev et al., 2021), especially when applications from relevant areas such as game-based learning (Deterding et al., 2011, Plass et al., 2015) are included, rather than simply entertainment (Landers and gaming, 2014, Susi et al., 2007, Michael and Chen, 2005). Game-based learning products are therefore widely considered as a subcategory of SGs (Connolly et al., 2012, Hainey et al., 2016). This thesis similarly considers SGs through the conceptualisation of digital games/ game-based learning design, to consider how they promote behaviour change and/or the development of knowledge, skills and attitudes (Hainey et al., 2016, Connolly et al., 2012, De Lope and Medina-Medina, 2017).

#### 2.1.1 Benefits of SGs and computer games in education

Serious games represent an important asset in education, as there is rich evidence to show the benefits of SGs' use. For example, SGs can promote the development of transferrable knowledge, skills and attitudes by engaging players in activities driven by meaning-making, and by setting scenarios that mirror real-world situations (Fabricatore et al., 2019). Bressler et al. (2021) indicate that mobile augmented reality games are an effective strategy that can provide benefits by supporting serious science learning. Hwang and Chen (2017) have utilised SGs to improve students' critical thinking and problem-solving. Furthermore, SGs have also been shown to successfully enhance teachers' professional development, facilitating the acquisition of domain-specific knowledge and skills (Johnson et al., 2021, Meletiou-Mavrotheris and Prodromou, 2016, Alyaz and Genc, 2016, Lameras et al., 2014) and fostering positive attitudes towards modern pedagogical approaches and tools such as game-based learning (Kennedy-Clark et al., 2011, Annetta et al., 2014).

In particular, SGs have been shown to have impacts on learners' motivation and engagement (Vrasidas and Solomou, 2013, Cózar-Gutiérrez and Sáez-López, 2016, Lorenzini et al., 2015, Farrell and Moffat, 2014). According to Deci and Ryan's (2000) Self-Determination Theory (SDT) which summarises human motivation, the individual has innate tendencies toward psychological growth, which can be influenced by ability, autonomy, and relatedness. From this theoretical perspective, serious games are constructed to improve abilities through tasks, offer opportunities to enable player autonomy (Blumberg et al., 2013), and link with individual values that exist outside of the game environment (Ryan et al., 2006). Hence, motivation and engagement are closely linked to the development of SGs. Hookham and Nesbitt (2019), for example, have found that some potential evaluation approaches such as questionnaires, interviews, time and

performance on tasks can be effectively utilised to measure motivation and engagement in SGs.

Computer games can also provide other positive impacts on education. Many studies (i.e. Enochsson et al., 2004; Guy et al., 2005; Loparev et al., 2014; Erickson et al., 2021) have reported that playing computer games can promote the development of diverse learning skills such as spatial skills (Enochsson et al., 2004, Mitchell and Savill-Smith, 2004) and graduate skills (i.e. communication, adaptability and resourcefulness) (Barr, 2017). For example, the experience of playing with spatial models is provided by computer games to develop students' abilities for creation and design (Coyne, 2003, Guy et al., 2005). Thomas and Brown (2011) have advocated that a new learning experience can be afforded by digital games. They highlighted that in order for multiple methods of completing a task to be found, players are required to be adaptable through thriving on change and strategically handling game resources. Yet other advantages are pointed out by scholars such as Squire, 2008, Boyle et al., 2016, Mayer, 2019, Fabricatore et al., 2019, De Freitas, 2006 and Bayeck, 2020, who argue that games can effectively enhance learners' motivation, engagement, and learning outcomes.

Another potential benefit of games is in improving social learning (Bakhanova et al., 2020, Medema et al., 2016, Den Haan and Van der Voort, 2018). Since games have provided the necessary collaboration, and involve participatory stakeholders' (e.g. learners) communication, positive social learning achievements are created (Ampatzidou et al., 2018, Becu et al., 2017, Medema et al., 2016, Salvini et al., 2016). For example, the game
experience can promote the development of students' diverse learning skills (i.e. Enochsson et al., 2004; Guy et al., 2005; Loparev et al., 2014; Erickson et al., 2021) such as communication, adaptability and resourcefulness (Barr, 2017).

Other potential benefits of games include providing social spaces, social comparison/ competition, and social interaction. Playing video games is seen to be a social activity according to Eklund (2015), as digital games (e.g. massively multiplayer online games) are usually social environments (Squire, 2008) that offer cooperation and competition (Fox et al., 2018) between the millions of players that are attracted by such games (Quandt et al., 2014). For instance, players utilise competitive scoring and they have the skills to help each other achieve success (Loparev et al., 2014). More specific impacts have also been reported, for example by Fox et al. (2018) who found a positive correlation between players' selfmotivated and engaging in social comparison with other players. Several studies (e.g. Trepte et al., 2012; Cole and Griffiths, 2007) have also proposed that online and offline games promote social interaction involving the development of friendship, social capital and social support.

#### 2.2 Exploring serious games supporting IBL

IBL has been regarded as an approach of student-centred learning (Barron and Darling-Hammond, 2008, Condliffe et al., 2016, Kuhlthau et al., 2015), such as observations, investigations, concluding answers or explanations (NRC, 2000, Booven, 2015). Researchers indicated that IBL can effectively support learning (Pedaste et al., 2015, Hwang et al., 2012, Lim, 2004), such as improved students' conceptual understanding (Gillies and Nichols, 2015, Wilson et al., 2010) and their higher-order thinking (Raes et al., 2012). Kogan and Laursen (2014) demonstrated this by conducting an experiment on the comparison of student performance between non-IBL and IBL courses. The study showed that students who took the IBL program scored better than those who participated in non-IBL courses.

Many previous studies had proved that using information technology (IT) as a support in the process of IBL positively affected learning effectiveness (Kuhn et al., 2000, Hwang et al., 2012). It meant information-technology-assisted IBL provided students with the choice to develop their high-order abilities and engaged them with the learning performance of problem-solving (Oliver, 2008, Ucar and Trundle, 2011). Over the last decade, serious games as a part of IT, have been applied to support IBL for teaching and learning (San Chee et al., 2011, Sabourin et al., 2012, Kennedy-Clark et al., 2013, Kalz et al., 2014, Meesuk and Srisawasdi, 2014, Firssova et al., 2014, Dorji et al., 2015, Hwang and Chen, 2017, Srisawasdi et al., 2019). These research outcomes were not only discussed using serious games to promote students' problem-solving skill in IBL but also indicated that serious games assist students to gain other achievements. For example, Sabthisin et al. (2012) improved the student's ability to collect information through tests of inquiry performance in a game-based learning environment.

Through systematical reviewing literature of the concept of engagement in serious games published from 1970 to 2015, Hookham and Nesbitt (2019) found that many studies (e.g. Adamo-Villani et al., 2013; Lorenzini et al., 2015; Farrell and Moffat, 2014) directly evaluated engagement in a serious game. However, Hookham and Nesbitt (2019) did not investigate whether or not these serious games were relevant to IBL. This means a research gap on how IBL-based serious games may be related to impacts on students' engagement needs to be addressed.

Hence, when using serious games to carry out IBL activities, it remains two challenges: a) to understand what games features were used to support IBL and how they have been designed to develop learning, b) to understand how IBL games may be related to impacts on students' achievement and engagement. In order to explain these questions, the literature overview has been utilised to analyse the relationship between serious games and IBL.

#### 2.2.1 Three core characteristics of IBL

There are three core characteristics of IBL that need to be present to support students' learning:

1) The structure of IBL

The literature agrees that IBL needs to be structured in stages in order to guide students in their inquiry process (Bybee et al., 2006, Pedaste et al., 2015, Quintana et al., 2004, Bell et al., 2010). Several possible structures have been proposed by the literature, based on different theoretical approaches. For example, Bell et al (2010) indicate that IBL should be structured in phases of question definition, data gathering, elaboration of explanations, creation of connections, and building of justifications. Pedaste et al (2015) identify different key phases: orientation, conceptualization, investigation, conclusion and discussion. Beyond their differences, they highlight that IBL should follow these stages in a flexible way, allowing students to move forward or go back to previous stages to elaborate coherent answers and sound conceptual explanations to the problems they need to solve. See Table 2.1.

Table 2.1: The possible IBL structures have been proposed by scholars

Process (Bell T, 2010)	Phases (Pedaste et al., 2015)
Creating: proposing interest question	Orientation: inspiring interest
Obtaining: using evidence to solve problems	Conceptualization: formulating hypotheses
Explaining: interpretation	Investigation: conducting a survey
<b>Connection</b> : communicating the information to the others collected in an investigation	<b>Conclusion</b> : concluding from the collected data
Justification: discussion	<b>Discussion</b> : including communication and reflecting

2) IBL environments

This means IBL should provide an internally coherent learning environment where students can perform learning actions by themselves (Wang et al., 2014). Moreover, some authors highlight that IBL should be situated in real-world environments (or virtual recreated versions), otherwise learning may not actually happen (Brown et al., 1989, Huang et al., 2014). Activities contextualised in real scenarios promote students linking their prior knowledge to the new one (Brown et al., 1989, Hwang et al., 2011). Studies have shown that IBL is promoted when students observe real phenomena and practice in a natural environment (Lee and Butler, 2003, Hwang et al., 2012, Oliver, 2008).

#### 3) Guidance

Guidance is key in the whole process of IBL, however, the degree of guidance can vary according to students' abilities to self-direct their learning (see Table 2.2). Banchi and Bell (2008) indicate that teachers (or other supporting entities) need to guide students from level 1 (strong support to inquiry), progressively reducing the support until students are able to carry out inquiry processes in a more autonomous way (Level 4). However, even when students are able to guide themselves, the open inquiry is not likely to be successful unless students have a positive motivation (Yoon, 2012), which highlights the importance of guidance and support even when their inquiry skills seem more developed.

 Table 2.2: Four distinct levels of inquiry (Banchi and Bell, 2008)

Level	Description
1.Confirmation inquiry	Teachers select a theme to teach, then promote questions that guide the student completing the activity to find the result. This method is very good for strengthening concept teaching, guiding students to learn to follow the

Level	Description
	procedure, collecting and recording data correctly, identifying and deepening
	their understanding.
2. Structured	Teachers provide the outline of the procedure and the starting questions, and
	students should develop an explanation of what they have found by evaluating
inquiry	and analysing the data collected.
3. Guided	Teachers provide students with research problems, but designing and carrying
	out the research is students' responsibility. Then, students share their findings
inquiry	with teachers and other students.
1 Open/true	Students develop the research question, define inquiry methods and carry out
4. Open/true	the inquiry process by themselves. Then, students share their findings with
inquiry	teachers and other students.

## 2.2.2 Game-based IBL

Due to their motivational potential and interactive nature, games have been proposed as a suitable format to support IBL (Meesuk and Srisawasdi, 2014, Firssova et al., 2014, Dorji et al., 2015, Hwang and Chen, 2017). Some game features may be particularly relevant to support IBL. Games can implicitly or explicitly provide problems and goals to achieve (Hwang et al., 2015, Firssova et al., 2014); an environment with resources and action possibilities that are meaningful to the problems to be solved (Hwang and Chen, 2017, Hwang et al., 2015); opportunities to actively explore, manipulate and build a comprehension of the inner workings of the game (Kennedy-Clark et al., 2013); and immediate feedback to correct errors, inform about performance and expand knowledge (Sabthisin et al., 2012). In addition, games can present specific rules and gameplay flow oriented to supporting players to understand what to do and when (San Chee et al., 2011). Furthermore, games can give players control over their actions and foster motivation by

presenting challenging tasks and providing relevant rewards (Lameras et al., 2014, Dorji et al., 2015).

## 2.2.3 Research question

Despite the potential that games have to support IBL, little attention has been given to understand how game features have been used to support the inquiry processes, and how these may be related to impacts on achievement and engagement. The aim of this study to review the literature focused on IBL and games, to identify game features that have been used in games and gamified strategies designed to support IBL, and synthesise their impacts on students. The research questions are:

- (a) What are the characteristics of a game-based IBL strategy?
- (b) What game features have been used to support IBL?
- (c) What aspects of IBL are being supported by game features?
- (d) What are the reported impacts on students?

## 2.3 The value creation in COI

According to Garrison et al. (1999), the concept of Community Of Inquiry (COI) refers to a group of individuals collaboratively engaged in critical dialogue and reflection to construct personal meaning, shared knowledge and confirm mutual understanding. The theoretical framework evaluates virtual learning environments through reciprocal interaction of activities among multiple presences, such as cognitive presence (CP), social presence (SP), teaching presence (TP) (Garrison, 2011, Kozan and Richardson, 2014), emotional presence (EP) (Cleveland-Innes and Campbell, 2012, Stenbom et al., 2016a) and learning presence (LP) (Shea et al., 2013, Hayes et al., 2015). The framework has been used broadly (Kozan and Richardson, 2014, Stenbom et al., 2016b) to enhance the effectiveness of online education research and practice (Kozan and Richardson, 2014), improve students' learning skills (Mikroyannidis et al., 2016, Secundo and Grippa, 2010), and promote critical thinking (Chanprasitchai and Khlaisang, 2016, Eteokleous and Ktoridou, 2012, Junus et al., 2017, Keles, 2018, Padilla and Kreider, 2018).

Value creation refers to the values generated by social learning, through interaction and communication in communities and networks (Reed et al., 2010, Wenger et al., 2011). Scholars indicate that social learning is one of the most influential learning theories (e.g. behaviourism and cognitivism) (Pinho et al., 2020). As it uses the continuous interaction among cognition, behaviour and environmental influences to explain human behaviour (Bandura and Walters, 1977, Prestridge, 2019). And given social learning includes

attention, memory and motivation, it is often used to bridge behaviourist learning theories and cognitive learning theories (Muro and Jeffrey, 2008).

The evaluation of learning values has an important meaning for social learning. This process can explore the nature of social learning (Wenger et al., 2011) and make related stakeholders of the value (e.g. participants of a virtual learning community, the organizations of a community and their sponsors) achieve desired outcomes (Smith et al., 2017). For example, managers are likely to be more interested in performance (value achieved) (Wenger et al., 2011).

Wenger et al.'s (2011) value creation framework represents a suitable tool to evaluate value creation in online environments. This framework indicates that learners' activities in Virtual Learning Communities (VLC) can create and achieve immediate value, potential value, applied value, realized value, and reframing value. In the framework of value creation, social learning is seen as the practical process and results of value creation in the networks (van Amersfoort et al., 2012). This means that the framework is an effective way of seeing implicit learning outcomes (van Amersfoort et al., 2012). For example, the framework is utilised broadly to capture knowledge cocreation (Gidhagen et al., 2011) and understand what community members achieved from their participation (Dingyloudi and Strijbos, 2015). Meanwhile, there an emerging body of literature (e.g. van Amersfoort et al., 2012; Smith et al., 2017; Booth and Kellogg, 2015; Dingyloudi and Strijbos, 2015), reiterating the importance of Wenger et al.'s framework in the value evaluation of virtual

communities and networks. Hence, given its nature, this research uses Wenger et al.'s framework to construct an investigation.

As COI is a specific type of VLCs, it may be assumed that they create the values that have been identified as being created by VLCs. However, few studies have investigated the actual values that COIs can contribute to support learning experiences in virtual environments and the factors that determine the generation of these values. This research adopts a mixed content analysis (inductive and deductive) to investigate values generated in COI, in order to address such a gap.

## 2.3.1 Theory background

#### 2.3.1.1 The presences of COI

The literature agrees that TP, SP and CP are three original/ core presences of the COI framework (Chanprasitchai and Khlaisang, 2016, Garrison, 2011, Garrison et al., 1999, Shields, 2003). The three presences are used as a theoretical reference by this research. TP refers to key strategies that teachers can use to create a COI among learners (Bangert, 2008), and supports the design of activities in the virtual world (Burgess et al., 2010, Dalgarno and Lee, 2010, Pellas and Boumpa, 2017). SP provides social context and interaction to achieve knowledge (Pardales and Girod, 2006, Seixas, 1993), identify with and connect among participants in an online learning environment (Chanprasitchai and Khlaisang, 2016). According to Garrison et al. (2011), CP means to investigate the process of learners building meaning through discussion and reflection. The categories and indicators of the three presences are shown in Table 2.3.

Table 2.3: COI categories and indicators (Chanprasitchai and Khlaisang, 2016,Shields, 2003)

Presences	Categories of activities	indicators
	Design and organization	Setting environment and methods
ТР	Promoting discourse	Shaping constructive communication
	Direct instruction	Focusing and resolving questions
	Individual/affective	Self-projection/expressing emotions
SP	Open communication	Learning circumstance/risk-free expression
	Group cohesion	Collaboration
	Triggering event	Sense of puzzlement
СР	Exploration	Information exchange
CP	Integration	Connecting ideas
	Resolution	Apply new ideas

## 2.3.1.2 A theory of five-cycle value creation

In order to interpret the learning values, Wenger et al. (2011) propose a theory of five-cycle value creation that promotes and evaluates values in the community and a networked environment. The five values include immediate value, potential value, applied value, realized value and reframing value. See Table 2.4.

Definition	Explanation	Example
Immediate value Members' activities and interactions have value. Potential value The immediate value may not be shown immediately but is preserved as intellectual capital. This value is potential that is likely to	The interaction and communication among communities' members directly create immediate value. The value does not appear immediately. It includes some forms of knowledge capital such as human capital, social capital, tangible capital, intangible	ExampleHelping each other; meetings; dialogues; advice; a story be discussed; a project; a visitLooking at new perspectives of the problem, new ways to solve problems; inspiration, care, confidence, status reputation, mutual understanding, accompany, learning challenges among members; accessing to information, resources, tools; the reputation of the
achieve in the future. <b>Applied value</b> The knowledge capital is used that requires adjustments to the capital and its application to specific	capital, learning capital. The changes, or innovations of actions, practices, tools, methods, system, are due to the application of intellectual capital in different	community; transformation of learning knowledge Using the knowledge learned in the community Using the resources provided in the community to practice
situations. Realized value It is not enough to adopt new practices or tools. Changing practice does not necessarily improve performance. Therefore, it is important to find out how the application of intellectual capital affects what is important to stakeholders.	Members are not satisfied with applications, new practices or tools and expect to practice new ideas or to use community resources in order to improve performance.	Achievements; comprehension; improvement of abilities
<b>Reframing value</b> Learning triggers a reflection for the definition of success. It includes re-engineering strategies and aims. When learners interact and share their knowledge, they can	When social learning leads to a rethinking of the definition of success, reshaping will be realized.	Definition of assessment criteria; suggestions for community development; self-reflection; new standards for learner's evaluation

# Table 2.4: The theory of five-cycle value creation (Wenger et al., 2011)

Definition	Explanation	Example
recognise the quantity of		
their knowledge and assess		
the values themselves.		

According to the definition of reframing value, it is consistent with critical reflection. Scholars indicate that critical reflection means rethinking a person's choices, decisions and behaviours (Brookfield, 2015), to make people challenge and question the validity of a meaning or idea that is long taken for granted (Mezirow, 1990).

The five values represent the different desired outcomes of different stakeholders (Wenger et al., 2011). This means that different values may be important to different stakeholders. For example, a coordinator may be more focused on successful activities or the generation of outputs (immediate value and potential value) (Wenger et al., 2011).

In order to elaborate on possible consequences within the process of generating values, Wenger et al. (2011) set some key questions for every value (i.e. potential value: Was I getting new skills or knowledge?). See Appendix B. In this research, these questions are used as a reference for evaluating values. Learning is not a linear process of producing and applying knowledge. Wenger et al. (2011) propose that value creation interweaves into the process of learning. In other words, the cycle of value creation is also not a linear process. For example, every value can be analysed independently. When a five-cycle of value creation has finished, it does not mean the next cycle begins (Wenger et al., 2011).

#### 2.3.2 Research question

Wenger et al. (2011) indicate that the learning value of Virtual Learning Communities (VLC) stems from the ability which promotes learning. In other words, the development of learning values means the enhancement of the ability of learning. In order to promote the development of learning values, the primary research is to identify these actual values that COIs can contribute to support learning experiences in virtual environments and the factors that determine the generation of these values. For this, the following research questions (RQ) are addressed:

1) What are the actual values that COIs can contribute to improving learning experiences in virtual environments?

2) To what extent do TP, SP and CP represent COI aspects determining the value generation process?

## 2.4 A model of game features promoting reframing value

#### **2.4.1** Theory background for constructing the model

The concept of reframing value is derived from the theory of value creation which was first developed by Wenger et al. (2011). They propose that reframing value triggers rethinking of the definition of success both for participants and for their environment, including multiple changes of reflection such as rethinking for methods, aims and values, and reshaping the new criteria of success (Wenger et al., 2011). The changes of reflection are one of the learning values, such as immediate value, potential value, applied value, realized value and reframing value, that are generated in social learning.

There exists, therefore, a relationship between games and learning values, with reframing value generated in social learning, and the use of games supporting social learning (Bakhanova et al., 2020, Medema et al., 2016, Den Haan and Van der Voort, 2018). This research proposes that games can be utilised as an educational strategy as they offer opportunities to create reframing value.

This research introduces the findings of a study clarifying the relationship between games and reframing value, based on Wenger et al.'s (2011) concept. It evaluates the reframing value of games within the context of role-playing games (RPG).

Since Wenger et al. (2011) suggested a framework of five-cycle value creation (immediate, potential, applied, realized and reframing values), the importance of this framework for evaluating values of social learning in communities has been reiterated by several authors

as part of an emerging body of literature (e.g. Van Amersfoort et al., 2012; Booth, 2015; Dingyloudi, 2015; Smith, 2017).

Social learning is used to explain human behaviour (Bandura and Walters, 1977, Prestridge, 2019) as one of the most popular learning theories (e.g. behaviourism and cognitivism) (Pinho et al., 2020). For Wenger et al. (2011)'s framework, social learning is seen as the actual process of generating values in the community (Van Amersfoort et al., 2012). This means that implicit learning outcomes, which include learning reflection, can be shown through the analysis of the five values (Van Amersfoort et al., 2012). Meanwhile, each value can be analysed independently (Wenger et al., 2011). Wenger et al.'s (2011) concept of reframing value was therefore adopted in this research as an appropriate indicator.

This research of the first step found that RPGs suit to support IBL given its nature. Tychsen (2006) proposes that RPGs are distinguished from other game genres through controlling the virtual characters with the use of storytelling in a fictional reality, this is consistent with the guidance of IBL. For example, Hwang, Chiu, and Chen (2015) use RPGs to explore the learning of financial knowledge. The storyline of the game guilds learners to enter Saving Island (game scene) to collect all the scrolls (game tools). Then they implement tasks in the Finances Bridge (game scene) following the storyline. Finally, learners achieve enough scores to participate in the competition in Investment Island (game scene) until the end of the game (Hwang et al., 2015).

Videogames offer a very rich ground for studies on virtual communities (VC) (Tardini and Cantoni, 2005) because the game is seen as a specific community where interaction can be generated (Bolter et al., 2000). According to Gee (2007), players have different identities when playing games (e.g. the real player, the virtual character). Tardini and Cantoni (2005) indicate that, based on the relationship between the real player's and the virtual character's identities, two different VCs can be constructed, namely the playing community and the player's community. In the playing community, for example, the virtual characters interact in the game's world with other characters or non-playing characters to complete tasks such as killing monsters. In the players' community, for instance, the interaction is generated between the real players who are interested in the same game (Newman, 2013).

The literature proposes that RPGs are broadly applied as one of the most popular game genres in multiple educational domains (Cheng et al., 2015, Hao and Lee, 2019, Jancic and Hus, 2018) to support science education (e.g. Wesselow and Stoll - Kleemann, 2018; Moreau et al., 2019; King, 2018), promote inquiry-based learning (IBL) (e.g. Hwang et al., 2015; Sabthisin et al., 2012; San Chee et al., 2011; Lameras et al., 2014; Kennedy-Clark et al., 2013) and develop language learning (e.g. Zhang et al., 2019; Kostikova et al., 2019; Cornillie et al., 2012; Yang and Quadir, 2018). This means that the popularity of RPGs can provide rich resources (e.g. feedback, comments and cases) for this research. Furthermore, scholars have highlighted that the slower game playing speed within RPGs provides more space and time to promote reflection (Squire and Barab, 2004).

As RPG game is a specific type of digital games, it can highlight values that any games can potentially create, and given the nature of this game genre and its popularity for education, RPGs are well-suited to the purpose of this investigation.

In particular, this study proposes that games can be used to promote reframing value. Promoting reframing value in education is important and yet there is a lack of empirical studies in related areas (from the perspective of games promoting critical reflection and generating learning value).

Promoting reframing value plays a significant role in education. The literature proposes that critical reflection means challenging and questioning the validity of ideas (Mezirow, 1990) and reflecting on the person's choices, decisions and behaviours (Brookfield, 2015). The core of critical reflection lies in "the idea of systematically and rigorously examining an idea, an experience, a problem, with an attitude of open-mindedness, whole-heartedness, curiosity, and responsibility" (Saric and Steh, 2017, p. 70-71). The definition of reframing value is consistent with critical reflection. Hence, it may be assumed that reframing value is as significant in education as critical reflection, which improves learning skills and problem-solving abilities. The studies show that learners reconstruct their pre-knowledge using critical reflection to improve learning skills (e.g. Brookfield, 2015; Lonka et al., 2001; Blatt et al., 2007; Mamede et al., 2008). Dewey (1933) proposed that reflective action is the active component of behavioural intervention, so reflective action is a useful tool to address difficult issues (Howard, 2003). However, this research mainly analyses the relationship between games and learning value. Reframing value highlights that it is a

learning value to show the changes of reflection, so the author considers how reframing value is promoted rather than critical reflection.

In addition, the perspective of stakeholders also highlights the importance of promoting reframing value. The reframing value in Wenger et al.'s (2011) study included the consideration of institutions/organizations and stakeholders in communities. For example, learners are the stakeholders of the reframing value because they focus on solving problems that occur in their study and when defining success (Wenger et al., 2011). When RPGs replace communities as the social environment which evaluates values, these institutions/organizations and stakeholders play specific roles (e.g. the participants of a game and their sponsors/developers) that can gain reframing value.

Many previous studies have shown that using games promotes critical reflection (e.g. Apperley and Beavis, 2011; 2013; Hsiao, 2007; Squire and Barab, 2004; Johnson and Mayer, 2010). For example, games as action-based projects provide the non-visual and non-textual elements of gameplay that can lead to critical reflection (Apperley and Beavis, 2011). Video games are utilised to simulate bias and inaccuracies in order to foster students' rethinking (Squire and Barab, 2004). Game experience scaffolds learning and offers instant feedback/reflection (Hamari et al., 2016). These studies from multiple perspectives such as gameplay, simulation and game experience analyse the effects of the use of games in promoting critical reflection.

Several studies consider games triggering accumulation which refers to the previous actions of players that are reflected in the current situation of the game (O Neil, 2007, Raessens and Goldstein, 2005, Vogel et al., 2006, Vorderer and Bryant, 2012). The accumulation can be explicit through some specific forms including game characteristics (e.g. points, badges). For example, points, badges, and leader boards are used to offer positive reinforcement, motivating a learner's performance (Skinner, 1965, Woolfolk, 1998) and enabling them to measure progression and feedback on behaviour (Gnauk et al., 2012) inspiring student self-efficacy (Bandura, 1982). This is likely to make students rethink learning content and values through their reflection in order to reshape themselves (Suls et al., 2002). This means that game features, to some extent, influence critical reflection. However, few studies have been carried out to investigate the way in which these features impact critical reflection.

Furthermore, little attention has been paid to analysing game features that support learning value. Although there are studies to discuss the value created by video games, the value involves the economic value of the games industry (Marchand and Hennig-Thurau, 2013, Gidhagen et al., 2011) not learning value. Hence, the research is important as it empirically explores how the game features can be related to reframing value and thus contributes to filling the lack of research in this area.

#### 2.4.1.1 Four criteria of reframing value

Reframing value involves multiple changes of reflection, including rethinking for methods, aims and values, and reshaping the new criteria of success (Wenger et al., 2011). The

changes of reflection both for participants and for their environment are shown in Table

2.5.

Typical indicators	Some potential sources
	New learning agenda
Community aspirations	New discussions about the value
	New horizon
Aggggmont	New indicators
Assessment	New evaluation process
Deletionship with	Different discussions with stakeholders
Relationship with	New stakeholders' involvement
stakeholders	New expectations
Institutional abongo	The new strategic trend that reflects the new situation
Institutional change	understanding
New frameworks	New social, institutional, legal or political systems
	(appear or create)

 Table 2.5: The indicators for changes of reflection (Wenger et al., 2011)

According to the indicators for changes of reflection, Wenger et al. (2011) inductively spit its assessment criteria into four functions: rethinking learning content and values through reflection, changing the processes and patterns of evaluation, reframing the assessment standard for other people, reshaping the new expectation of whole community development.

 Table 2.6: The assessment criteria of reframing value (Wenger et al., 2011)

Criteria	Reflecting results on value creation: key questions
<i>Promoting reflection</i> (rethinking learning content and values through reflection)	Does the social learning process lead to a reflection on important issues? If this can change people' perspective of understanding important things?
<i>The change of evaluation</i> (changing the process and pattern of evaluation)	Does this mean that new standards and new indicators should be included in the assessment?

Criteria	Reflecting results on value creation: key questions
Generating new assessments for other people (reframing the assessment standard for other people)	How those people who have the right to decide standards of success are impacted by the new understanding?
Suggestions for community development (reshaping the new expectation of the whole community development)	Can the new understanding change into institutional change? Due to this new understanding, is there a new framework or system developed or created?

## 2.4.2 Research questions for constructing the model

In order to clarify the relationship between game features of RPG and the four criteria of reframing value, this study presents identifying game features and the way that these features promote reframing value. For this, the research investigated the following questions:

- (1) Which game features may influence reframing?
- (2) Which evaluative criteria of reframing value are affected by game features?
- (3) How game features promote these criteria of reframing value?

### 2.4.3 Theory background for the reliability of the model

An instrument was developed to refine the model of using game features promoting reframing value and capture experts' perceptions for the relationship between game features and criteria of reframing value. This instrument was designed according to the method of Moore and Benbasat (1991) since they propose that an instrument can measure diverse perceptions based on an individual adopts innovation of information technology

(IT). Many studies (e.g. Yılmaz and Olgan, 2020; Gyaurov et al.,2019; Zhang et al., 2011) utilise the instrument as a useful tool for measuring the adoption of study and diffusion of IT innovations (Moore and Benbasat, 1991).

Cronbach's alpha (Cronbach, 1951) was used to measure the reliability of questionnaires. the reliability of the scale depends on the value of Cronbach  $\alpha$  to judge, including  $\alpha$  is above 0.8 (very good),  $\alpha$  is above 0.7 (acceptable),  $\alpha$  is above 0.6 (should be revised) and  $\alpha$  is lower than 0.6 (needs to be redesigned) (Eisinga et al., 2013).

#### 2.4.4 Research questions for the reliability of the model

In order to ensure the model of game features promoting reframing value is reliable, the problem needs to be solved: what is the reliability of the model?

#### **2.4.5** Theory background for the validity of the model

This research designs a game experience test (GET) to evaluate normal RPGs participants interacted with in. The GET is built on Chertoff et al.'s (Chertoff et al., 2010) virtual experience test (VET). Since the VET has been presented by Chertoff et al. (2010), there is an emerging body of literature (e.g. Valkov and Flagge, 2017; Palomba, 2020; Turchet et al, 2012) indicating the effectiveness of the VET in the evaluation of virtual experience including digital games.

Such VET refers to a holistic method that measures virtual environment experiences integrating five dimensions sensory, relational, affective, cognitive and active (Chertoff et al., 2010). Every dimension involves related items and 24 items in total are evaluated by

the VET scale using a five-point Likert scale (from 1=strongly disagree to 5=strongly agree).

According to Chertoff et al. (2010), they propose the following explanations of the five dimensions: the sensory dimension regards the use of senses when interacting with the virtual environment. For example, "I found the visual display hardware to be of high quality" (Chertoff et al., 2010, p.106).

The cognitive dimension contains five heuristics, including the ability that the environment supports multiple solutions for the same task, the interest of perceived tasks, description of environmental rules, and the ability of task participation supported by the environment. For example, "I felt that the virtual environment allowed me to complete my task in several different ways" (Chertoff et al., 2010, p.106).

The emotional dimension focuses on the user's expected emotional impact on the scene and the environment conveys the desired emotion of users through dialogue, gestures, facial expressions and audio. For example, "I felt a variety of emotions while working on the environment's tasks" (Chertoff et al., 2010, p.106).

The relational dimension contains the expected level and quality of the interaction between the user and the agent in the environment. For example, "I think that the environment was able to support multiple human users at the same time" (Chertoff et al., 2010, p.106).

The active dimension involves the degree to which users see themselves as characters in the context, the expected level of reusing materials, and the level of narratives used. For example, "I feel that I could construct a story about my actions in the environment" (Chertoff et al., 2010, p.106).

Nevertheless, this GET is only related to four dimensions relational, affective, cognitive and active since the main aim is not to explore the sensory content or the consistency of sensory information. Rather author is interested in emotions generated in the game, the judgment of the relationship between game features and criteria of reframing value, participants' perceived abilities to complete tasks and to understand the characters' behaviours or environments rules, and the assessment of proposed tasks.

## 2.4.6 Research questions for the validity of the model

In order to investigate the validity of the model, the research question is proposed: what is the validity of the model?

## 2.5 Theory background for the research design

Saunders et al. (2003) suggest that a research process is like an "onion". When people before seeing the centre of the onion, many "onion skin" needs to be peeled off. From the outside to the inside, these "onion skins" represent respectively: research philosophy, research methods, research strategies, time horizons, and data collection methods.

Since Saunders et al. (2003) proposed the research onion, many studies (e.g. Sinha et al., 2018; Hayat et al., 2013; Fitrianto, 2020) have utilised it as an effective tool to help the organisation of the research and development of the research design according to the layers of the research onion (Melnikovas, 2018).

## 2.5.1 Positivism (research philosophy)

Positivism is one of the research philosophies (e.g. scientific realism) in Saunders et al.'s (2003) "research onion". They indicate that positivism depends on social forces and the nature that can restrict people's opinions and behaviours to recognize the interpretations constructed by people in society or the individual subjective reality (Saunders et al., 2003).

According to Saunders et al. (2009), research using positivism philosophy means that the work with observable social reality can lead to credible data and the final product of such research can be a derivation of laws or rules (Remenyi et al., 1998).

#### 2.5.2 Research methods

Saunders et al. (2003) indicate that deduction and induction are the main research methods. They suggest that the deductive process is researchers designing a research strategy to verify their hypothesis. The inductive method means that data analysis based on data collection and theoretical development.

#### 2.5.3 Case study (research strategy)

The layer of research strategy involves multiple methods of problem-solving such as case study (Saunders et al., 2003). Yin (2003) proposes that the case study approach refers to the experience that is used in a real environment to investigate current phenomena involving single and multiple case designs.

This study adopted the multiple-case study method given its nature and advantages. First of all, the strategy of case study is broader used to answer questions such as "what", "why", and "how to do" (Saunders et al., 2003, Yin, 2003). Since it can better explain the scope and the process of the study (Morris and Wood, 1991). Secondly, the multiple-case study approach usually provides rich sources in order to make the whole research is more convincing (Yin, 2003). Finally, various items such as literature and comments can be collected as research evidence (Yin, 1989).

## 2.6 Summary

This chapter describes related theory backgrounds, including serious games, the importance of IBL for education, the definition of COI, value creation theory and the relationship between game features and critical reflection. Through reviewing these theory backgrounds, the thesis clarifies what has been done before and which research gaps still need to be addressed.

## 3 Methodology

This chapter clarifies the research approaches. This includes related research process, data capture, data analysis, sample size, calculation of inter-coder consistency and testing the reliability and validity for the model of using role-playing games (RPG) to promote reframing value.

The research onion had been utilised throughout the methodology of this thesis as an effective tool to help the organisation of the research and development of the research design according to the layers of the research onion (i.e. research philosophy, research methods, research strategies, time horizons, and data collection methods).

## 3.1 Exploring serious games supporting inquiry-based learning

Sahay (2016) indicates that when a relationship is being studied, the philosophy of positivism generally is used by researchers. The first step of this study considered positivism as the research philosophy given the aim of the study was to explore the relationship between game characteristics and three core features of IBL. Game features and the three core characteristics of inquiry-based learning (IBL) were inductively extracted. According to Saunders et al. (2003), the research process was shown using the *"Research Onion"*:



Figure 3.1 Methodology of exploring serious games supporting IBL adopted from *Saunders' Research Onion* (2003)

This study conducted a systematic database search using Summon (an integrated search engine that provides access to scholarly material across multiple databases and library collections, including Scopus, PsycARTICLES, sycCRITIQUES, and PsycINFO). Due to the lack of articles in this research field, the author looked for articles published between 2008 and 2021 using "\*nquiry-based learning" and "serious games" as keywords. The search yielded 108 papers, which were then screened by title and abstract to exclude non-pertinent documents.

Criteria for exclusion were: (a) addressing IBL or games separately; (b) presenting games and/or data presented in another publication (in these cases, only one was retained). Most papers did not discuss both aspects of IBL and educational games, and hence 91 papers were excluded. The remaining 17 papers were fully read, and 12 papers were selected for this review.

Data from the papers were extracted using a specially created form involving the method that game features support IBL and related evidence in projects reviewed. Game features were then described and analysed based on the function they had over the three core IBL features. The similar game features found in different games and game-based strategies were grouped in broader categories and general mechanisms supporting IBL were synthesised.

## **3.2** The value creation in communities of inquiry

The second step of this study investigated the relationship that actual values created by communities of inquiry (COI) and weaknesses generated during the process of value creation in COI. Positivism was considered as this research philosophy, as Sahay (2016) indicates that when a relationship is being studied, the philosophy of positivism generally is used by researchers. Activities of COI were collected using the inductive method. The values created by these activities adopted the deductive method that gave research problems and then solved them by analysing activities collected. Based on Saunders et al. (2003), the research process was drawn as follow:



Figure 3.2 Methodology of the value creation in COI adopted from *Saunders' Research Onion* (2003)

#### 3.2.1 Data selection

The literature reviewed came from Summon (an integrated search engine that provides access to scholarly material across multiple databases and library collections, including Scopus, PsycARTICLES, sycCRITIQUES, and PsycINFO). The keywords: ((community) AND ((virtual) OR (online))) AND (((inquiry) AND (COI)) OR ((enquiry) AND (COE))) were used within journal articles and conference proceeding. Because to gain the lastest documents, this research chose papers published within 5 years currently. The articles had to be peer-reviewed, published between 2013 and 2018 and written in English, thus yielding 829 papers. This result was screened based on abstracts and titles to exclude irrelevant 770 articles. 59 papers were fully reviewed, 51 papers were excluded as they did not clarify the activities of communities. Eight papers were finally selected for this review.

#### **3.2.2 Data analysis**

This investigation aimed to find the values in every activity described in COI. Mixed content analysis (deductive and inductive) was conducted according to Miles et al. (2014). The following codebook was edited using NVivo software (v 12 Pro). See Figure 3.3.

Name	Files	References
Chanprasitchai and Khlaisang(2016)	0	
applied value	0	
30 How is it different from my practice or life or background	0	
E. CP	0	
	1	
Resolution	1	
Triggering event	1	
immediate value	0	
potential value	0	
realized value	0	
- C reframing value	0	
Eteokleous and Ktoridou(2012)	0	
Junus et al.,(2017)	0	
Keles. (2018)	0	
Padilla & Kreider (2018)	0	
Pellas & Boumpa (2017)	0	
Setiani and MacKinnon(2015)	0	

## Figure 3.3 Codebook excerpt

## **3.2.3** The process of coding

In the codebook, a theory of five-cycle value creation (immediate, potential, applied, realized and reframing values) was the main category. According to the key questions of main categories, three core presences (TP, SP, and CP) of COI and their indicators were deductively coded as subcategories. For example, coders needed to identify if these

activities of CP associated with the key question "Does the social learning process lead to a reflection on important issues?"

The descriptions of activities and their classification within presences of COI were analysed as manifest and latent contents in all cases. Manifest contents were directly collected as they were already clearly reported in these papers reviewed. For example, Eteokleous and Ktoridou (2012) directly recorded an activity in which students explored, investigated and discussed the concepts of the course. The coding of latent contents for the descriptions of activities (e.g. showing doubts/confusion, identifying and correcting misunderstandings of myself or others) (Junus et al., 2017) was inductively conducted, according to the action verbs described in each paper (e.g. "guide", "encourage" and "convey"). After this, these manifest and latent contents were required coders to specifically classify activities into presences of COI.

The coding schema was developed and refined through preliminary field testing before it was applied. The two coders (one was the thesis author, another was a PhD student) classified the same data units with subsequent comparison of their results. They negotiated with each other before the process of coding in order to ensure the consistency of information of 109 activities provided by them. They agreed that a high level of activity in the network often indicated a sign that something valuable was happening. Hence, if emoticons, photos, stickers and hyperlinks were used very frequently, then they could be considered valuable. In order to calculate the inter-coder reliability of this analysis, Cohen's kappa (90% estimates of the confident interval) statistic was used (Cohen, 1960), obtaining

kappa scores consistently equal to 0.86. This indicated a high level of agreement between coders (McHugh, 2012).
### **3.3 A model of game features promoting reframing value**

The third step of this study aimed to extract game features that promote related value in order to rectify the weaknesses generated during the process of value creation in COI. This relationship between game features and reframing value was investigated, which meant that this research considered positivism was used as the research philosophy. This was consistent with Sahay (2016) who indicates that when a relationship is being studied, the philosophy of positivism generally is used by researchers. Game features were inductively identified and the model of game features promoting reframing value was presented utilising the deductive method. A "*Research Onion*" (Saunders et al., 2003) showed this research process.



Figure 3.4 Methodology of the model adopted from Saunders' Research Onion (2003)

### **3.3.1 Sample selection**

This study took two stages for sample selection. Firstly, RPGs were collected according to a specific ranking list. Secondly, game participants' comments were selected from these RPGs which were identified in the first stage.

RPGs were selected based on China games weight rank (CGWR) (CGWR is constructed by the company of *Sina*, which is the most authoritative, professional and fair game list in China. The list brings all the online game products tested and operated together in Europe, the United States, Japan, Korea and China. It has been divided into six major scores and several options for game information. It creates the most trusted platform for new game recommendations for Chinese players). The ranks of CGWR were published on August 13th, 2019 and the addresses were as follow. See Figure 3.5 and Figure 3.6.

	10S游戏榜 	odifacija filian Schrachta	安本版成目前總	网络排行物	38187 <b>18</b> 9	7973于监网 💿	
	2017 - <b>10</b> 7	hone 🗌 iPad		能入面积479名称。		0.授索	
	他门搜索关键词: 网丝	土廠 文時 初期友	18菜 街竹寺手	学指記版	55.11 T.T.	ALL BA	
Classification of	games		2007 +00 9-000 90	#85 ¥ ▲KXX ##	94 1834 8347	)地和: (木)利	
		INSIDE 例如 - 新聞建設 - 用色設置 - 州王 (Inside) 用雪塘的图风和卷响的影響为人家基 的對位,即且素書和雪成功。进入部成后,就是	大小:1.27 1001 1127 1001 1127 1011 1127 1	[样子说: [編集的] "拯救"的故事,游戏的	から・5 	- 9.0	Stand-alo
Life is strange		奇异人生 #41 - #3988 · 用色粉素 · 文件 (奇界人生)与其完整-位却成不如说是一部 步的处历整个故事情节,而无意识是一边的选择	市政 登録 市設備交互的年齢未約点		(注意市入町市、一)	- 0.9	Stand-alo
Identity 5	3 	第五人格 N近 - 用包約清 - 第第 - 第第 一款1948的形象运动之,近远思维无辜的编档 则需要相互配合新开造能力调正面。1948分别	■ <u>■</u> 週 週代充為7洋洋的原体之		BERMEN.	- 8./	Online
Chinese Palad Sword and Fai		新仙剑奇侠传(单机) 学机 - 角色的词 - 印闪 - 重度 项计逻辑的移植,同时在现在方面针对触得做出	1904 AU	价格:¥60.00 件子数 等级的 D在子机上回味最好的的	NRMOR	e 8.7	Stand-alor
Hunter	r 4	地字猎手4 回题 - 角色画調 - 五作 - 總好杂手 (地宇路手4)是一款名相其实的大作,无论从 来看,想至示了该游戏的传动大气和暗异制作。	(1955) (1953) (1955) (1955) (1953) (1955) (1955) (1953) (1955) (1955) (1953) (1955) (1955) (1953	样子說 [編828] 1成,打击感等明节过是》	9時:是 第二日日日 訪戏的簡件如何等力	- 0.0	Online
		生命线(Lifeline) ●11 · 体用 · 角色版用 · 化用表 2015年6月4月4月5,代入版设置,19435	大小 : 22.7MB 1955 - 1835		60月11日 - 15 1823-1844 1823-1844	cows 8.6	Stand-alor

### Figure 3.5 The rank of role-play games in CGWR for iPhone

6	Mos游戏榜	schillenten flaten Dentranktan	安卓游坦时行使 网络排行使 页游排行使 97973	F3000 (6)
		hone	输入查询APP名称_	Q授業
Classification of	games		日本         日本         手板公式         第三部         第二部         第二         第二部         第二         第二         第二         第二         第二         第二         第二         第二         第二         第二	215 5
	7	INSIDE 新述 - 新述語句 · 角色和两 - 建築 (neide) 用型癌的面积和透晶的副體为大家語 的制位,供用氧用用面成的、进入超来后,研究	RPG         星西内町:西           御玩         道道         単子鉄         展成的         第2016年1月           第二十分         第二十分         第二十分         第二十分         第二十分           第二十分         第二十分         第二十分         第二十分         第二十分	Stand-alone
Life is strange			大小:936 MI5 价值:\$18.00 是否內容:否 一些批 查班 样子语 偏低的 的资格意味和 和历程确认可的电影未的变化应该,在这次中也读得到主角带入剧情。一步一 都是到后信产生一些不同剧影响、现实生活与部项。	Stand-alone
Identity 5	3		大小:661.7 MB 价值:免费 量而内码:西 一般玩 担迫 样子族 腐蚀的 新家地家礼包 而风玩酒了深深的惊嘆之味。显觉者在游戏中还在用自物吗可加减,求生者 满度哲慧是走走会行,多样说的话景口动与技。	Online
Chinese Palad Sword and Fai		新仙剑奇侠传(单机) 平民 - 和四日期 - 时间 - 王氏 政计原味的移植,同时在操作力面积对触界和3	大小:148 G8 价值:¥69.00 是5内码:是 世代 型组 样子说 有我的 新放地的机能 出了优化。能够让他的明丝们在于机上回转最初的感动。	Stand-alone
Hunter	5		大小:950MB 价值: 8 世 世活内码: 是 世玩 查班 种子族 察说的 医动脉的 和 商品 音乐 场景 技能地位、打击信等相可这是从游戏的能体实响等方面 加之引入入胆的副情,他对是一款富获治作者;29-	CGWS 8.6 2001Le
			大小: 22.7M5 / 价钱: ¥1.00 / 是否内约: 合 / 把助 _ 应应 _ 样子进 _ 程格的	<b>8.6</b> Stand-alone

Figure 3.6 The rank of role-play games in CGWR for iPad

(http://ios.97973.com/ios/hotlist?f=0\_11\_0\_0\_0&f2=0\_0\_11\_0\_0&platform=ipad)

The sample of RPGs was selected according to the following criteria: (1) to choose the game genre of RPG; (2) to choose the Top5 games in the grossing rank for iPhone operation system (IOS) (e.g. iPhone or iPad); (3) the Android market was excluded because it has not a uniformed and authoritative ranking platform in China.

The comments were collected from the official website of the RPGs or the forum of Steam (Steam is one of the largest platforms which integrate digital products in the world. It enables players to purchase, download, discuss, upload and share games and software). The following criteria were used to select comments: (1) the theme of the comment was the unit of the sample; (2) comments came from Steam. However, players did not post comments in Steam that is why coders selected these comments from the official website. If the comments of the game were not to be found both in Steam and the official website, the game (e.g. *Hunter4*) was excluded; (3) the timescale for the comments was 6 months which started from the first theme posted in the forum of the game. For example, the first comment posted in the forum of *Lifeline* was March 9th, 2017, so the time was selected from March 9th, 2017 to September 9th, 2017; (4) due to different contents being posted for the same theme or different themes by the same participant, coders counted the number of comments and themes rather than counted the players' number. Finally, 1020 themes of players' comments were collected in total. See table 3.1.

The top 5 games	The time	The number	The number of
	scale	of the themes	players' comments
1. Inside			
This is an RPG game that focuses on solving			
puzzles on a black and white game screen. The			
player controls the little boy to walk, run,	12/06/2017-	45	2391
swim, climb, and use objects. When the player	12/12/2017	45	2391
fails to quickly solve the puzzles (e.g. because			
of drowning underwater for too long) the boy			
controlled by the player will die.			
2. Life is strange			
The protagonist of the game is Max. She finds			
that she has the ability to return to the past.			
This leads to a butterfly effect whenever she			
uses this ability. The game storyline is that	30/01/2015-		
Max foresees a storm will come, she must take	31/07/2015	28	879
responsibility to protect the town. Players'	01/0//2010		
actions will affect the development of the			
story. The game allows and encourages players			
to use Max's ability (i.e. manipulating time) to			
reshape the plot.			
3. Identity5			
This is a massively multiplayer online role-	09/07/2018-	927	2026
playing game. The players control different game avatars and generate interaction in the	09/01/2019	827	2036
game. 4. Chinese Paladin: Sword and Fairy			
The game is designed based on incorporating			
elements of martial, magic and fantasy.	19/09/2017-		
According to the setting of the storyline, the	19/03/2018	98	523
player controls the characters to complete	17,00,2010		
diverse tasks (e.g. finding lost items).			
5. Lifeline			
As the spacecraft crashed, astronaut Taylor			
was trapped on a strange planet. The players	00/02/2017		
help Taylor solve various problems using real-	09/03/2017-	22	111
time communication with him in the game	09/09/2017		
(e.g. messages). And players can choose the			
branch of the plot to guide Taylor's adventure.			

# Table 3.1: The table for the details of participants' comments

### 3.3.2 Data analysis

This study used a mixed content analysis (quantitative and qualitative) to analyse 5940 comments of players. These comments were coded with the NVivo software (v 12 Pro), full codebook available upon request from the corresponding author. See Figure 3.7.

top 5	games 🖲 Search Project		~	Otasks for problem-solving ×
۲	Name	/ Files	References	∃ ▼
• 0	Generating new assessments for other people	0	0	
• 0	Promoting reflection	1	1	Reference 2 - 0.47% Coverage
• 0	Suggestions for community development	2	2	
• 0	The change of evaluation	0	0	1 : 327,355 - 662,411
	O game avatars	20	25	<files\\identity &="" &<="" -="" 09072018-09012019\\bugs="" 2="" 5="" bugs="" is="" page="" problems\\camping="" td="" unbalanced2=""></files\\identity>
	O game tools	10	14	Problems - IdentityV> - § 2 references coded [12.09% Coverage]
	O tasks for problem-solving	32	82	Reference 1 - 3.78% Coverage
				Just bait the hunter to hit the survivor on chair or use coordinator, camping is not a superior stratagy, ther is always room to improve, learn to kite properly, use the pallates effectively, know some small in game mechanics
				Reference 2 - 8.31% Coverage
				I think most of people complaining about camping not play with hunter don't get me wrong i said the same things before i start to play with hunter and with some friends that i meet in the game, my advice is for you, is to start to learn how to play with some hunter, then you will see that survivor are

Figure 3.7 Codebook excerpt (the number of the themes was collected in Files; the number of players' comments was collected in References)

The four criteria of reframing value (e.g. promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people) were coded as the main category of the codebook. Coders needed to extract game features and identify if the game features associated with these criteria according to the comments which involved evidence of reframing.

The two coders (one was the thesis author, another was a PhD student) classified the same data units with subsequent comparison of their results. Specifically, two coders independently coded to extract reframing content in all comments, based on whether the comments provide evidence of the reframing process firstly. The data included (1) evidence of effected reframing (players directly posted their opinions without discussion with other players); (2) evidence of stimulated reframing (players were triggered by other players' comments to present their ideas).

Secondly, these reframing comments were classified according to the criteria of reframing value. Because the reframing value was evaluated in RPGs, some potential descriptions were considered based on the assessment criteria of reframing value (Wenger et al. 2011): (1) promoting reflection -rethinking game content and values through reflection; (2) the change of evaluation -changing the gameplay processes and patterns/presenting new game strategies that help players to win or solve problems; (3) generating new assessments for other people -proposing the assessment standard (e.g. players directly posted their opinions regarding characters' behaviours; players changed their opinions regarding characters' behaviours; players changed their opinions regarding self, teammates, enemies and other players; (4) suggestions for community development-reshaping the new expectation of the whole game development, which means that game players provide suggestions (e.g. revising game design/errors/modes) to game developers.

Finally, game features were extracted qualitatively from the comments which involved evidence of reframing. The two coders through negotiation before the process of coding confirmed the consistency of information. For example, they agreed that players proposed game strategies that involved suggestions to game developers or other players. According to this, coders needed to negotiate whether these game strategies (e.g. the application of the game tools, revising game time) were proposed regarding game developers or other players. This study obtained a high level of agreement (kappa=0.85) between coders (McHugh, 2012) utilising Cohen's (1960) kappa statistic ( 90% estimates of the confident interval).

The model used a specially created form that described strategies of game features promoting the four evaluative criteria of reframing value (e.g.*promoting reflection*, *suggestions for community development, the change of evaluation* and *generating new assessments for other people*).

### 3.3.3 An instrument for evaluating the reliability of the model

Based on Saunders et al. (2009), research using positivism philosophy means that the work with observable social reality can lead to credible data. This research considered positivism as research philosophy given the aim was to measure the reliability of the model (i.e. credible data) through analysing the questionnaire completed by related experts according to their game-teaching experience (i.e. the work with observable social reality). This process of measurement used the deductive method that the high score evaluated by 39 experts confirmed the model was reliable.

According to Saunders et al.'s (2003) Research Onion, the research process was designed.



Figure 3.8 Methodology for measuring the reliability of the model adopted from *Saunders' Research Onion* (2003)

An instrument was developed to refine the model of using game features promoting reframing value and capture experts' perceptions for the relationship between game features and criteria of reframing value. This instrument was designed according to the method of Moore and Benbasat (1991) since they propose that an instrument can measure diverse perceptions based on an individual adopts innovation of information technology (IT). This instrument consists of three stages: generation of the instrument items; categories for the organisation of items; instrument testing.

The aim of the first stage was to confirm the clarity of game features as created items. Two game-analysis experts who were familiar with the theoretical background of games assessed these definitions of game features. They conceptually pilot tested these game features with kinds of RPGs in order to refine the definitions of game features. For example,

"attractive" is a highly subjective word, the concept of "attractive game stories" was not objective. The experts suggested that when defining game features should be as objective as possible. After this, the author defined the game feature as "game stories" in order to exclude the ambiguity of the game feature.

The objective of the second stage was to classify the game features confirmed. The four criteria of reframing value were instrument scales in order to evaluate the game features: "if players rethink the game content to lead to a reflection on important issues or change people's understanding for important things", "if players reframe the assessment standard for other people", "if players reshape the new expectation of the whole community development" and "if players change the process and pattern of evaluation via the game experience".

The third stage was to reword 10 game features according to instrument scales to a questionnaire form for evaluation. For example, in the criteria for "change of evaluation," it is investigated "if players change the process and pattern of evaluation via the game experience" by: (1) game avatars; (2) the game design of particular style; (3) tasks for problem-solving; (4) game tools. The promotion of reframing value was assessed via a single item on a five-point liker scale (1: "strongly disagree; 5: "strongly agree").

The expert evaluation was conducted to assess the reliability of the analysis instrument. According to Gay and Diehl (1992), at least 30 subjects are required as an acceptable number of respondents in correlational research. So 39 game-teaching experts coming from China participated in the test independently. All testers were familiar with critical reflection theory and the underpinning model. These experts according to their game -teaching experience completed the online questionnaire. The full Chinese version is available from the address: https://www.wjx.cn/jq/54588262.aspx. The English version is listed in Appendix C.1.

Cronbach's alpha (Cronbach, 1951, Eisinga et al., 2013) was used to measure the reliability of the questionnaire, obtaining Cronbach  $\alpha$ =0.940. This meant that the reliability was high (Eisinga et al., 2013).

### 3.3.4 The validity of the model

The following "Research Onion" showed the research process.



# Figure 3.9 Methodology for measuring the validity of the model adopted from *Saunders' Research Onion* (2003)

Based on Saunders et al. (2009), research using positivism philosophy means that the work with observable social reality can lead to credible data. This research considered positivism as research philosophy given the aim was to measure the validity of the model (i.e. credible data) through analysing related questionnaires completed by the testers after playing a specific game (i.e. the work with observable social reality).

### 3.3.4.1 Game sample

The three RPGs (*Chinese Paladin: Sword and Fairy 3, Limbo* and *World of Warcraft*) were collected to conduct this investigation, given the nature of these games and their popularity.

*Chinese Paladin: Sword and Fairy 3* is a stand-alone RPG game developed by *Dayu* Company. It was published in China in 2003. This game won the most popular RPG game in China voted by players. The game story is designed based on Chinese martial arts. Players control the character to attend events and fight monsters, completing a series of tasks.

*Limbo* is an RPG game developed by *Playdead Studios*. So far, the game has won more than 100 game awards, such as the best download game (*GameInformer*). The player controls a little boy in the game to seek peace on the edge of hell.

*World of Warcraft* is a multiplayer online RPG game produced by *Blizzard* Company. According to the statistics of China's office web in January 2020, about 900,000 people log in to the game in a single day. The plot of the real-time strategy game "*Warcraft*" was

utilised as the historical background of this game. Players play a specific character (i.e. Magician and elf) to explore, adventure, and complete tasks in this virtual world.

### 3.3.4.2 Sample size and participants

The sample size of factor analysis was determined according to Gorsuch's (1983) suggestion that the required number of participants is obtained using the total numbers of items being multiplied to 5. This method which uses a subject-to-item ratio of 5:1 to construct validity in the sample size is preferred by many studies (e.g. Alsubaie et al.,2020; Bunz et al.,2020; Diao et al.,2020).

The testers (106) coming from China were divided into three groups (34 participants evaluated 5 items in *Chinese Paladin: Sword and Fairy 3*, 37 participants evaluated 7 items in *Limbo* and 35 participants evaluated 5 items in *World of Warcraft*). Three tutors managed testers' participation in order to ensure testers completing related questionnaires after playing a specific game for three days. The three groups were composed respectively of 19 teachers and 15 students, 20 teachers and 17 students, 18 teachers and 17 students. These testers were selected randomly, some usually played digital games and some did not play.

The questionnaires included these items: (1) participants' status (teacher or student). (2) If the related game features were covered by the game testing. (3) 17 items were utilised involving four dimensions, relational (i.e. "I thought that the rank system of the game was able to support multiple players at the same time"), affective (i.e. "I felt a sense of responsibility in controlling roles' destiny while working on the options of multipleendings of the game"), cognitive (i.e. "I felt that game tools allowed me to complete my task in several new strategies of matching") and active (i.e. "I believed that I was the game avatar I was controlling"). (4) What the extent to game features supporting criterion of reframing value is. This used a five-point liker scale (1: "strongly disagree; 5: "strongly agree") and there is an "other" point that needs evaluators to provide their other attitudes to explain how the specific feature could be improved. The full Chinese version is available from the address: https://www.wjx.cn/report/63422586.aspx. The English vision is listed in Appendix C.2.

This evaluation used Cronbach's alpha (Cronbach, 1951) to measure the reliability of three questionnaires, Cronbach α equals to 0.892 (5 items, *Chinese Paladin: Sword and Fairy 3*), 0.919 (7 items, *Limbo*) and 0.905 (5 items, *World of Warcraft*), respectively. The reliability of the three questionnaires was robust (Eisinga et al., 2013).

Furthermore, due to the difference in the testers' status (e.g. teacher and student), this thesis author considered that assess differences between teachers and students mentioned with respect to promoting reflection. the Multifactor Analysis of Variance (ANOVA) was used to measure the major influence across variables between teachers and students on the consent of game features.

### 3.4 Summary

This chapter describes the research methods. All studies used positivism as the research philosophy because research data could be obtained by observable social reality (i.e. IBL-based games, activities of COI).

For exploring serious games supporting IBL, this step presents a literature review that aims at determining the game features utilised in the game and the gamification strategy designed to promote IBL and their effects on students' achievement. Through a systematic database search, 12 studies were included in the review. An inductive method was utilised to extract game features of IBL-based games.

The second step of this research analysed eight papers to review activities of COI as the key link which connects COI and value creation. This investigation used a mixed-method (inductive and deductive) to analyse data.

The third step of this research proposes a model to clarify the relationship between digital RPGs and reframing value. According to 5940 comments of players, coders constructed the coding book and inductively extracted RPG game features. After this, questionnaires were used as investigation tools to test the reliability and validity of the model of game features promoting reframing value. The deductive method was utilised in the process of measuring reliability and validity, respectively.

### **4** Research results

This chapter proposes related research results such as an overview of characteristics of a game-based inquiry-based learning strategy; the actual values that communities of inquiry (COI) can contribute to improving learning experiences in virtual environments; game features that may influence reframing value have been extracted.

These research results are given according to related research questions mentioned before one by one.

### 4.1 Exploring serious games supporting inquiry-based learning

# 4.1.1 What are the characteristics of a game-based inquiry-based learning strategy?

The study found that serious games reported by 12 papers had some certain similarities. For example, all of the papers discussed the participants' characteristics, most papers mentioned the arrangement of game time and the type of game. Therefore, the participants in games, playing game time and the game genres as three characteristics of game-based inquiry-based learning strategy were proposed in Table 2.3.

Most studies (n=9) concerned with the use of serious games support inquiry-based learning (IBL) of students. However, only a few studies (n=3) reported that IBL serious games promoted teacher training.

Regarding this issue, this has a work expanded in Appendix A that from the perspective of empirical studies, explore the feasibility of serious games applied supporting teachers' professional development.

Most articles reviewed indicated that students involved aged six to eighteen including primary schools and secondary schools. Only one paper discussed the field of higher education. See Table 4.1.

	Study	Characteristics of game-based IBL strategy			
		Participants in games	Playing game time	Game genres	
1	Hwang, Chiu, and Chen (2015) Lameras and	P, 6th-grade students	120 minutes in classroom	RPG	
2	Moumoutzis (2015)	P,S,H	out of classroom	not reported	
3	Hwang and Chen (2017)	P, 6th-grade students	3 weeks for 4 periods per week in classroom	adventure game	
4	Kennedy-Clark, Galstaun, and Anderson(2013)	т	2 hours in classroom	RPG	
5	Kalz M et al. (2014)	S, 8th grade students	5 weeks for 2 and 4 hours a week in classroom	not reported	
6	Meesuk and Srisawasdi (2014)	S, Study 1: 12th- grade students P, Study 2: 10th- grade students	study1 20 minutes; study2 not reported in classroom	shooting game	
7	Sabthisin et al. (2012)	S, 8th grade students	55 minutes in classroom	RPG	
8	Chee et al. (2011)	S	twice a week during 4 weeks and 6 sessions	RPG	

Table 4.1: Characteristics of a game-based IBL strategy

	Study	Characte	ristics of game-based IBL	strategy		
		Participants in games	Playing game time	Game genres		
			for each session 120 minutes in classroom			
9	Lameras et al. (2014)	Т	not reported	RPG		
10	Firssova et al. (2014)	S, 8th grade students	5 weeks for 2 and 4 hthiss per week in classroom	not reported		
11	Dorji, Panjaburee and Srisawasdi (2015)	S, 10th grade students	not reported	strategy game		
12	Okada et al. (2019)	т	not reported	Augmented reality (AR)		
P: p	P: primary, S: secondary, T: teacher, H: higher education					

This research found that there was mixed evidence for the arrangement of playing game time. Some were periodical, which was consistent with the stage structure of IBL, others were arranged as an intact time. For instance, Kalz M et al. (2014) arranged the playing time into five weeks between two to four hours per week. When the first week ended, learners just completed a part process of IBL. Sabthisin et al. (2012) suggested that the playing game time was 55 minutes. Most studies (n=8) showed that the time was spent in a formal classroom rather than out of the classroom.

A variety of game genres were used including strategy, role-playing, and shooting (Wolf, 2001), but RPG was discussed in most studies (n=4) reviewed.

#### 4.1.2 What game features have been used to support IBL?

This analysis identified thirteen game features supporting different aspects of IBL (Table 2.4). Half of the games and game-like activities included six or more features. The lowest number of game features was found in Lameras and Moumoutzis (2015) and Kalz et al. (2014), which included two features each, while the highest number of features supporting IBL was present in the games by Hwang, Chiu and Cheng (2015) (nine features), Hwang and Chen (2017) and Sabthisin et al (2012) (eight features each).

The most used game features found in the different games and game-based activities were the use of a game context mirroring real-world contexts (n=9), and coherent game environment with specific rules, artefacts, potential goals and action possibilities (n=8). The least used game features were contextualised feedback (n=1), task assessment mechanics (n=1), and non-linear exploration of information (n=3).

### 4.1.3 What aspects of IBL are being supported by game features?

Most game features found in this review were oriented at supporting one (n=5) or two (n=6) core aspects of IBL. For example, game tips and scoring systems were aimed primarily at providing guidance, while interactive artefacts and contextualised feedback were used to support guidance and the creation of a meaningful environment. Only two game features, namely storyline and structured game tasks, were used to support the three core characteristics of IBL.

#### 4.1.3.1 Game features

(1) **Game tips** (Hints given by the game at certain points which like some dialogues with virtual characters). For example, energy-consuming graphs show the ratio of the energy consumption of various household appliances and the formula "your saving=income-cost of energy" is used to prompt students (Dorji et al., 2015).

(2) Scoring system (The game report students' scores or achievement, which students can check). For example, students choosing the way to save money can decide their score to be increase or decrease, which stimulates their inquiry by the feedback of the scores (Hwang et al., 2015).

(3) Storyline (Central compelling storyline linking several gaming tasks which may coexist with specific inner stories for separate tasks). For example, the outer storyline is to participate in a competition for winning a villa. Students are guided to participate in three gaming steps, where they follow the gaming inner storylines, to learn the relationships between the prizes and risks of different ways of investment (Hwang et al., 2015).

(4) Interactive artefacts (Different materials, icons, and objects in the environment with which the player can perform actions). For example, the blue shield provides hints as to what to look for or where to look for information (Hwang and Chen, 2017).

(5) Structured game tasks (Series of goal-oriented tasks presenting a structured organisation of steps to complete them). For example, one of six tasks provides sea

knowledge to students' learning, where is designed involving some questions and prompts of real sea creatures (Hwang and Chen, 2017).

(6) **Contextualised feedback** (Information given to the player regarding his/her performance with is meaningful to gameplay or game context). For example, in a microbiology scenario, a camp nurse reviews the diagnosis worksheet elaborated by students (Sabourin et al., 2012).

(7) Nonlinear exploration of information (Completion of tasks is done through iterative and non-consecutive processes). For example, students can gather data using different strategies and methods that serve to fill in, correct and update information in diagnosis worksheets and laboratory tests (Sabourin et al., 2012).

(8) Gameplay-context coherence (A real-like or fictitious game environment with specific rules, related artefacts and action possibilities, which convey internally coherent meanings). For example, a virtual medieval background is built in a game to help learners conduct detective activity (Kennedy-Clark et al., 2013).

(9) Task assessment mechanics (Evaluation system is provided for students to test skills and/or knowledge understanding). For example, two tests were used in the game to evaluate the students' achieving process and assist to review previous learning concepts (Hwang and Chen, 2017).

(10) Multiple weapons (Choice among weapons to attack monsters or produce specific impacts or results). For example, students' weapons have different ammunitions to help them escape from the lab (San Chee et al., 2011).

(11) Several gaming contexts (Different scenes and situations where students' actions take place). For example, students can shift between two gaming contexts of real school and virtual scenario via swiping their QR code (Hwang and Chen, 2017).

(12) Game context mirroring real-world contexts (Situations in the game simulate the real world). For example, piggy bank and Financial Institution are designed in the game to show the real world (Hwang et al., 2015).

(13) Student assumes a role coherent with the game context (Students control player token through role-playing). For example, the student plays the role of a young man who needs to choose among different ways to invest his family's money, exploring different types of financial investments (Hwang et al., 2015).

### 4.1.3.2 Game features support the core IBL characteristics

IBL-supporting functions of each game feature are described in the following figures.



Figure 4.1 Game features supporting one core IBL characteristic



Figure 4.2 Game features supporting two core IBL characteristics



Figure 4.3 Game features supporting three core IBL characteristics

### 4.1.4 What are the reported impacts on students?

Only seven studies measured impacts on students. Positive impacts on achievement were reported in all of them, either in terms of conceptual learning or information-seeking strategies. For example, Kalz et al. (2014) found significant knowledge improvement in secondary students who used an IBL-based game to learn about energy consumption. Sabthisin et al. (2012) observed that 8<sup>th</sup>-grade students displayed more effective problem-solving strategies while playing a microbiology IBL-supporting game, although knowledge improvement was correlated to data collection behaviours performed before playing. Six of the seven studies measuring knowledge improvements found significant impacts compared to pre-tests or control groups in school-age students.

Mixed evidence was found regarding the impacts of IBL supporting games and game-like strategies on engagement. Hwang and Chen (2017), Hwang, Chiu, and Chen (2015) and Meesuk et al. (2014) found medium-size impacts on intrinsic motivation when students played IBL-supporting games, but Kalz et al. (2014) and Firssova et al. (2014) observed no significant effects on triggering or maintaining motivation. See Table 4.2.

Study	impact: engagement	impact: achievement	impact: other	impact on
	How did using games benefit learners' motivation	How did using games benefit learners' achievement	which others benefit games generate	- UII
Hwang, Chiu, and Chen (2015)	* Small effects of computer games IBL on motivation	* Medium effects in financial knowledge in comparison to the group with conventional IBL activities (non -computer games IBL)* Large effects in students with an active learning style	*Medium effects on flow in comparison to the group with conventional IBL activities (non- computer games IBL)*Small effect on satisfaction in comparison to conventional IBL activities	6th- grade students
Hwang and Chen (2017)	* Medium size effects regarding intrinsic motivation for learning of control group (CG); no difference in extrinsic motivation between groups	*Statistically significant results. The medium effects in students' performance in the game group than students in the traditional inquiry group (eta2= 0.110, corresponding to a medium effect size)	*Students explore more high- frequency activities of seeking help in games than in conventional IBL and promote participation in learning activities. Experimental group (EG) students performed more help- seeking actions than CG* EG students performed more field observation, comparison and data search behaviours* EG High relationship between the provision of clues and field observation* EG students prefer to look more information about the task after the solution was submitted (probably revealing a tendency to confirm	6th- grade students

# Table 4.2: The impacts of serious games used on IBL

Study	impact:	impact:	impact:	impact
	engagement	achievement	other	on
	How did using games benefit learners' motivation	How did using games benefit learners' achievement	which others benefit games	
	learners mouvation	acmevement	generate success in the solution)* CG	
			performed read more supplementary material to find a solution* EG	
			interacted more with the real world to	
			find a solution and assess success* CG presented more Trial-error behaviours	
Kennedy-Clark, Galstaun , and Anderson (2013)	not reported	*Participants were fairly consistent between the pre-and post-tests for an understanding of these terms: gameplay, understanding of serious games and the virtual world	*Significant shifts of teachers' response from negative in the pre-test to positive in post-test, including the use of information and communication technologies (ICT), is enhanced the change of teachers' attitudes toward games, and teachers' selection for games in the classroom*For current game was used and preferred games, the result showed that men played the most common games which are first- person shooting games (FPS) and RPGs, However, females played phone applications*The chi-square test revealed that between gender and frequency of playing the game there	teachers

Study	impact:	impact:	impact:	impact
Study	engagement	achievement	other	on
	How did using games benefit	How did using games benefit learners'	which others benefit games	
	learners' motivation	achievement	generate	
			was no statistically significant correlation ( $\chi^2 = 7.744$ , p = .021)	
Kalz M et al. (2014)	*Did not impact on using games to inquiry motivate students to be interested in topics that they are not interested in	*Significant effects in using games promote knowledge acquisition*Students with low prior knowledge benefitted the greatest. However, effect sizes seem small (not calculated)	not reported	8th grade students
Meesuk and Srisawasdi (2014)	*Students' science motivation in SAGOI class is more change than their SAGOI learning experience* Statistically significant results (the data of eta2 correspond to a medium effect size).*Medium effects in intrinsic motivation (eta2= 0.10), self-determination (eta2= 0.25), self-efficacy(eta2=0.21), and career motivation(eta2=0.15) impact on the increase of science motivation, which is different from the control group and the experimental group participating with the learning instruction	not reported	*Research 1 points out that using the IE War game have effects in influencing students' flow, satisfaction and enjoyment. However, effect sizes are not calculated	Study 1 12th grade students Study 2 10th- grade students

Study	impact: engagement	impact: achievement	impact: other	impact on
	How did using games benefit learners' motivation	How did using games benefit learners' achievement	which others benefit games generate	
Sabthisin et al. (2012)	not reported	*The increased data collection behaviour prior to attempting to resolve a problem is associated with a more effective problem resolution. Students spend less time and have fewer total attempts than students who did not participate in information gathering before solving the problem*The behaviour of collecting information before the problem is solved, which will not lead to better learning benefits for students	not reported	8th grade students
Chee et al. (2011)	not reported	*The summary chemical separation task of the game group is superior to the control group in separation effect measurement and chemical concept understanding	not reported	students
Firssova et al (2014)	The use of games does not consistently maintain student motivation	not reported	*Using the scenario of the game make students experience the "usefulness" of game tasks	8th grade students
Dorji,Panjaburee and Srisawasdi (2015)	not reported	*Significant effects in promoting the learning performance of conceptual knowledge through play RES-battle (eta2=13.887, corresponding to a significant effect size)*Medium effects in learning gain of students (0.3 <eta2=0.47<0.7, corresponding to a medium effect size)*</eta2=0.47<0.7, 	not reported	10th grade students

Study	impact:	impact:	impact:	impact
Study	engagement	achievement	other	on
	How did using games benefit	How did using games benefit learners'	which others benefit games	
	learners' motivation	achievement	generate	
		having a high degree of recognition and		
		satisfaction using RES-battle serious games		
		to improve their conceptual knowledge		

### 4.2 The value creation in communities of inquiry

The eight papers discussed activities of COI improving the development of learning in medical education, teachers' skills, course practising, civic education, learning strategy and semi-synchronous language exchange. Four papers particularly described that the contexts of activities were formal learning environments (see Table 4.3).

Study	Activities of COI	Setting where activities were accessible
Wang, Y. et al.(2016)	Using WeChat as a communication tool of COI to study two languages (Mandarin Chinese and English)	not reported
Chanprasitchai and Khlaisang (2016)	Solving traditional medicine problems	not reported
Eteokleous and Ktoridou (2012)	Using Google applications to study systems of management information	formal (classroom)
Junus et al. (2017)	Interacting and communicating to practice students' learning strategies	not reported
Keles (2018)	Using Facebook completed community services practices	formal (classroom)
Padilla and Kreider (2018)	Studying clinical practice management	formal (classroom)
Pellas and Boumpa (2017)	Interacting and communicating to study a foreign language	not reported
Setiani and MacKinnon (2015)	Studying civic education course	formal (classroom)

### **Table 4.3: Activities of COI**

These activities mainly oriented to sharing documents and information, learning the experience of each other, guiding, helping and supporting each other, learning knowledge together, inspiring changes, and providing new types of development chances. For example,

Eteokleous and Ktoridou (2012) indicate that teachers provide convenient and direct guidance to set up communication and interaction between teachers and students.

## 4.2.1 The actual values that COIs can contribute to improving learning experiences in virtual environments (answering research question RQ1)

This review found that immediate, potential, applied, realized and reframing values were promoted by the activities of COI. Most activities generated three or more types of value. The lowest number of types of the value was found in Keles (2018) (two types of value), while the highest number of types of the value was presented in the activities by Chanprasitchai and Khlaisang (2016) (five types).

In particular, immediate value and potential value were frequently promoted with positive outcomes. During students' exploration, for example, they would be likely to gain new knowledge, based on evaluating the topic, planning their methods of learning and constructing meaning. This leads to promoting potential value (Padilla and Kreider, 2018).

However, only one activity that promoted reframing value was reported in these papers reviewed. Chanprasitchai and Khlaisang (2016) indicate that students positively discuss experts' advice for further clarification to propose solutions for clinical problems, which could trigger students' reflection with connecting their ideas.

Furthermore, some similar types of activities generated the same value. For example, teachers check students' WeChat messages every day and replied on time (Wang et al., 2016). Teachers provide links to learning resources and gave feedback (Junus et al., 2017).

These activities generate similar interactions (e.g. teachers' help and feedback for students) to promote immediate value.

The same activity helped to generate more than one value. For instance, students could freely explain their opinions and openly communicate with peers using their virtual avatars or interactive text or voice communication (Chanprasitchai and Khlaisang, 2016). For this activity, students interacted with each other, which not only create immediate value but also likely to gain something new (e.g. strategies, knowledge) from other persons through their communication to create potential value.

Some unappropriated activities were likely to lead to passive effects. Keles (2018) describes that during the implementation process of COI, some activities did not create value and even brought negative effects. For example, some participants very rarely show their feelings of anger or shyness and never share content with the Facebook group, which is not useful for communication and interaction. The quantity of content was more focused on by participants than the quality of content and use the method of 'copy and paste' posts to increase the number of comments. This is not beneficial for sharing information and also wastes teachers' time.

# 4.2.2 TP, SP and CP represent COI aspects determining the value generation process (answering RQ2)

This review indicated that most activities of TP promoted immediate value, especially its categories of promoting discourse (n=22) and direct instruction (n=16). For instance, to shape constructive communication (promoting discourse), from different perspectives,

teachers initiate and guide the weekly topics to promote the discussion of the topics and ensure the continuity and frequency of the comments within groups (Keles, 2018). This is likely to attract students' interest so that promote immediate value.

Most activities of SP promoted immediate and potential values. Open communication (n=13) positively generated immediate value. This is consistent with Wenger et al. (2011) who propose that interactions or communications can generate immediate value in online communities. For example, the students used the online tools to express their opinions freely, participate in the course discussion and frequently communicate with their peers or other participants (Eteokleous and Ktoridou, 2012).

The activities of CP promoted three or more values. For example, the resolution of CP helped to promote potential, applied and realized values. Junus et al.(2017) discuss that Identify owns or others' misconception and correct them. These activities could improve the ability to achieve important aims for self or other stakeholders. Hence, it promoted realized value. However, some values were significantly promoted by specific activities, including triggering event (n=5) promoting immediate value, exploration (n=9) supporting potential value, integration (n=4) and resolution (n=4) generating applied value. See Table 4.4.
		Numbers of activities				
presences	categories	immediate value	potential value	applied value	realized value	reframing value
	Design and organization	4	1	0	0	0
ТР	Promoting discthisse	22	0	0	0	0
	Direct instruction	16	0	0	0	0
SP	Individual/affective	7	2	0	0	0
	Open communication	13	3	0	0	0
	Group cohesion	9	4	0	0	0
	Triggering event	5	2	0	0	0
СР	Exploration	4	9	6	2	0
	Integration	2	1	4	1	1
	Resolution	0	2	4	1	0

# Table 4.4: The statistics form that the appearing numbers of activities in three presences

#### 4.3 A model of game features promoting reframing value

#### 4.3.1 Which game features may influence reframing?

Ten game features were identified including multiple endings of the game, tasks with emotional experience, a real no-play character (NPC) image, the game design of particular style, game stories, game avatars, tasks for problem-solving, the rank system of the game, game tools and game mode.

In order to extract game features, an inductive method was used to analyse the comments which were selected based on evidence of reframing. For example, "game story" was extracted as a game feature in light of these comments: "......While reading the journal entries of the survivor who is stuck in the manor and witnessed someone leaving, it describes that on a certain day after the rain they receive a note to go to the dining room......" (*Identity5*); ".....also in Before the storm we can see that Max was very worried about parting with Chloe and they promised to call each other.....all the more after the death of Chloe (leading character in the game)'s father, Max knew how bad Chloe feels......" (*Life is strange*).

# 4.3.2 Which evaluative criteria of reframing value are affected by game features?4.3.2.1 The relationship between game features and criteria of reframing value(1) Multiple endings of the game

Multiple endings of the game allow players to choose what they want to get and why. Players presented their opinions according to different endings, stimulating more discussions and reflecting on themselves. For example, players would consider the advantages and disadvantages of the different endings to decide which one was they want. The game *Life is strange* has two different endings, involving protecting the town Arcadia Bay or rescuing Chloe's life. Players post and discuss their comments to explain the reasons for choosing one option.

Multiple endings of the game support promoting reflection.

i) *Inside* (Stand-alone): when players experienced one of the endings, they reflected through discussion on the meanings that antagonistic characters (e.g. the dogs, the masks and the mermaid) appeared in the game.

ii) *Life is strange* (Stand-alone): the different endings of Chloe' life inspired the reflection of the player who decides which game ending is better for Chloe.

#### (2) Tasks with emotional experience

When players in the game complete the tasks involving emotional experience such as friendship, love, betrayal, and adventure in a constantly changing virtual world, they are likely to get emotional satisfaction to trigger rethinking. For example, in the game *Life is strange*, a player presents comments to discuss the friendship of the two characters (Max and Chloe). This leads to players' various ideas. For example, players suggest that people tend to forget childhood friends and move on with their lives or others suggest that children's world cannot be understood by adults.

Tasks with emotional experience support promoting reflection.

i) *Inside* (Stand-alone): the task inspired the player's empathy to rethink the reason that causes the chickens to starve to death.

ii) *Life is strange* (Stand-alone): the experience of the player's love in the task triggered his rethinking for love with his personal experience in the real world.

*Identity 5* (Online): the player could not rescue his teammate in a task, which made him feel bad and presented his reflection about the importance of friendship.

#### (3) A real NPC image

When players interact with an NPC who is close to a real social character, this will trigger players to evaluate the virtual NPC. For instance, David (NPC) is carved by the game (*Life is strange*) as a retired soldier. Through understanding his background, the players assess his personality and consider David can be a good step-dad for Chloe if he learns to put his ego aside and genuinely care about her as a person.

A real NPC image support promoting reflection.

i) *Life is strange* (Stand-alone): the players proposed that Warren (NPC) was a good man and Nathan (NPC)'s behaviour needed to receive punishments. As Warren and Nathan have been looked real people, the players used moral regulations of real-world to reflect the two NPC's behaviours.

#### (4) The game design of particular style

The game design of particular style involves the themes of the game, details of design, music and voices. These could impact players' thoughts that reflect the game or living meaning or design issues. For example, in the game *Inside*, players used the poems to interpret the theme of the game and tried to set up the relationship between a poem and the game. This meant that they reconsidered game contents and meaning. From the perspective of details of design, players always wanted to know the function of a strange box in the game. They spent many months exploring a reasonable explanation for the function of the box. The players were curious about the voices from the secret room where the barn was full of pigs. In the game *Life is strange*, the players listened to the music of the game to consider the relationship between lyrics and their lives.

The game design of particular style supports promoting reflection.

i) *Inside* (Stand-alone): the players found the huddle scene was a funny design detail, they openly discussed if its design supported the plots.

ii) *Life is strange* (Stand-alone): most characters in this game are female, which made some players seriously consider if this game was designed based on a feminist theme.

The game design of particular style promotes suggestions for community development.

i) *Identity5* (Online): according to the player's observation, he suggested that the developer revised the distance between hunters and survivors (details of design). He thought that would be a chance to save the survivor who was camping.

ii) *Chinese Paladin: Sword and Fairy* (Stand-alone): through comparing the animation effects (details of design) of entering battles between old and new versions, the players found that the animation in the new version was unappropriated. They suggested that the developer revised the new version.

iii) *Inside* (Stand-alone): a player suggested that an error of the scene should be revised as the scene disappeared sometimes.

iv) *Life is strange* (Stand-alone): the player reflected that the application (e.g. the theme diversity of games such as the theme of females was applied in mainstream video games) was an innovation. He suggested that the publisher explored more similar applications in related games.

#### (5) Game stories

A game story can attract players' attention, including whether or not the game has rich and colourful plots, good characters and humorous narrative construction. When players play a good game, their feelings are similar to experiencing good movies or art, which is likely to trigger players' curiosity to further divergent thinking. When players feel unsatisfied with the game story, they want to change it. For example, for *Identity5* players rewrote the story themselves or gave some suggestions to developers. The game story should make players feel that the game required their participation to solve problems. For example, a friend has been kidnapped or a family reputation has been threatened. These situations need players to think about the ways of problem-solving (*Life is strange*).

Game stories support promoting reflection.

i) *Identity5* (Online): players attracted by the game story gathered some evidence and reread game content in order to present their theory that proved Gardner (the game character) actually was the hidden killer.

ii) *Inside* (Stand-alone): the player through analysing some details of the game story generated the theory that the last place in the game could be a real place, this meant it was not a fantasied product of the boy. It was used to explain some illogical situations such as why the huddle starts to laugh when you try breaking the wooden wall at the end.

iii) *Life is strange* (Stand-alone): the game story mainly described the relationship among multiple females such as friendship and love, which made players reflect if the game theme was designed from the perspective of females.

iv) *Lifeline* (Stand-alone): the player reflected the life rule that risks and rewards are both existences through considering the story about how to keep Taylor (game character) alive yet progress forward.

Game stories promote suggestions for community development.

i) *Life is strange* (Stand-alone): because the player was attracted by the emotional story of the game, he reflected kinship can come in many forms, such as a strong friendship could be looked equalling to kinship. Then players suggested writing more emotional plots for Max (game character).

ii) *Identity5* (Online): the player was extremely intrigued by the story of survivors and hunters. He wondered what the story would discover next. Hence, he suggested designing a story scenario where the player plays the detective to find clues as to what the game is about and who started it.

#### (6) Game avatars

Game avatars have original skills and personalities, which would lead to players' reflections. This includes revising skills, proposing the new strategies of the skills' matching and giving suggestions for creating new characters. For example, in the game *Identity5*, a player suggested revising the Joker's (game avatar) original skill which was weak in the player's opinion. Another player suggests creating a new character in order to make the game more interesting (*Identity5*).

In addition, game avatars were controlled by the players. The behaviours of the game avatars, to some extent, decide players' behaviours. These avatars impact each other as partners or enemies, which is likely to lead to players' reflections on avatars' performances. For example, a player of *Identity5* suggested that the survivors (avatars) needed to work together properly, which could ensure the success of teamwork.

Game avatars promote suggestions for community development.

i) *Identity5* (Online): the player used some evidence to prove that hunters' skills were more powerful than survivors'. They rethought the way of revising game avatar' skills to suggest

the developer would give survivors more time to do something rather than escape from hunters.

ii) *Life is strange* (stand-alone): the player considered that Max didn't call Chloe and stopped all communications with her after moving to Seattle. This did not correspond to the personality of Max (game avatar). Hence, he suggested the developer needed to address the issue that enables game avatars' personality was consistent with reasonable people's personality in real life.

Game avatars promote the change of evaluation.

i) *Identity5* (Online): the player through multiple tests suggested a new strategy that at the beginning of the battle, the hunter (game avatar)'s tentacle summon (skill) can be used to enhance a normal attack of the hunter.

Game avatars promote generating new assessments for other people.

i) *Identity5* (Online): the player through rethinking the skills of these survivors who controlled by other players showed his new assessment for these survivors: survivors should have more options themselves to decide the victory of the battle rather than relying on other teammates' abilities.

ii) *Life is strange* (stand-alone): the player thought Chloe (game avatar) was gay before, but after playing the game he changed his idea that from the perspective of bisexual to assess Chloe' behaviours.

#### (7) Tasks for problem-solving

Tasks for problem-solving mean players faced challenges such as intellectual challenges and annoying pitfalls. Players need to consider various methods in order to solve these problems. At this more abstract level, players use existing resources and information to determine the best course of action. For example, when players encounter more powerful monsters, they need to constantly master and reflect on new skills or different strategies in order to ensure their win (e.g. *Identity5*, players decided how to organise the marching order in order to guarantee their survival in case of a surprise attack).

Tasks for problem-solving promote suggestions for community development.

i) *Identity5* (Online): in order to stop campers from camping (the mission), some players suggested the developer added more players.

Tasks for problem-solving promote the change of evaluation.

i) *Inside* (Stand-alone): a player tried to deactivate some orbs to glow the left bulb (new method) in order to make the secret elevator becomes accessible.

ii) *Identity5* (Online): for solving the task that how to deal with camping hunters, a player presented many new strategies, such as a normal survivor used his or her body to block the escaped one; a specific skill was used to hit the hunter into a wall from a distance; nurses and coordinators (other avatars) were selected as teammates.

iii) *Chinese Paladin: Sword and Fairy* (Stand-alone): the player utilised the wine and roasted meat to synthesize gold silkworm (new strategy), which could receive the achievement of earning much money (task).

#### (8) The rank system of the game

The rank system of the game involved game outcome and pairing. Players' attentions for game outcome and pairing was likely to lead to their rethinking. For example, in *Identity5*, one player reflected that the counting of game points based on the players' skills rather than the performance of the whole team.

The rank system of the game promote suggestions for community development.

i) *Identity5* (Online): the player suggested that prohibited this pairing of duplicate character selection (e.g. two coordinators' combination and two perfumers' combination) in order to ensure the fairness of battles.

#### (9) Game tools

The application of game tools shows players' new strategies, which improve characters' skills. For example, in *Identity5* players put down a blue music box in the middle of the battle area to delay hunters' movement and slow down the speed of the attack. This method was found through players' multiple tests in the game.

Game tools support promoting reflection.

i) *Inside* (Stand-alone): the player through the use of the shockwave (game tool) in the game divergently thought how to use the decibel scale to measure voice.

Game tools promote suggestions for community development.

i) *Identity5* (Online): players expected to win the battle with more rate, then they suggested that add first aid kits (game tool) somewhere on every map in order to increase the chance of survivors' self-healing.

Game tools promote the change of evaluation.

i) *Inside* (Stand-alone): the players utilised the same vertical box in different situations (ledges or the roof of buildings) to help them complete difficult jump or movement.

ii) *Identity5* (Online): the player recommended using broken windows (game tool) with every single character, and also tide turner (game tool) to increase the escape time.

iii) *Chinese Paladin: Sword and Fairy* (Stand-alone): in order to improve fast the ranks of the character, players advised new strategies for problem-solving, such as using transmission gear (game tool) or using the special brush (game tool) to assist the battles.

#### (10) Game mode

Game mode refers to game rules and functions. Players will give suggestions that revise or optimise game mode. For instance, a player wanted to revise the game rule that allowed participants to play against real people after 15 competitions. This is because player thought

that they could not learn concrete skills by playing with/against robot mates or enemies (*Identity 5*).

Game mode promotes suggestions for community development.

i) *Identity5* (Online): in order to more understand other friends, a player suggested that add a function to check another player who has the most owned characters and what character he or she used.

# 4.3.3 How game features promote these criteria of reframing value?

The following figures systematised the question.



Figure 4.4 The model of game features promoting one criterion of reframing value



Figure 4.5 The model of game features promoting two criteria of reframing value



Figure 4.6 The model of game features promoting three criteria of reframing value

Most game features supported *suggestions for community development* (n=6) and *promoting reflection* (n=7). The 3 game features (game stories, game tools and game avatars) broadly promoted reframing value, including *suggestions for community development, promoting reflection* and *the change of evaluation*. The Top 5 games cover a range of game features. The lowest number was found in Lifeline (n=1), while the highest numbers were in Identity5 (n=6) and Life is strange (n=8).

Some game features showed positive effects on promoting reframing value. For example, two game features (the game design of particular style and game stories) supported *promoting reflection*. *Suggestions for community development* was most impacted by game avatars and *the change of evaluation* was most impacted by tasks for problem-solving and game avatars.

*Generating new assessments for other people* was only promoted by game avatars. This meant that game features had little impact on this criterion of reframing value. See Figure 4.7.

Name	/ 🝔	Files	References
Generating new assessments for other people		0	
game avatars		13	2
Promoting reflection		1	
a real no-play character (NPC) image		5	1
game stories		16	4
game tools		1	
multiple- endings of the game		2	
Tasks with emotional experience		4	
The game design of particular style		23	9
Suggestions for community development		2	
game avatars		50	7
game mode		7	
game stories		3	
game tools		6	
task for problem-solving		5	
The game design of particular style		8	-
the rank system of the game		6	
The change of evaluation		0	
game avatars		20	2
game tools		10	-

# Figure 4.7 The relationship between game features and reframing value

# 4.3.4 An instrument for evaluating the reliability of the model

# 4.3.4.1 The reliability of the model

The reliability of the model was shown through the feedback of 39 experts' agreement. The average percentage of "agree" and "strongly agree" for game features promoting reframing value was 87.6%, which meant experts acknowledged the 10 game features could promote reframing value. See Table 4.5.

Criteria of reframing value and involved game features	game feature	Percentage (%) of agree + strongly agree
	a real NPC image	84.6
	game stories	94.9
	multiple endings of the game	89.8
Promoting reflection	the game design of particular style	89.8
	tasks with emotional experience	89.7
	game tools	76.9
Generating new assessments for other people	game avatars	79.5
	game stories	96.6
	game avatars	79.5
	game mode	97.1
Suggestions for community development	the game design of particular style	89.7
	tasks for problem-solving	92.3
	game tools	79.5
	the rank system of the game	87.2
	game avatars	89.7
The change of evaluation	Tasks for problem-solving	89.7
	game tools	82.1

# Table 4.5: Participants' feedback

# 4.3.5 The validity of the model

# 4.3.5.1 The related game features

From the perspective of the response rate (RR), most participants agreed on the following facts: *Chinese Paladin: Sword and Fairy 3* covered 4 game features, including multiple endings of the game RR=76.47%, task with emotional experience RR=79.41%, a real NPC image RR=70.59% and game stories RR=79.41%, Figure 4.8; *Limbo* covered three game features, including the game design of particular style RR=89.19%, game tools RR=86.49% and tasks for problem-solving RR=81.08%, Figure 4.9; *World of Warcraft* covered three game features,

including game avatars RR=74.29%, the rank system of the game RR=82.86% and game mode RR=88.57%, Figure 4.10.



Figure 4.8 The frequency of the response (Chinese Paladin: Sword and Fairy 3)



Figure 4.9 The frequency of the response (*Limbo*)



Figure 4.10 The frequency of the response (*World of Warcraft*)

The relationship between game features and four criteria of reframing value (answering RQ: what is the validity of the model)

The following results showed that 10 game features have strong associations with four criteria of reframing value. This was consistent with this model of game features promoting reframing value.

The factor analysis was performed for the three investigations respectively. the KMO measure of sampling adequacy (0.814 *World of Warcraft*, 0.722 *limbo*,0.843 *Chinese Paladin: Sword and Fairy 3*) was high and Bartlett's Test of Sphericity (p < 0.05) respectively reached statistical significance (Kaiser, 1958, Bratt and Fagerström, 2020).

For the factor analysis of the game experience test (GET) (*World of Warcraft*), three factors (game avatars, the rank system of the game and game mode) were extracted and the cumulative explanation of variance after rotation was high (93.89%) (Kaiser, 1958). This meant that three factors promote related criteria of reframing value. The indicators included game avatars promote *suggestions for community development* (0.846 common variances), *the change of evaluation* (0.948 common variances) and generating new assessments for other people (0.949 common variances); the rank system of the game promote *suggestions for community development* (0.955 common variances); game mode promotes *suggestions for community development* (0.997 common variances). See Figure 4.11.



Figure 4.11 Factor analysis (factor loading) with varimax rotation of measures of the relationship between game features and criteria of reframing value (*World of Warcraft*)

The cumulative explanation of variance after rotation was 90.69% in the factor analysis of GET (*Limbo*). Three factors (the game design of particular style, game tools and tasks for problem-solving) extracted have strongly correlated with related criteria of reframing value. The indicators included the game design of particular style supports *promoting reflection* (0.581 common variances) and promotes *suggestions for community development* (0.914 common variances); game tools support *promoting reflection* (0.957 common variances) and *suggestions for community development* (0.964 common variances); tasks for problem-solving promote *suggestions for community development* (0.964 common variances) and *the change of evaluation* (0.964 common variances). The loading on the factor (game tools) promoting the change of evaluation was low (less than 0.400). According to Sass (2010), the factor loading did not represent the primary factor (game tools). However, comparing responses of participants was utilised to further check the validity of game tools promoting *the change of evaluation*. The result showed that the percentage of "agree" and "strongly agree" for game tools promoting *the change of evaluation* was 89.19%. See Figure 4.12.



Figure 4.12 Factor analysis (factor loading) with varimax rotation of measures of the relationship between game features and criteria of reframing value (*Limbo*)

Four factors were extracted (multiple- endings of the game, tasks with emotional experience, a real NPC image and game stories) in the factor analysis of GET (*Chinese Paladin: Sword and Fairy 3*). The cumulative explanation of variance after rotation was 97.10%. The indicators included multiple endings of the game support *promoting reflection* (0.960 common variances); tasks with emotional experience support *promoting reflection* (0.846 common variances); a real NPC character image support *promoting reflection* (0.993 common

variances); game stories support *promoting reflection* (0.727 common variances) and *suggestions for community development* (0.914 common variances). See Figure 4.13.



Figure 4.13 Factor analysis (factor loading) with varimax rotation of measures of the relationship between game features and criteria of reframing value (*Chinese Paladin: Sword and Fairy* 3)

Furthermore, this research did not analyse participants' suggestion that improves the game feature to promote reframing value. Only one participant in the questionnaires gave suggestions.

In addition, the Multifactor Analysis of Variance (ANOVA) was used to measure the major influence across variables between teachers and students on the consent of game features. The testers (106) coming from China were divided into three groups (34 participants evaluated 5 items in Chinese Paladin: Sword and Fairy 3, 37 participants evaluated 7 items in Limbo and

35 participants evaluated 5 items in World of Warcraft). The three groups were composed respectively of 19 teachers and 15 students, 20 teachers and 17 students, 18 teachers and 17 students. These testers were selected randomly, some usually played digital games and some did not play.

For *Limbo*, there was a statistically significant difference (p<0.05) between teachers and students, involving tasks with emotional experience promoting the change of evaluation, Table 4.6.

Game feature	Criteria of reframing value and	(Mean±Std	F			
Game leature	involved game features	teacher(n=20) student(n=1)		r	р	
The game design of	Promoting reflection	4.40±0.60	4.35±1.00	0.031	0.860	
particular style	Suggestions for community development	4.30±0.86	4.29±1.05	0.000	0.985	
	Promoting reflection	4.40±0.88	4.41±1.00	0.001	0.970	
Game tools	Suggestions for community development	4.65±0.67	4.18±1.13	2.485	0.124	
	The change of evaluation	4.60±0.60	4.06±1.39	2.496	0.123	
Tasks for problem-	Suggestions for community development	4.60±0.50	4.12±1.05	3.316	0.077	
solving	The change of evaluation	4.75±0.44	4.29±0.69	5.926	<mark>0.020</mark>	
p<0.05, there was a st	atistically significant difference.	1	1	<u> </u>	<u> </u>	

Table 4.6: Limbo- Results of One-way ANOVA Analysis

For *World of Warcraft* and *Chinese Paladin: Sword and Fairy 3* the results (p>0.1) illustrated that there was not a statistically significant difference, see Table 4.7,4.8.

Game feature	Criteria of reframing value and	(Mean±Std	F	n		
Game leature	involved game features	teacher(n=18)	student(n=17)		р	
	Suggestions for community development	4.50±0.71	4.47±1.28	0.007	0.933	
Game avatars	The change of evaluation	4.67±0.49	4.29±1.05	1.860	0.182	
	Generating new assessments for other people	4.78±0.43	4.35±1.11	2.265	0.142	
The rank system of the game	Suggestions for community development	4.61±0.50	4.24±1.56	0.940	0.339	
Game mode	Suggestions for community development	4.22±1.00	4.41±1.06	0.294	0.591	
p>0.1, there was not a statistically significant difference.						

Table 4.7: World of Warcraft- Results of One-way ANOVA Analysis

Table 4.8: Chinese Paladin: Sword and Fairy 3- Results of One-way ANOVA Analysis

Game feature	Criteria of reframing value and	(Mean±Std	F	n	
Game leature	involved game features	teacher(n=19)	student(n=15)	-	р
Multiple endings of the game	Promoting reflection	4.53±0.96	4.40±0.99	0.141	0.710
Tasks with emotional experience	Promoting reflection	4.42±1.30	4.00±1.13	0.978	0.330
A real NPC image	Generating new assessments for other people	4.05±1.08	3.80±1.26	0.395	0.534
Game stories	Suggestions for community development	4.53±0.84	4.00±1.13	2.417	0.130
Game stories	Promoting reflection	4.53±0.51	4.07±1.16	2.394	0.132
p>0.1, there was not a statistically significant difference.					

# 4.4 Summary

This chapter describes the research results. For exploring serious games supporting IBL, the results revealed that a wide variety of game elements were used including storylines, interactive artefacts, gameplay-context coherence and task assessment mechanics. These elements were mainly oriented at supporting three core features of IBL.

For value creation in COI, the results showed that value creation was really impacted by TP, SP and CP. The process of creating values was not linear. Most activities of TP promoted immediate value. Immediate and potential values were positively promoted by most activities of SP. The activities of CP have specifically promoted three or more values. Reframing value was not positively promoted by COI.

For constructing a model of game features promoting reframing value, the results showed that both the reliability and validity of the model were high. Participants of tests agreed that 10 game features were extracted such as multiple endings of the game, game stories, the task with emotional experience, game avatars and game tools. These features were oriented at promoting four criteria of reframing value (e.g. *promoting reflection, the change of evaluation, generating new assessments for other people* and *suggestions for community development*).

# **5** Discussion

This chapter gives related discussions. For example, the suitability of role-playing games for supporting inquiry-based learning (IBL); the importance of promoting reframing value; some special game features that only appear in stand-alone or online games to promote reframing value.

# 5.1 Exploring serious games supporting IBL

IBL serious games can be used in formal education (e.g. classroom). This research proposed that playing game time was spent in the classroom in most studies reviewed (n=8). This is consistent in many kinds of literature that serious games were broadly applied in formal education (Beavis et al., 2012, Gee, 2007, Steinkuehler, 2010, Arnab et al., 2012).

RPGs suit to support IBL given its nature. This study indicated that most projects reported used RPG as their game genre. Tychsen (2006) proposed that RPGs are distinguished from other game genres through controlling the virtual characters with the use of storytelling in a fictional reality, this is consistent with the guidance of IBL. For example, Hwang et al. (2015) use RPGs to explore the learning of financial knowledge. The storyline of the game guilds learners to enter Saving Island (game scene) to collect all the scrolls (game tools). Then they implement tasks in the Finances Bridge (game scene) following the storyline. Finally, learners achieve enough scores to participate in the competition in Investment Island (game scene) until the end of the game (Hwang et al., 2015).

The diversity of game genres in IBL contexts is an option for multi-dimensional solutions (e.g.one solution for multiple questions or multiple solutions for one question). For example, on one hand, using RPGs solve biology questions (Sabourin et al., 2012) or study chemical knowledge (San Chee et al., 2011). On another hand, San Chee et al. (2011) allow learners to explore chemical knowledge using RPGs, but Meesuk and Srisawasdi (2014) chose the shooting game.

This review indicates that there is wide variability in how games incorporate game features to support IBL. Some of the reviewed games included six or more game features that supported two or all defining attributes of IBL, while others used two or three game features in what seemed to be more gamified learning activities in fully structured games.

The stage-based but flexible structure of IBL was mainly supported by the use of storylines, structures game tasks, and the possibility to explore information in a non-linear way. Storylines provided a narrative that structures goals and tasks in a logical way. In this way, concepts and processes were broken into smaller units, with their own progression and dependencies, but maintaining coherence with the general story. In addition, several games activities presented a goal-oriented task structure consistent with IBL stages, structuring game actions and knowledge in an incremental way, despite being designed following different IBL theoretical models. For example, in Hwang et al. (2015), the game's storyline provided a general frame to contextualise financial knowledge. Tasks were organised so students explored simple concepts (e.g. savings) before tackling more complex ones (e.g. investments). San Chee et al (2011) used a game following a 6-stage model of inquiry (question identification, hypothesis

formulation, investigation, analysis, synthesis, and evaluation), in which each level had a specific task with goals and learning activities. Completing a level meant completing a stage of inquiry, which likely promoted students' progressive understanding of complex concepts and fostered their sense of mastery and feeling gratified. Interestingly, only a reduced number of games supported the cyclic and multi-directional nature of the inquiry, allowing students to explore information and activities in a non-linear manner. IBL has been conceptualised as a process developed in loops within and between stages (Pedaste et al., 2015), which can generate different inquiry cycles. These findings suggest that game features have been designed primarily to structure the stages of the inquiry process as a sequenced set of steps, and less oriented at providing students with the opportunities to create their own pathways and inquiry cycles.

This is consistent with these findings regarding how game features supported the guidance that student need when participating in IBL activities. This analysis revealed that games used several game features such as game tips and/or interactive artefacts to guide completing tasks to advance to the following stage of the learning process. Furthermore, in five games it was necessary to complete the previous task before the next one was unlocked, which evidenced that games were designed to guide students through stages in an orderly and sequenced manner. This suggests that game features have primarily been used in games to provide stronger forms of guidance which may be classified in level 1 (confirmatory) and level 2 (structured) according to Banchi and Bell (2008) categorisation.

In terms of the meaningful environment key to IBL, these findings indicate that game features like storylines, coherence in of the meaning of the context, gameplay actions and player roles, and contextualised feedback, were designed to create a game environment that promotes immersion, fosters the construction of situated knowledge, the integration of prior and new knowledge, and the applicability of knowledge in different environments.

Most studies recreated game scenarios that mirrored real-world contexts or dynamics. These game designers seem to have taken the approach of simulations when moving into game design. On one hand, integrating games and simulations has a positive impact on learning goals (Vlachopoulos and Makri, 2017). On the other hand, games things do not have to be realistic to be effective. As National Research Council (2011) showed that when games and simulations were compared that there would be more positive results from the game rather than from the simulation because if the simulation was not 100% accurate to the real-life (i.e. introducing an element of error) then the negative transfer of learning would take place (Novick, 1988). Furthermore, Mimicking authentic environments has been considered one essential aspect for IBL (Brown et al., 1989, Huang et al., 2014), to enable the exploration and application of knowledge in contexts that will facilitate transferability of learning and develop problem-solving skills.

According to the literature, students' achievement of understanding knowledge and problemsolving strategies showed positive results while them playing an IBL-supporting game. The author suggests that when constructing knowledge and carrying out problem- solving, adopt serious games to apply IBL is a good approach. Furthermore, some research (i.e. Hwang, et al., 2015, Hwang and Chen, 2017, Kalz M et al., 2014, Meesuk and Srisawasdi, 2014, Firssova et al., 2014) focuses on the motivation of students, mixed results were found regarding the impacts on engagement. However, the author could not find the regularity involving the engagement impacts of serious games support IBL.

# 5.2 The value creation in communities of inquiry

# **5.2.1 Discussion**

The activities of communities of inquiry (COI) collected positively relevant with some questions of value creation. See Table 5.1.

The five values	reflecting questions of value creation		
	What happened/were the important things?		
Immediate value	Is it interesting/attractive?		
	Who did I contact or have established with whom?		
<b>Potential value</b> Do I believe they are enough to ask them for help and am I not feeling so			
Applied valueHow is it different from my practice/life/background?			
Realized value         Am I more successful?			
Reframing value	Does the social learning process led to a reflection on important issues?		

Table 5.1: The activities of COI involving some questions of value creation

Most activities met one or three questions of values, there is much room for improvement as activities seem to address a few questions. For example, every value includes multiple questions to reflect, according to these questions the content of activities can be redesigned, which is likely to develop more values.

Most values can be created by activities of COI in a formal learning environment where TP, SP and CP positively affected immediate and potential values, see Table 5.2. However, this result cannot explain if a formal learning environment is a precondition for generate values. Further research will focus on the relationship between values and the learning environment in the future.

<b>F</b>		r	I	
		SP	Group cohesion	Keles (2018)
				Setiani and MacKinnon (2015)
				Eteokleous and Ktoridou (2012)
			Open	Padilla and Kreider (2018)
			communication	Eteokleous and Ktoridou (2012)
	Immediate value		Individual or	Etacklasus and Ktoriday (2012)
			affective	Eteokleous and Ktoridou (2012)
		TP	Promoting discourse	Keles (2018)
				Setiani and MacKinnon (2015)
			discourse	Padilla and Kreider (2018)
				Keles (2018)
			Direct instruction	Padilla and Kreider (2018)
			Design and	Padilla and Kreider (2018)
			organization	Eteokleous and Ktoridou (2012)
		<b></b>	Triggering event	Setiani and MacKinnon (2015)
		CP		Padilla and Kreider (2018)
	Potential value	SP	Group cohesion	Keles (2018)
Formal				Eteokleous and Ktoridou (2012)
learning			Open	
environment			communication	Eteokleous and Ktoridou (2012)
			Individual or	
			affective	Eteokleous and Ktoridou (2012)
		ТР	Design and	
			organization	Eteokleous and Ktoridou (2012)
		СР	Exploration	Setiani and MacKinnon (2015)
				Padilla and Kreider (2018)
			Triggering event	Eteokleous and Ktoridou (2012)
		СР		Setiani and MacKinnon (2015)
	Applied value		Integration	Eteokleous and Ktoridou (2012)
			Resolution	Eteokleous and Ktoridou (2012)
				Padilla and Kreider (2018)
			Exploration	Eteokleous and Ktoridou (2012)
	Realized value		· ·	
		СР	Integration	Eteokleous and Ktoridou (2012)
			Resolution	Eteokleous and Ktoridou (2012)
			Exploration	Eteokleous and Ktoridou (2012)
			•	

 Table 5.2: The values and formal learning environment

#### 5.2.2 The importance of promoting reframing value

This research found that reframing value was not positively promoted. Given the characteristic that every value can be analysed independently, the author proposed to consider the importance of promoting reframing value.

Firstly, promoting reframing value is similar to promoting critical reflection that can improve learners' skills (e.g. Brookfield, 2015; Lonka et al., 2001; Blatt et al., 2007; Mamede et al., 2008) playing an important role in education. For example, Wenger et al. (2011) indicate that through communication in communities, learners not only contribute their experience but get new knowledge from others. This may be trigger reconsidering old knowledge construction of learners to carry out their self-development.

Furthermore, from the perspective of stakeholders, they expected to promote reframing value. For example, students who were the stakeholder of reframing value care more about addressing issues that happen in their practice and defining success (Wenger et al., 2011). In other words, students are likely to be more focused on the development of reframing value.

Generally, educational strategies could provide opportunities to promote reframing value, many studies indicate that some educational strategies include games (e.g. Apperley and Beavis, 2011; 2013; Hsiao, 2007; Squire and Barab, 2004). For example, games (e.g. role-playing, adventure, simulation ) provide a slower pace to proceed so that more space and time are offered to reflect (Squire and Barab, 2004).
### 5.3 A model of game features promoting reframing value

This study indicates that there is wide variability in how game features promote criteria of reframing value. Some criteria include one or three game features that supported two or three criteria, while others used mixed game features (e.g. some supported two or three criteria, others supported one).

This research could not find a clear association between game features. However, some identified game features are consistently oriented at supporting the same criterion of reframing value. For example, tasks with emotional experience, game avatars and game stories together impact on *promoting reflection*. A player's comment was collected: "when it comes to Max being gay or straight, I am thinking as what others are saying, she might be interested in both but has yet had the chance to explore it, which she will likely get with her interactions with Chloe, which I admit is a bad influence, but Max is also good for her, there have been many real-life people that have had bad habits, smoking drinking, drugs and when they meet someone, they could be just friends or more and their presence and friendship or whatever it is, allows him to be transformed into something better for them. That is what likely will happen here with Max and Chloe" (*Life is strange*). This comment shows that the tasks involving love and friendship emotional experience (Max and Chloe) increasingly impact the player to post his ideas about the sexual orientation of game avatars (Max). Furthermore, the player explains his opinion (Max being bisexual) based on the interaction of Max and Chloe (the game story) and his own reflections on the real world (e.g. some people could be influenced by people they

care for). Hence, in this example, the three features (tasks with emotional experience, game avatars and game stories) concurrently support the reframing value (*promoting reflection*).

Some factors which are not game features also trigger players' rethinking involving promoting *suggestions for community development* and *promoting reflection*. For example, players suggested that the developer used only one English name for the game *Chinese Paladin: Sword and Fairy*. This could assist customers to recognize the game on different game platforms (*suggestions for community development*). The whole game experience of *Identity5* makes the player feel confident. He/she rethinks personal value to share the experience with people who are in the same community (*promoting reflection*).

Some special game features only appear in stand-alone or online games to promote reframing value. This means that the game classification is likely, to some extent, to impact reframing value. For instance, multiple endings of the game appear in stand-alone RPGs, which positively influence reframing value (*promoting reflection*). The rank system of the game and game mode appears in online RPGs to promote reframing value (*suggestions for community development*).

### 5.3.1 An instrument for evaluating the reliability of the model

### **5.3.1.1** The limitations of this instrument

The result suggests that the instrument would be an effective tool for examining the potential that game features elicit the creation of reframing value. However, this instrument which evaluates the reliability of the model is only at an exploratory stage.

## **5.4 Summary**

The chapter gives diverse discussions including:

1) limitations of this study, and what could be done to address this. For example, the samples of games selected were limited from the IOS system. The author needs samples of games collected from broader game markets such as the markets of the IOS system and the android system; 2) relevance of the findings to inform practice. For instance, based on these findings, games could be used as an educational tactic to enhance reframing value; 3) how these findings can inform future research. For example, reframing value did not be promoted by COI, the author proposes that consider the importance to promote reframing value.

# 6 Conclusions and Future work

This chapter presents an overall conclusion of this study and according to related limitations in current research proposes suggestions for future work.

### **6.1** Conclusions

Games are playing an increasingly important role in education (e.g. Fabricatore et al., 2019; Bressler et al.,2021; Hwang and Chen, 2017; Erickson et al., 2021; Barr, 2017). Literature indicated that both theories and demonstration have shown that games in multiple domains can effectively strengthen learners' motivation, engagement (Squire, 2008, Mayer, 2019, Fabricatore et al., 2019, De Freitas, 2006, Bayeck, 2020) and critical reflection (Apperley and Beavis, 2011, Apperley and Beavis, 2013, Hsiao, 2007, Squire and Barab, 2004). In particular, the use of digital games and game-based activities can positively impact social learning (Bakhanova et al., 2020, Medema et al., 2016, Den Haan and Van der Voort, 2018).

The thesis aims to investigate an exploratory study that digital games are used to promote the development of a virtual inquiry-based learning community (VILC). VILC has been proved that is an important framework given that it enhances students' problem-solving skills (e.g. Eteokleous and Ktoridou, 2012; Setiani and MacKinnon, 2015; Chanprasitchai and Khlaisang, 2016; Suarez et al., 2016). However, little attention has been paid that promotes the value in social learning generated by VILC.

There exists, a relationship between games and learning values, with learning value generated in social learning, and the use of games supporting social learning (Bakhanova et al., 2020, Medema et al., 2016, Den Haan and Van der Voort, 2018). This thesis proposes that in order to improve the development of the learning value of VILC, digital games are used as specific communities (Bolter et al., 2000) as they offer opportunities to create value. A value creation framework i.e. Wenger et al.'s framework (2011) is utilised to evaluate social learning in games or communities. As the community of inquiry (COI) is related to the nature of knowledge construction in inquiry-based learning (IBL), VILC involves both COI and IBL. Utilising games to support inquiry-based learning (IBL) and communities of inquiry (COI) considerably challenges the related issues that identifying game features and the extent to which they promote IBL and COI. According to this, this research carries out investigations involving the use of game characteristics to support IBL and promote learning values of COI.

Meanwhile, the research onion (Saunders et al., 2009) had been utilised throughout the methodology of this thesis as an effective tool to help the organisation of this research and development of the research design according to the layers of the research onion (i.e. research philosophy, research methods, research strategies, time horizons, and data collection methods). The research conclusions finished in the thesis are showed as follows.

### 6.1.1 Exploring serious games supporting IBL

The first step of this research presented a review to analysing the characteristics of a gamebased IBL strategy, determining the game features utilised in the games and the gamification strategies designed to promote IBL and their effects on students. The results revealed that the game genre of RPG is used given the high suitability to support IBL. A few studies involve utilising IBL-serious games to support teacher training. Several game features can be used to support inquiry processes, including storylines, interactive artefacts, gameplay-context coherence and task assessment mechanics. From the perspective of game impacting IBL, positive impacts were identified both in terms of conceptual learning and information seeking strategies, but mixed evidence regarding the impacts on engagement was shown. This study suggests that these game features can be designed to provide different levels of support to students, structure the learning progression according to different theoretical models and create diverse environments to support making sense of knowledge in context. Games present intrinsic features that can naturally support the core aspects of IBL: students can take active control of the pace of their progress, exert autonomy for decision-making and timely observe the relationships between actions, concepts and context. There is much space for these features to be explored and used by games aiming at supporting IBL.

### 6.1.2 The value creation in COI

In order to clarify why games are used to support COI and how to promote COI, a systematic synthesis is firstly proposed for the investigation of value creation in COI, and the extent to which TP, SP and CP represent COI aspects determining the value generation process. The results of the second step showed that the activities of COI mainly generated immediate, potential and applied values and value creation was really impacted by teaching, social and cognitive presences. For example, most activities of TP promoted immediate value, especially its categories of promoting discourse and direct instruction. Most activities of SP promoted

immediate and potential values. Open communication positively generated immediate value. The activities of CP promoted three or more values. In summary, this study can provide an important mean to explore the nature of social learning; be utilised as an effective way of seeing implicit learning outcomes of COI; make stakeholders (e.g. participants of a virtual learning community, the organizations of a community and their sponsors) achieve their expected outcomes; optimize the design of related activities to maximize learning value.

Meanwhile, the research result revealed that reframing value was not positively promoted by COI. Due to the fact that promoting reframing value is similar to promoting critical reflection, this research proposes that games are utilised as an effective educational strategy providing opportunities to create reframing value.

### 6.1.3 A model of game features promoting reframing value

The third step of this research constructs a model in order to identify which game features can support generating reframing value and how they promote reframing value. RPGs are used as context games given that it is highly suitable to support IBL and popularity of applying education. Ten game features were qualitatively extracted, including multiple endings of the game, tasks with emotional experience, a real no-play character image, the game design of particular style, game stories, game avatars, tasks for problem-solving, the rank system of the game, game tools and game mode. This model bridged game features of RPG and four criteria of reframing value, which will benefit game designers, related educators and researchers who use games aiming at promoting reframing value or regard the need for a model to link games features with learning value. The relationship between digital RPGs and reframing value can

support reflection of multiple dimensions (i.e. four criteria of reframing value ) in the process of social learning, create related environments assisting students to understand the knowledge and provide a reference for selecting and designing games oriented at promoting critical reflection. A guideline was attached in order to provide support to related users of the model.

### 6.2 Future work

In this thesis, games have been investigated as a robust tool to support IBL and COI. However, some limitations still exist in the current work. For example, the top 5 RPGs which were collected come from the IOS system, this brings some limitations. The author needs samples of games collected from broader game markets such as the markets of the IOS system and android system.

Further research to address the issues which find within research processes (i.e. a few studies involve utilising IBL-serious games to support teacher training). This research from a broader perspective investigates the impacts of serious games (SG) on teachers' professional development. The author has finished some studies, including investigating what contexts SGs have been used for teacher education, what types of games have been used for teacher education and what impacts SGs can have on teachers' professional development. However, the author plans to study associations between impacts, characteristics of the participants and game types/features, as this could more robustly inform the planning and evaluation of game-based teacher education, by allowing to identify these game features suitable to promote desirable effects in a specific target population.

# **Publications**

**Refereed Conference Papers** 

Gao, L., Fabricatore, C. and Lopez, M.X., 2019, September. Game Features in Inquiry Game-based Learning Strategies: A Systematic Synthesis. In 13th European Conference on Games Based Learning (pp. 854-862). Academic Conferences and Publishing International. The content of this paper involves the first step of my research.

Gao, L., Fabricatore, C. and Lopez, M.X., 2020, September. Feasibility of Using Serious Games for Teachers' Professional Development: A Review of the Current Literature. In 14th European Conference on Games Based Learning (pp. 701-710). Academic Conferences and Publishing International. The content of this paper addresses the new issue (e.g. only a few studies involve utilising IBL-serious games to support teacher training) which find within research processes.

Gao, L., Ward, R., Fabricatore, C. and Lopez, M.X., 2021, January. The Value Creation in Communities of Inquiry. In 2020 International Joint Conference on Information, Media and Engineering (IJCIME) (accepted). IEEE. The content of this paper involves the second step of my research.

# References

- ADAMO-VILLANI, N., HALEY-HERMIZ, T. & CUTLER, R. Using a serious game approach to teach'operator precedence'to introductory programming students. 2013 17th International Conference on Information Visualisation, 2013. IEEE, 523-526.
- AIVELO, T. & UITTO, A. 2016. Digital gaming for evolutionary biology learning: The case study of parasite race, an augmented reality location-based game. *LUMAT: International Journal on Math, Science and Technology Education,* 4, 1-26.
- ALSUBAIE, S., ALMATHAMI, M., ABOUELYAZID, A., ALQAHTANI, M. M., ALSHEHRI, W. & ALAMRI, A. 2020. Prevalence of obsessive compulsive disorder: A Survey with Southern Saudi Arabian samples. *J Psychiatry Depress Anxiety*, 6, 031.
- ALYAZ, Y. & GENC, Z. S. 2016. Digital game-based language learning in foreign language teacher education. *Turkish Online Journal of Distance Education*, 17.
- AMPATZIDOU, C., CONSTANTINESCU, T., BERGER, M., JAUSCHNEG, M., GUGERELL, K. & DEVISCH, O. J. U. P. 2018. All work and no play? Facilitating serious games and gamified applications in participatory urban planning and governance. 3, 34-46.
- ANNETTA, L., LAMB, R., MINOGUE, J., FOLTA, E., HOLMES, S., VALLETT, D. & CHENG, R. 2014. Safe science classrooms: Teacher training through serious educational games. *Information Sciences*, 264, 61-74.
- APPERLEY, T. & BEAVIS, C. 2011. Literacy into action: Digital games as action and text in the English and literacy classroom. *Pedagogies: An International Journal*, 6, 130-143.
- APPERLEY, T. & BEAVIS, C. 2013. A model for critical games literacy. *E-learning and Digital Media*, 10, 1-12.
- ARNAB, S., BERTA, R., EARP, J., DE FREITAS, S., POPESCU, M., ROMERO, M., STANESCU, I. & USART, M. 2012. Framing the adoption of serious games in formal education. *Electronic Journal of e-learning*, 10, 159-171.
- ARNAB, S., PETRIDIS, P., DUNWELL, I. & DE FREITAS, S. 2011. Enhancing learning in distributed virtual worlds through touch: a browser-based architecture for haptic interaction. *Serious Games and Edutainment Applications*. Springer.
- BAKHANOVA, E., GARCIA, J. A., RAFFE, W. L., VOINOV, A. J. E. M. & SOFTWARE 2020. Targeting social learning and engagement: what serious games and gamification can offer to participatory modeling. 104846.
- BANCHI, H. & BELL, R. 2008. The many levels of inquiry. Science and children, 46, 26.

- BANDURA, A. & WALTERS, R. H. 1977. *Social learning theory*, Prentice-hall Englewood Cliffs, NJ.
- BANDURA, A. J. A. P. 1982. Self-efficacy mechanism in human agency. 37, 122.
- BANGERT, A. 2008. The influence of social presence and teaching presence on the quality of online critical inquiry. *Journal of Computing in Higher Education*, 20, 34-61.
- BARR, M. 2017. Video games can develop graduate skills in higher education students: A randomised trial. *Computers Education*, 113, 86-97.
- BARRON, B. & DARLING-HAMMOND, L. 2008. Teaching for meaningful learning: A review of research on inquiry-based and cooperative learning. *Powerful learning: What we know about teaching for understanding*, 11-70.
- BAYECK, R. Y. 2020. Examining Board Gameplay and Learning: A Multidisciplinary Review of Recent Research. *Simulation & Gaming*, 1046878119901286.
- BEAVIS, C., O'MARA, J. & MCNEICE, L. 2012. *Digital games: Literacy in action*, Wakefield Press.
- BECU, N., AMALRIC, M., ANSELME, B., BECK, E., BERTIN, X., DELAY, E., LONG, N., MARILLEAU, N., PIGNON-MUSSAUD, C., ROUSSEAUX, F. J. E. M. & SOFTWARE 2017. Participatory simulation to foster social learning on coastal flooding prevention. 98, 1-11.
- BELL T, U. D., SCHANZE S, PLOETZNER R 2010. Collaborative inquiry learning: models, tools, and challenges. *Int J Sci Educ*, 32(3), 349–377.
- BELL, T., URHAHNE, D., SCHANZE, S. & PLOETZNER, R. 2010. Collaborative inquiry learning: Models, tools, and challenges. *International journal of science education*, 32, 349-377.
- BENNETT, M. 2015. The Invisible Hand of Inquiry-based Learning. *Childhood Education*, 91, 388-389.
- BIGGS, J. 1996. Enhancing teaching through constructive alignment. *Higher education*, 32, 347-364.
- BLATT, B., PLACK, M., MARING, J., MINTZ, M. & SIMMENS, S. J. 2007. Acting on reflection: the effect of reflection on students' clinical performance on a standardized patient examination. *Journal of general internal medicine*, 22, 49-54.
- BLUMBERG, F. C., ALMONTE, D. E., ANTHONY, J. S. & HASHIMOTO, N. J. T. O. H. O. M. P. 2013. Serious games: What are they? What do they do? Why should we play them. 334-351.

- BOCHENNEK, K., WITTEKINDT, B., ZIMMERMANN, S.-Y. & KLINGEBIEL, T. 2007. More than mere games: a review of card and board games for medical education. *Medical teacher*, 29, 941-948.
- BOLTER, J. D., GRUSIN, R. & GRUSIN, R. A. 2000. *Remediation: Understanding new media*, mit Press.
- BONTCHEV, B. P., TERZIEVA, V., PAUNOVA-HUBENOVA, E. J. I. T. & EDUCATION, S. 2021. Personalization of serious games for learning.
- BOOTH, S. E. & KELLOGG, S. B. 2015. Value creation in online communities for educators. *British Journal of Educational Technology*, 46, 684-698.
- BOOVEN, V. 2015. Revisiting the authoritative–dialogic tension in inquiry-based elementary science teacher questioning. *International Journal of Science Education*, 37, 1182-1201.
- BOYLE, E. A., HAINEY, T., CONNOLLY, T. M., GRAY, G., EARP, J., OTT, M., LIM, T., NINAUS, M., RIBEIRO, C. & PEREIRA, J. 2016. An update to the systematic literature review of empirical evidence of the impacts and outcomes of computer games and serious games. *Computers & Education*, 94, 178-192.
- BRATT, A. & FAGERSTRÖM, C. 2020. Self-compassion in old age: confirmatory factor analysis of the 6-factor model and the internal consistency of the Self-compassion scale-short form. *Aging & Mental Health*, 24, 642-648.
- BRESSLER, D. M. & TUTWILER, S. 2021. 'Play Is Serious Learning': Using Mobile Augmented Reality Gaming to Support Science Learning. *Handbook of Research on Innovations in Non-Traditional Educational Practices.* IGI Global.
- BROOKFIELD, S. D. 2015. Critical reflection as doctoral education. *New Directions for Adult and Continuing Education*, 2015, 15-23.
- BROWN, J. S., COLLINS, A. & DUGUID, P. 1989. Situated cognition and the culture of learning. *Educational researcher*, 18, 32-42.
- BUNZ, U., SEIBERT, J. & HENDRICKSE, J. 2020. From TAM to AVRTS: development and validation of the attitudes toward Virtual Reality Technology Scale. *VIRTUAL REALITY*.
- BURGESS, M. L., SLATE, J. R., ROJAS-LEBOUEF, A. & LAPRAIRIE, K. 2010. Teaching and learning in Second Life: Using the Community of Inquiry (CoI) model to support online instruction with graduate students in instructional technology. *The Internet and Higher Education*, 13, 84-88.
- BYBEE, R. W., TAYLOR, J. A., GARDNER, A., VAN SCOTTER, P., POWELL, J. C., WESTBROOK, A. & LANDES, N. J. C. S., CO: BSCS 2006. The BSCS 5E instructional model: Origins and effectiveness. 5, 88-98.

- CHANPRASITCHAI, O.-A. & KHLAISANG, J. 2016. Inquiry-based learning for a virtual learning community to enhance problem-solving ability of applied Thai traditional medicine students. *TOJET: The Turkish Online Journal of Educational Technology*, 15.
- CHENG, M.-T., CHEN, J.-H., CHU, S.-J. & CHEN, S.-Y. J. J. O. C. I. E. 2015. The use of serious games in science education: a review of selected empirical research from 2002 to 2013. 2, 353-375.
- CHERTOFF, D. B., GOLDIEZ, B. & LAVIOLA, J. J. Virtual Experience Test: A virtual environment evaluation questionnaire. 2010 IEEE virtual reality conference (VR), 2010. IEEE, 103-110.
- CLEVELAND-INNES, M. & CAMPBELL, P. 2012. Emotional presence, learning, and the online learning environment. *The International Review of Research in Open and Distributed Learning*, 13, 269-292.
- COHEN, J. 1960. A coefficient of agreement for nominal scales. *Educational and* psychological measurement, 20, 37-46.
- COLE, H., GRIFFITHS, M. D. J. C. & BEHAVIOR 2007. Social interactions in massively multiplayer online role-playing gamers. 10, 575-583.
- CONDLIFFE, B., VISHER, M. G., BANGSER, M. R., DROHOJOWSKA, S. & SACO, L. 2016. Project-based learning: A literature review. *New York, NY: MDRC*.
- CONNOLLY, T. M., BOYLE, E. A., MACARTHUR, E., HAINEY, T. & BOYLE, J. M. 2012. A systematic literature review of empirical evidence on computer games and serious games. *Computers & education*, 59, 661-686.
- CORNILLIE, F., CLAREBOUT, G., DESMET, P. J. P.-S. & SCIENCES, B. 2012. The role of feedback in foreign language learning through digital role playing games. 34, 49-53.
- COUNCIL, N. R. 2011. *Learning science through computer games and simulations*, National Academies Press.
- COYNE, R. J. D. S. 2003. Mindless repetition: Learning from computer games. 24, 199-212.
- CÓZAR-GUTIÉRREZ, R. & SÁEZ-LÓPEZ, J. M. 2016. Game-based learning and gamification in initial teacher training in the social sciences: an experiment with MinecraftEdu. *International Journal of Educational Technology in Higher Education*, 13, 2.
- CRONBACH, L. J. 1951. Coefficient alpha and the internal structure of tests. *psychometrika*, 16, 297-334.
- CROWSTON, K. & PRESTOPNIK, N. R. Motivation and data quality in a citizen science game: A design science evaluation. 2013 46th Hawaii International Conference on System Sciences, 2013. IEEE, 450-459.

- DALGARNO, B. & LEE, M. J. 2010. What are the learning affordances of 3 D virtual environments? *British Journal of Educational Technology*, 41, 10-32.
- DE FREITAS, S. 2006. Learning in immersive worlds: A review of game-based learning.
- DE FREITAS, S. & LIAROKAPIS, F. 2011. Serious games: a new paradigm for education? *Serious games and edutainment applications.* Springer.
- DE LOPE, R. P. & MEDINA-MEDINA, N. 2017. A comprehensive taxonomy for serious games. *Journal of Educational Computing Research*, 55, 629-672.
- DECI, E. L. & RYAN, R. M. J. P. I. 2000. The" what" and" why" of goal pursuits: Human needs and the self-determination of behavior. 11, 227-268.
- DEN HAAN, R.-J. & VAN DER VOORT, M. C. J. S. 2018. On evaluating social learning outcomes of serious games to collaboratively address sustainability problems: A literature review. 10, 4529.
- DETERDING, S., KHALED, R., NACKE, L. E. & DIXON, D. Gamification: Toward a definition. CHI 2011 gamification workshop proceedings, 2011. Vancouver BC, Canada, 1-79.
- DEWEY, J. 1933. How We Think. A Restatement of the Relation of Reflective Thinking to the Educative Process, Boston etc.(DC Heath and Company) 1933.
- DI FUCCIO, R., FERRARA, F. & DI FERDINANDO, A. 2019. The DoCENT role play game: a tool for the training og the digital creativity for teachers.
- DIAO, Y.-H. & CHEN, C.-S. 2020. Research on the Relationship Between Job Competence and Job Well-Being in Service Industry—Based on the Mediating Effect of Job Insecurity. *International Business Research*, 13, 1-10.
- DINGYLOUDI, F. & STRIJBOS, J.-W. Examining value creation in a community of learning practice: Methodological reflections on story-telling and story-reading. Seminar. net, 2015.
- DORJI, U., PANJABUREE, P. & SRISAWASDI, N. 2015. Gender differences in students' learning achievements and awareness through residence energy saving game-based inquiry playing. *Journal of Computers in Education*, 2, 227-243.
- EISINGA, R., TE GROTENHUIS, M. & PELZER, B. 2013. The reliability of a two-item scale: Pearson, Cronbach, or Spearman-Brown? *International journal of public health*, 58, 637-642.
- EKLUND, L. J. J. O. G. & WORLDS, V. 2015. Playing video games together with others: Differences in gaming with family, friends and strangers. 7, 259-277.

- ENOCHSSON, L., ISAKSSON, B., TOUR, R., KJELLIN, A., HEDMAN, L., WREDMARK, T. & TSAI-FELLÄNDER, L. J. J. O. G. S. 2004. Visuospatial skills and computer game experience influence the performance of virtual endoscopy. 8, 874-880.
- ERICKSON, L. V., SAMMONS-LOHSE, D. J. E.-L. & MEDIA, D. 2021. Learning through video games: The impacts of competition and cooperation. 18, 1-17.
- ETEOKLEOUS, N. & KTORIDOU, D. Community of inquiry developed through cloud computing for MIS courses. Global Engineering Education Conference (EDUCON), 2012 IEEE, 2012. IEEE, 1-4.
- FABRICATORE, C., GYAUROV, D. & LOPEZ, X. 2019. An exploratory study of the relationship between meaning-making and quality in games. *Multimedia Tools and Applications*, 78, 13539-13564.
- FARRELL, D. & MOFFAT, D. Applying the self determination theory of motivation in games based Learning. European conference on games based learning, 2014. Academic Conferences International Limited, 118.
- FIRSSOVA, O., KALZ, M., BÖRNER, D., PRINSEN, F., RUSMAN, E., TERNIER, S. & SPECHT, M. Mobile inquiry-based learning with sensor-data in the school: Effects on student motivation. European Conference on Technology Enhanced Learning, 2014. Springer, 112-124.
- FITRIANTO, E. M. 2020. Brand Ambassador Performance and the Effect to Consumer Decision Using VisCAP Model on Online Marketplace in Indonesia. SRIWIJAYA INTERNATIONAL JOURNAL OF DYNAMIC ECONOMICS AND BUSINESS, 4, 21-30.
- FOX, J., GILBERT, M., TANG, W. Y. J. N. M. & SOCIETY 2018. Player experiences in a massively multiplayer online game: A diary study of performance, motivation, and social interaction. 20, 4056-4073.
- FU, K., HAINEY, T. & BAXTER, G. A systematic literature review to identify empirical evidence on the use of computer games in business education and training. European Conference on Games Based Learning, 2016. Academic Conferences International Limited, 232.
- GALSTAUN, V., KENNEDY-CLARK, S. & HU, C. The impact of TPACK on pre-service teacher confidence in embedding ICT into the curriculum areas. EdMedia+ Innovate Learning, 2011. Association for the Advancement of Computing in Education (AACE), 3439-3448.
- GARRISON, D. R. 2011. E-learning in the 21st century: A framework for research and practice, Routledge.

- GARRISON, D. R., ANDERSON, T. & ARCHER, W. 1999. Critical inquiry in a text-based environment: Computer conferencing in higher education. *The internet and higher education*, 2, 87-105.
- GAY, P. 1992. LR & Diehl, "Research Methods for Business and Management New York. MacMillan Publishing Company," New York MacMillan Publ. Co.
- GEE, J. P. 2007. What video games have to teach us about learning and literacy, Basingstoke;New York;, Palgrave Macmillan.
- GIDHAGEN, M., RIDELL, O. P. & SÖRHAMMAR, D. J. M. S. Q. A. I. J. 2011. The orchestrating firm: value creation in the video game industry.
- GILLIES, R. M. & NICHOLS, K. 2015. How to support primary teachers' implementation of inquiry: Teachers' reflections on teaching cooperative inquiry-based science. *Research in Science Education*, 45, 171-191.
- GNAUK, B., DANNECKER, L. & HAHMANN, M. Leveraging gamification in demand dispatch systems. Proceedings of the 2012 Joint EDBT/ICDT workshops, 2012. 103-110.
- GORSUCH, R. L. & ORTBERG, J. 1983. Moral obligation and attitudes: Their relation to behavioral intentions. *Journal of Personality and Social Psychology*, 44, 1025.
- GRAHAM, C. R. 2011. Theoretical considerations for understanding technological pedagogical content knowledge (TPACK). *Computers & Education*, 57, 1953-1960.
- GUY, B., BIDWELL, N. J. & MUSUMECI, P. Gameplan: Serious gaming for place making. Proceedings of the second Australasian conference on Interactive entertainment, 2005. 252.
- GYAUROV, D., FABRICATORE, C. & LOPEZ, X. An analysis instrument for gameplay information flows supporting sustainability complex problem-solving. 13th European Conference on Games Based Learning, 2019. Academic Conferences and Publishing International, 863-871.
- HAINEY, T., CONNOLLY, T. M., BOYLE, E. A., WILSON, A. & RAZAK, A. 2016. A systematic literature review of games-based learning empirical evidence in primary education. *Computers & Education*, 102, 202-223.
- HAMARI, J., SHERNOFF, D. J., ROWE, E., COLLER, B., ASBELL-CLARKE, J. & EDWARDS, T. J. C. I. H. B. 2016. Challenging games help students learn: An empirical study on engagement, flow and immersion in game-based learning. 54, 170-179.

- HAO, K.-C. & LEE, L.-C. J. I. L. E. 2019. The development and evaluation of an educational game integrating augmented reality, ARCS model, and types of games for English experiment learning: an analysis of learning. 1-14.
- HARRIS, J., MISHRA, P. & KOEHLER, M. 2009. Teachers' technological pedagogical content knowledge and learning activity types: Curriculum-based technology integration reframed. *Journal of research on technology in education*, 41, 393-416.
- HAYAT, E., AMARATUNGA, D. & HAIGH, R. 2013. Research Approach to Evaluate Post-Disaster Road Reconstruction in Indonesia.
- HAYES, S., SMITH, S. U. & SHEA, P. 2015. Expanding learning presence to account for the direction of regulative intent: self-, co-and shared regulation in online learning. *Online Learning*, 19, 15-31.
- HERZ, J. C. 1997. Joystick nation: How videogames ate our quarters, won our hearts, and rewired our minds, Atlantic/Little, Brown.
- HOOKHAM, G. & NESBITT, K. A systematic review of the definition and measurement of engagement in serious games. Proceedings of the Australasian Computer Science Week Multiconference, 2019. 1-10.
- HOWARD, T. C. J. T. I. P. 2003. Culturally relevant pedagogy: Ingredients for critical teacher reflection. 42, 195-202.
- HSIAO, H.-C. A brief review of digital games and learning. 2007 First IEEE International Workshop on Digital Game and Intelligent Toy Enhanced Learning (DIGITEL'07), 2007. IEEE, 124-129.
- HU, C. & FYFE, V. 2010. Impact of a new curriculum on pre-service teachers' Technical, Pedagogical and Content Knowledge (TPACK). *Curriculum, technology & transformation for an unknown future. Proceedings ascilite Sydney*, 185-189.
- HUANG, Y.-M., LIAO, Y.-W., HUANG, S.-H., CHEN, H.-C. & SOCIETY 2014. Jigsawbased cooperative learning approach to improve learning outcomes for mobile situated learning. *Journal of Educational Technology*, 17, 128-140.
- HUNG, H.-T., YANG, J. C., HWANG, G.-J., CHU, H.-C. & WANG, C.-C. 2018. A scoping review of research on digital game-based language learning. *Computers & Education*.
- HWANG, G.-J., CHIU, L.-Y. & CHEN, C.-H. 2015. A contextual game-based learning approach to improving students' inquiry-based learning performance in social studies courses. *Computers & Education*, 81, 13-25.
- HWANG, G.-J., CHU, H.-C., LIN, Y.-S. & TSAI, C.-C. 2011. A knowledge acquisition approach to developing Mindtools for organizing and sharing differentiating knowledge in a ubiquitous learning environment. *Computers Education*, 57, 1368-1377.

- HWANG, G.-J., TSAI, C.-C., CHU, H.-C., KINSHUK, K. & CHEN, C.-Y. 2012. A contextaware ubiquitous learning approach to conducting scientific inquiry activities in a science park. *Australasian Journal of Educational Technology*, 28.
- HWANG, G. J. & CHEN, C. H. 2017. Influences of an inquiry based ubiquitous gaming design on students' learning achievements, motivation, behavioral patterns, and tendency towards critical thinking and problem solving. *British Journal of Educational Technology*, 48, 950-971.
- HWANG, G. J. & WU, P. H. 2012. Advancements and trends in digital game based learning research: a review of publications in selected journals from 2001 to 2010. *British Journal of Educational Technology*, 43, E6-E10.
- JACOBS, R. S. J. M. & COMMUNICATION 2021. Winning over the players: Investigating the motivations to play and acceptance of serious games. 9, 28-38.
- JANČIČ, P. & HUS, V. J. I. J. O. G.-B. L. 2018. Teaching Social Studies With Games. 8, 68-79.
- JANTKE, K. P. Toward a taxonomy of game based learning. 2010 IEEE International Conference on Progress in Informatics and Computing, 2010. IEEE, 858-862.
- JARVENPAA, S. L. 1989. The effect of task demands and graphical format on information processing strategies. *Management science*, 35, 285-303.
- JOHNSON, C. I. & MAYER, R. E. J. C. I. H. B. 2010. Applying the self-explanation principle to multimedia learning in a computer-based game-like environment. 26, 1246-1252.
- JOHNSON, L. L., KIM, G. M. J. E. T. P. & CRITIQUE 2021. Experimenting with game-based learning in preservice teacher education.
- JUNUS, K., SANTOSO, H. B., SADITA, L., R-SURADIJONO, S. H. & SUHARTANTO, H. The Community of Inquiry model training for beginners: Patterns of interaction and student learning strategies. 2017 7th World Engineering Education Forum (WEEF), 2017. IEEE, 343-348.
- KAISER, H. F. 1958. The varimax criterion for analytic rotation in factor analysis. *Psychometrika*, 23, 187-200.
- KALZ, M., FIRSSOVA, O., BÖRNER, D., TERNIER, S., PRINSEN, F., RUSMAN, E., DRACHSLER, H. & SPECHT, M. Mobile inquiry-based learning for sustainability education in secondary schools. 2014 IEEE 14th International Conference on Advanced Learning Technologies, 2014. IEEE, 644-646.
- KELES, E. 2018. Use of Facebook for the Community Services Practices course: Community of inquiry as a theoretical framework. *Computers & Education*, 116, 203-224.

- KEMBER, D. 1997. A reconceptualisation of the research into university academics' conceptions of teaching. *Learning and instruction*, 7, 255-275.
- KENNEDY-CLARK, S., GALSTAUN, V. & ANDERSON, K. Using game-based inquiry learning to meet the changing directions of science education. ASCILITE, 2011.
- KENNEDY-CLARK, S., GALSTAUN, V. & ANDERSON, K. 2013. Death in Rome: Using an Online Game for. *Cases on Digital Game-Based Learning: Methods, Models, and Strategies: Methods, Models, and Strategies*, 364.
- KING, C.-H. 2018. Enhancing Science Self-Efficacy of Young Females: An Exploratory Study Using Virtual Reality RPG Game.
- KLOPFER, E. & SQUIRE, K. 2008. Environmental Detectives—the development of an augmented reality platform for environmental simulations. *Educational technology research and development*, 56, 203-228.
- KOGAN, M. & LAURSEN, S. L. 2014. Assessing long-term effects of inquiry-based learning: A case study from college mathematics. *Innovative higher education*, 39, 183-199.
- KOSTIKOVA, I., HOLUBNYCHA, L., SHCHOKINA, T., SOROKA, N., BUDIANSKA, V. & MARYKIVSKA, H. 2019. A role-playing game as a means of effective professional english teaching.
- KOZAN, K. & RICHARDSON, J. C. 2014. Interrelationships between and among social, teaching, and cognitive presence. *The Internet and higher education*, 21, 68-73.
- KUHLTHAU, C. C., MANIOTES, L. K. & CASPARI, A. K. 2015. *Guided inquiry: Learning in the 21st century: Learning in the 21st century*, ABC-CLIO.
- KUHN, D., BLACK, J., KESELMAN, A. & KAPLAN, D. 2000. The development of cognitive skills to support inquiry learning. *Cognition and Instruction*, 18, 495-523.
- LAAMARTI, F., EID, M. & EL SADDIK, A. J. I. J. O. C. G. T. 2014. An overview of serious games. 2014.
- LAMERAS, P. & MOUMOUTZIS, N. Towards the gamification of inquiry-based flipped teaching of mathematics a conceptual analysis and framework. 2015 2015. IEEE, 343-347.
- LAMERAS, P., PETRIDIS, P., TORRENS, K., DUNWELL, I., HENDRIX, M. & ARNAB, S. Training science teachers to design inquiry-based lesson plans through a serious game. Proceedings of the Sixth International Conference on Mobile, Hybrid and Online Learning, 2014. 86-91.
- LANDERS, R. N. J. S. & GAMING 2014. Developing a theory of gamified learning: Linking serious games and gamification of learning. 45, 752-768.

- LEE, H.-S. & BUTLER, N. 2003. Making authentic science accessible to students. International Journal of Science Education, 25, 923-948.
- LIM, B. R. 2004. Challenges and issues in designing inquiry on the Web. *British Journal of Educational Technology*, 35, 627-643.
- LIN, J. J., MAMYKINA, L., LINDTNER, S., DELAJOUX, G. & STRUB, H. B. Fish'n'Steps: Encouraging physical activity with an interactive computer game. International conference on ubiquitous computing, 2006. Springer, 261-278.
- LONKA, K., SLOTTE, V., HALTTUNEN, M., KURKI, T., TIITINEN, A., VAARA, L. & PAAVONEN, J. 2001. Portfolios as a learning tool in obstetrics and gynaecology undergraduate training. *Medical education*, 35, 1125-1130.
- LOPAREV, A., LASECKI, W. S., MURRAY, K. I. & BIGHAM, J. P. 2014. Introducing shared character control to existing video games.
- LORENZINI, C., FAITA, C., BARSOTTI, M., CARROZZINO, M., TECCHIA, F. & BERGAMASCO, M. ADITHO–A Serious Game for Training and Evaluating Medical Ethics Skills. International Conference on Entertainment Computing, 2015. Springer, 59-71.
- MAMEDE, S., SCHMIDT, H. G. & PENAFORTE, J. C. 2008. Effects of reflective practice on the accuracy of medical diagnoses. *Medical education*, 42, 468-475.
- MARCHAND, A. & HENNIG-THURAU, T. J. J. O. I. M. 2013. Value creation in the video game industry: Industry economics, consumer benefits, and research opportunities. 27, 141-157.
- MATIJEVIĆ, M. & TOPOLOVČAN, T. 2019. Informal Learning among Teenagers through Video Games. *The Journal of Elementary Education*, 12, 1-25.
- MAYER, R. E. 2019. Computer games in education. *Annual review of psychology*, 70, 531-549.
- MCHUGH, M. L. 2012. Interrater reliability: the kappa statistic. *Biochemia medica: Biochemia medica*, 22, 276-282.
- MEDEMA, W., FURBER, A., ADAMOWSKI, J., ZHOU, Q. & MAYER, I. J. W. 2016. Exploring the potential impact of serious games on social learning and stakeholder collaborations for transboundary watershed management of the St. Lawrence River Basin. 8, 175.
- MEESUK, K. & SRISAWASDI, N. Implementation of student-associated game-based open inquiry in chemistry education: Results on students' perception and motivation. Proceedings of the 22nd International Conference on Computers in Education, 2014. 219-226.

- MELETIOU-MAVROTHERIS, M. & PRODROMOU, T. 2016. Pre-Service Teacher Training on Game-Enhanced Mathematics Teaching and Learning. *Technology, Knowledge and Learning*, 21, 379-399.
- MELNIKOVAS, A. 2018. Towards an explicit research methodology: Adapting research onion model for futures studies. *Journal of Futures Studies*, 23, 29-44.
- MEZIROW, J. 1990. Fostering critical reflection in adulthood, Jossey-Bass Publishers San Francisco.
- MICHAEL, D. R. & CHEN, S. L. 2005. Serious games: Games that educate, train, and inform, Muska & Lipman/Premier-Trade.
- MICHAUD, L., ALVAREZ, J. J. A., EDUGAMING, TRAINING... IDATE CONSULTING & RESEARCH 2008. Serious games.
- MIKROYANNIDIS, A., OKADA, A., CORREA, A. & SCOTT, P. 2016. Inquiry-Based Learning on the Cloud. Handbook of Research on Cloud-Based STEM Education for Improved Learning Outcomes, 291-310.
- MILES, M. B., HUBERMAN, A. M. & SALDAÑA, J. 2014. Qualitative data analysis: A methods sourcebook. 3rd. *ed: Thousand Oaks, CA: Sage*.
- MITCHELL, A. & SAVILL-SMITH, C. J. A. R. O. T. L. 2004. The use of computer and video games for learning.
- MOORE, G. C. & BENBASAT, I. 1991. Development of an instrument to measure the perceptions of adopting an information technology innovation. *Information systems research*, 2, 192-222.
- MOREAU, C., BARNAUD, C. & MATHEVET, R. J. S. 2019. Conciliate agriculture with landscape and biodiversity conservation: a role-playing game to explore trade-offs among ecosystem services through social learning. 11, 310.
- MORRIS, T. & WOOD, S. 1991. Testing the survey method: continuity and change in British industrial relations. *Work, Employment and Society*, 5, 259-282.
- MURO, M. & JEFFREY, P. 2008. A critical review of the theory and application of social learning in participatory natural resource management processes. *Journal of environmental planning and management*, 51, 325-344.
- NEWMAN, J. 2013. Videogames, London, Routledge.
- NOVICK, L. R. 1988. Analogical Transfer, Problem Similarity, and Expertise. Journal of experimental psychology. Learning, memory, and cognition, 14, 510-520.
- NRC 2000. Inquiry and theNational Science Education Standards. *Washington, DC: National Academy Press.*

- O NEIL, H. 2007. Computer games and team and individual learning, Elsevier.
- OKADA, A., KOWALSKI, R. P. G., KIRNER, C. & TORRES, P. L. J. I. L. E. 2019. Factors influencing teachers' adoption of AR inquiry games to foster skills for Responsible Research and Innovation. 27, 324-335.
- OLIVER, R. 2008. Engaging first year students using a web-supported inquiry-based learning setting. *Higher Education*, 55, 285.
- PADILLA, B. I. & KREIDER, K. E. 2018. Community of inquiry framework for advanced practice Nursing students. *The Journal for Nurse Practitioners*, 14, e87-e92.
- PALOMBA, A. 2020. How high brand loyalty consumers achieve relationships with virtual worlds and its elements through presence. *Journal of Media Business Studies*, 1-18.
- PARDALES, M. J. & GIROD, M. 2006. Community of Inquiry: Its past and present future. *Educational Philosophy and Theory*, 38, 299-309.
- PEDASTE, M., MÄEOTS, M., SIIMAN, L. A., DE JONG, T., VAN RIESEN, S. A., KAMP, E. T., MANOLI, C. C., ZACHARIA, Z. C. & TSOURLIDAKI, E. 2015. Phases of inquiry-based learning: Definitions and the inquiry cycle. *Educational research review*, 14, 47-61.
- PELLAS, N. & BOUMPA, A. 2017. Blending the CoI model with Jigsaw technique for preservice foreign language teachers' continuing professional development using Open Sim and Sloodle. *Education and Information Technologies*, 22, 939-964.
- PERTTULA, A., KIILI, K., LINDSTEDT, A. & TUOMI, P. 2017. Flow experience in game based learning–a systematic literature review. *International Journal of Serious Games*, 4, 57-72.
- PETERSEN, N., DE BEER, J. & DUNBAR-KRIGE, H. 2011. Use of a simulation game for HIV/AIDS education with pre-service teachers. *African Journal of AIDS Research*, 10, 73-81.
- PINHO, C., FRANCO, M. & MENDES, L. 2020. Exploring the conditions of success in elibraries in the higher education context through the lens of the social learning theory. *Information & Management*, 57, 103208.
- PLASS, J. L., HOMER, B. D. & KINZER, C. K. J. E. P. 2015. Foundations of game-based learning. 50, 258-283.
- PRENSKY, M. 2003. Digital game-based learning. *Computers in Entertainment (CIE)*, 1, 21-21.
- PRESTRIDGE, S. 2019. Categorising teachers' use of social media for their professional learning: A self-generating professional learning paradigm. *Computers & Education*, 129, 143-158.

- PRINCE, M. J. & FELDER, R. M. 2006. Inductive teaching and learning methods: Definitions, comparisons, and research bases. *Journal of engineering education*, 95, 123-138.
- QUANDT, T., CHEN, V., MÄYRÄ, F. & VAN LOOY, J. 2014. (Multiplayer) gaming around the globe? A comparison of gamer surveys in four countries. *Multi-player: the social aspects of digital gaming gaming.* Routledge.
- QUINTANA, C., REISER, B. J., DAVIS, E. A., KRAJCIK, J., FRETZ, E., DUNCAN, R. G., KYZA, E., EDELSON, D. & SOLOWAY, E. 2004. A scaffolding design framework for software to support science inquiry. *The journal of the learning sciences*, 13, 337-386.
- RAES, A., SCHELLENS, T., DE WEVER, B. & VANDERHOVEN, E. 2012. Scaffolding information problem solving in web-based collaborative inquiry learning. *Computers Education*, 59, 82-94.
- RAESSENS, J. & GOLDSTEIN, J. 2005. Handbook of computer game studies, The MIT Press.
- REED, M. S., EVELY, A. C., CUNDILL, G., FAZEY, I., GLASS, J., LAING, A., NEWIG, J., PARRISH, B., PRELL, C. & RAYMOND, C. 2010. What is social learning? *Ecology and society*, 15.
- REMENYI, D., WILLIAMS, B., MONEY, A. & SWARTZ, E. 1998. Doing research in business and management: an introduction to process and method, Sage.
- RUŽIC-BAF, M., STRNAK, H. & DEBELJUH, A. 2016. Online Video Games and Young People. *International Journal of Research in Education and Science*, 2, 94-103.
- RYAN, R. M., RIGBY, C. S., PRZYBYLSKI, A. J. M. & EMOTION 2006. The motivational pull of video games: A self-determination theory approach. 30, 344-360.
- SABOURIN, J., ROWE, J., MOTT, B. W. & LESTER, J. C. Exploring inquiry-based problemsolving strategies in game-based learning environments. International Conference on Intelligent Tutoring Systems, 2012. Springer, 470-475.
- SAHAY, A. J. R. G., ART 2016. Peeling Saunder's research onion. 1-5.
- SALVINI, G., VAN PAASSEN, A., LIGTENBERG, A., CARRERO, G., BREGT, A. J. E. S. & POLICY 2016. A role-playing game as a tool to facilitate social learning and collective action towards Climate Smart Agriculture: Lessons learned from Apuí, Brazil. 63, 113-121.
- SAN CHEE, Y., TAN, K. C. D., TAN, E. M. & JAN, M. Learning Chemistry Through Inquiry With the Game Legends of Alkhimia: An Evaluation of Learning Outcomes. Proceedings of the 7th European Conference on Management Leadership and Governance: ECGBL 2011, 2011. Academic Conferences Limited, 98.

- SARIC, M. & STEH, B. J. C. J. 2017. Critical reflection in the professional development of teachers: Challenges and possibilities. 7, 67-85.
- SASS, D. A. 2010. Factor loading estimation error and stability using exploratory factor analysis. *Educational and Psychological Measurement*, 70, 557-577.
- SAUNDERS, M., LEWIS, P. & THORNHILL, A. 2003. Research methods forbusiness students. *Essex: Prentice Hall: Financial Times*.
- SAUNDERS, M., LEWIS, P. & THORNHILL, A. 2009. Research methods for business students, Pearson education.
- SECUNDO, G. & GRIPPA, F. 2010. Designing, managing and assessing a Web 2.0 learning community to enhance inquiry based learning. *International Journal of Web Based Communities*, 6, 164-182.
- SEIXAS, P. 1993. The community of inquiry as a basis for knowledge and learning: The case of history. *American Educational Research Journal*, 30, 305-324.
- SETIANI, M. Y. & MACKINNON, A. M. 2015. A community of inquiry-based framework for civic education at Universitas Terbuka, Indonesia. *Distance Education*, 36, 351-363.
- SHAFFER, D. W. 2006. How computer games help children learn, Macmillan.
- SHEA, P., HAYES, S., UZUNER SMITH, S., VICKERS, J., BIDJERANO, T., GOZZA-COHEN, M., JIAN, S.-B., PICKETT, A. M., WILDE, J. & TSENG, C.-H. 2013. Online learner self-regulation: Learning presence viewed through quantitative content-and social network analysis. *International Review of Research in Open and Distributed Learning*, 14, 427-461.
- SHIELDS, P. M. 2003. The community of inquiry: Classical pragmatism and public administration. *Administration & Society*, 35, 510-538.
- SINHA, T., CLARKE, S. & FARQUHARSON, L. Shrek, Saunders and the Onion Myth: Using Myths, Metaphors and Storytelling. Proceedings of the 17th European Conference on Research Methodology for Business and Management Studies, 2018. 366.
- SKINNER, B. F. 1965. Science and human behavior, Simon and Schuster.
- SMITH, E. & GOLDING, L. 2018. Use of board games in higher education literature review. *MSOR Connections*, 16.
- SMITH, S. U., HAYES, S. & SHEA, P. 2017. A Critical Review of the Use of Wenger's Community of Practice (CoP) Theoretical Framework in Online and Blended Learning Research, 2000-2014. *Online Learning*, 21, 209-237.
- SQUIRE, K. & BARAB, S. Replaying history: Engaging urban underserved students in learning world history through computer simulation games. Proceedings of the 6th

international conference on Learning sciences, 2004. International Society of the Learning Sciences, 505-512.

- SQUIRE, K., JOHNSON, K., HOLLAND, W., NATAF, Z. & KLOPFER, E. 2002. MIT games-to-teach: Design document for: A platform for augmented reality gaming. *Environmental Detectives*.
- SQUIRE, K. D. J. E. T. 2008. Video games and education: Designing learning systems for an interactive age. 17-26.
- SRISAWASDI, N., PANJABUREE, P. J. J. O. S. E. & TECHNOLOGY 2019. Implementation of game-transformed inquiry-based learning to promote the understanding of and motivation to learn chemistry. 28, 152-164.
- STAVROULIA, K. E., MAKRI-BOTSARI, E., PSYCHARIS, S. & KEKKERIS, G. 2016. Emotional experiences in simulated classroom training environments. *International Journal of Information and Learning Technology*, 33, 172-185.
- STEINKUEHLER, C. 2010. Video games and digital literacies. *Journal of Adolescent & Adult Literacy*, 54, 61-63.
- STENBOM, S., HRASTINSKI, S. & CLEVELAND-INNES, M. 2016a. Emotional presence in a relationship of inquiry: The case of one-to-one online math coaching. *Online Learning*, 20, 41-56.
- STENBOM, S., JANSSON, M. & HULKKO, A. 2016b. Revising the community of inquiry framework for the analysis of one-to-one online learning relationships. *International Review of Research in Open and Distributed Learning*, 17, 36-53.
- SUAREZ, A., TERNIER, S., PRINSEN, F. & SPECHT, M. DojoIBL: nurturing communities of inquiry. European Conference on Technology Enhanced Learning, 2016. Springer, 533-536.
- SUÁREZ, Á., TERNIER, S., PRINSEN, F. & SPECHT, M. Nurturing communities of inquiry: A formative study of the DojoIBL platform. European Conference on Technology Enhanced Learning, 2016. Springer, 292-305.
- SULS, J., MARTIN, R. & WHEELER, L. J. C. D. I. P. S. 2002. Social comparison: Why, with whom, and with what effect? 11, 159-163.
- SUSI, T., JOHANNESSON, M. & BACKLUND, P. 2007. Serious games: An overview.
- TARDINI, S. & CANTONI, L. A semiotic approach to online communities: Belonging, interest and identity in websites' and videogames' communities. Proceedings of the IADIS International Conference e-Society, 2005. 371-8.
- THOMAS, D. & BROWN, J. S. 2011. A new culture of learning: Cultivating the imagination for a world of constant change, CreateSpace Lexington, KY.

- TREPTE, S., REINECKE, L. & JUECHEMS, K. J. C. I. H. B. 2012. The social side of gaming: How playing online computer games creates online and offline social support. 28, 832-839.
- TURCHET, L., NILSSON, N. & SERAFIN, S. Inside the boundaries of the physical world: audio-haptic feedback as support for the navigation in virtual environments. International Conference on Human Haptic Sensing and Touch Enabled Computer Applications, 2012. Springer, 577-588.
- TYCHSEN, A. Role playing games: comparative analysis across two media platforms. Proceedings of the 3rd Australasian conference on Interactive entertainment, 2006. Murdoch University, 75-82.
- UCAR, S. & TRUNDLE, K. C. 2011. Conducting guided inquiry in science classes using authentic, archived, web-based data. *Computers & Education*, 57, 1571-1582.
- VALKOV, D. & FLAGGE, S. Smooth immersion: the benefits of making the transition to virtual environments a continuous process. Proceedings of the 5th Symposium on Spatial User Interaction, 2017. 12-19.
- VAN AMERSFOORT, D., NIJLAND, F. & DE LAAT, M. Appreciating networked learning: Value creation in practice. Proceedings of the 8th International Conference on Networked Learning, 2012. 577-578.
- VAN ECK, R. 2006. Digital game-based learning: It's not just the digital natives who are restless. *EDUCAUSE review*, 41, 16.
- VLACHOPOULOS, D. & MAKRI, A. J. I. J. O. E. T. I. H. E. 2017. The effect of games and simulations on higher education: a systematic literature review. 14, 1-33.
- VOGEL, J. J., VOGEL, D. S., CANNON-BOWERS, J., BOWERS, C. A., MUSE, K. & WRIGHT, M. J. J. O. E. C. R. 2006. Computer gaming and interactive simulations for learning: A meta-analysis. 34, 229-243.
- VORDERER, P. & BRYANT, J. 2012. Playing video games: Motives, responses, and consequences, Routledge.
- VRASIDAS, C. & SOLOMOU, M. 2013. Using educational design research methods to examine the affordances of online games for teacher learning. *Educational Media International*, 50, 192-205.
- WANG, H.-Y., DUH, H. B.-L., LI, N., LIN, T.-J. & TSAI, C.-C. 2014. An investigation of university students' collaborative inquiry learning behaviors in an augmented reality simulation and a traditional simulation. *Journal of Science Education and Technology*, 23, 682-691.

- WANG, Y., FANG, W.-C., HAN, J. & CHEN, N.-S. 2016. Exploring the affordances of WeChat for facilitating teaching, social and cognitive presence in semi-synchronous language exchange. *Australasian Journal of Educational Technology*, 32.
- WEBB, M. & COX, M. 2004. A review of pedagogy related to information and communications technology. *Technology, pedagogy and education,* 13, 235-286.
- WENGER, E., TRAYNER, B. & DE LAAT, M. 2011. Promoting and assessing value creation in communities and networks: A conceptual framework. Rapport.
- WESSELOW, M. & STOLL KLEEMANN, S. J. T. G. J. 2018. Role playing games in natural resource management and research: Lessons learned from theory and practice. 184, 298-309.
- WILSON, C. D., TAYLOR, J. A., KOWALSKI, S. M. & CARLSON, J. 2010. The relative effects and equity of inquiry - based and commonplace science teaching on students' knowledge, reasoning, and argumentation. *Journal of Research in Science Teaching: The Official Journal of the National Association for Research in Science Teaching*, 47, 276-301.
- WOLF, M. J. 2001. Genre and the video game. The medium of the video game, 1.
- WOOLFOLK, A. E. 1998. Readings in educational psychology, ERIC.
- YANG, J. C., QUADIR, B. J. J. O. E. T. & SOCIETY 2018. Effects of prior knowledge on learning performance and anxiety in an English learning online role-playing game. 21, 174-185.
- Y1LMAZ, N. & OLGAN, R. 2020. Perceived attributes of instructional computer use in early childhood education: A scale adaptation and validation study. *Elementary Education Online*, 19, 1267-1283.
- YIN, R. K. 1989. Research design issues in using the case study method to study management information systems. *The information systems research challenge: Qualitative research methods*, 1, 1-6.
- YIN, R. K. 2003. Case study research: Design and methods (Vol. 5).
- YOON, H., JOUNG, Y. J., KIM, M 2012. The challenges of science inquiry teaching for preservice teachers in elementary classrooms: Difficulties on and under the scene. *Research in Science & Technological Education*, 42(3), 589–608.
- ZHANG, J., FENG, X. & CHAN, S. 2011. Development of an instrument to measure flow experience in computer game play. *Proceedings of Pre-ICIS HCI/MIS Work*.

ZHANG, Y., SONG, H., LIU, X., TANG, D., CHEN, Y.-E. & ZHANG, X. J. F. I. H. N. 2019. Corrigendum: Language Learning Enhanced by Massive Multiple Online Role-Playing Games (MMORPGs) and the Underlying Behavioral and Neural Mechanisms. 13, 307.

ZYDA, M. J. C. 2005. From visual simulation to virtual reality to games. 38, 25-32.

# Appendix A Feasibility of Using Serious Games for Teachers' Professional Development

As the result aforementioned (Chapter 4.1) that few studies involve utilising inquiry-based learning (IBL)-serious games to support teacher training, the author hopes from a broader perspective to find out the evidence of empirical studies that how serious games impact teachers' professional development. This research was carried out as follows.

### A.1 Introduction

Both digital and non-digital games have been leveraged for over three decades to foster learning at all stages of formal education (e.g. Connolly et al., 2012; Smith and Golding, 2018), in professional development contexts (e.g. Bochennek et al., 2007; Fu, Hainey and Baxter, 2016), and in informal environments (e.g. Matijevic and Topolovcan, 2019; Ruzicbaf et al, 2016). Research has amply theorised and demonstrated that games can effectively enhance learners' motivation, engagement, and learning outcomes in multiple domains (Squire, 2008, Boyle et al., 2016, Mayer, 2019, Fabricatore et al., 2019, De Freitas, 2006, Bayeck, 2020).

Teachers' professional development represents a particularly relevant scenario for the application of learning-oriented games. The modern role of teachers requires adapting to changing and increasingly complex contexts and demands and involves responsibilities that transcend the mere act of conveying information to students and assessing their coursework. Nowadays, teachers have to leverage pedagogical, technological, domain and context

knowledge in order to (i) prepare, convey and deploy appropriate contents and resources in learning environments (Graham, 2011, Harris et al., 2009, Galstaun et al., 2011, Hu and Fyfe, 2010), and (ii) guide students so that they can actively and autonomously construct knowledge (Biggs, 1996). These complex responsibilities require teachers to continuously develop, update and integrate knowledge, skills and attitudes relevant to tackle the scenarios that they face. This in turn requires novel teacher training approaches and tools (Webb and Cox, 2004, Kennedy-Clark et al., 2011).

Games represent an important asset in this context, as they can promote the development of transferrable knowledge, skills and attitudes by engaging players in activities driven by meaning-making, and set in scenarios mirroring real-world situations (Fabricatore et al., 2019). Indeed, games have been successfully leveraged to enhance teachers' professional development, facilitating the acquisition of domain-specific knowledge and skills (Meletiou-Mavrotheris and Prodromou, 2016, Alyaz and Genc, 2016, Lameras et al., 2014), increasing their motivation and engagement (Vrasidas and Solomou, 2013, Cózar-Gutiérrez and Sáez-López, 2016), and fostering positive attitudes towards modern pedagogical approaches and tools such as game-based learning (Kennedy-Clark et al., 2011, Annetta et al., 2014). However, to this knowledge, there is no systematic synthesis of evidence on the impacts of games in teacher education. This review addresses such a gap, aiming to better understand how serious games (SGs) can be used to enhance teachers' professional development of teachers' professional development, and with what effects. To this end, this review addressed the following questions:

- (a) In what contexts have SGs been used for teacher education?
- (b) What types of games have been used for teacher education?
- (c) What impacts can SGs have on teachers' professional development?

### A.2 Method

#### A.2.1 Study selection

This review focussed on empirical studies reporting the effects of the use of SGs in teacher education. SGs were conceptualised as digital games designed to promote behaviour change and/or the development of knowledge, skills and attitudes (Hainey et al., 2016, Connolly et al., 2012, De Lope and Medina-Medina, 2017). The author considered game-based learning products as forms of serious games, given that game-based learning is usually synonymised with or used as a subcategory of serious games (Connolly et al., 2012, Hainey et al., 2016).

Studies for inclusion in this review were identified through a systematic database search in Scopus, using the following query: (teacher AND (training OR (professional AND development))) AND ((serious AND games) OR (learning AND games) OR (game-based AND learning) OR (digital AND game-based)). Searches were performed in title, abstract or keywords. The author looked for the journal or conference articles, written in English, and published between 2010 and 2020. Automatic lemmatisation was used to ensure the inclusion of relevant variants of keywords (e.g. including "games" automatically triggered the search for "game"). The search yielded a total of 471 articles. Titles and abstracts were screened by two reviewers, to identify duplicate and non-pertinent documents. As a result, 400 articles were excluded. Titles, abstract and full bodies of the remaining 71 articles were then reviewed, to identify studies for final review. Articles were excluded if they met any

of the following criteria: (i) not reporting on applications of digital SGs (as defined above) in teacher education; (ii) not reporting qualitative or quantitative outcomes/impacts; (iii) focussed on interventions and/or data already covered in other selected articles. As a result, 63 articles were excluded, and 8 were kept for full analysis.

### A.2.2 Data analysis

Two researchers independently reviewed and coded the selected articles, extracting data using specially-created forms. In order to address RQ1, articles were coded through a deductive approach (Miles et al., 2014), identifying the description, purpose and country of the game-based teacher education intervention, the settings in which SGs were available to play, and key characteristics of the involved participants. To address RQ2, articles were analysed through an inductive coding process (Miles et al., 2014), focussing information describing the SGs used, and identifying key types of: (i) educational contents conveyed through SGs; (ii) hardware platforms on which SGs were used; and (iii) gameplay features, including roles impersonated by players, learning-conducive tasks involved in the game, settings in which gameplay unfolded, and virtual representations of players in the game space. RQ3 was addressed through further inductive coding, analysing information

Both manifest and latent contents were analysed in all cases. Latent contents concerning game genres were coded based on Herz' (1997) taxonomy of games. Coding of latent contents was subjected to inter-coder reliability testing through Cohen's kappa (Cohen,

1960), obtaining kappa scores consistently equal to 0.88. This indicated a high level of agreement between coders (McHugh, 2012). Disagreements were resolved through discussion to consensus.

### A.3 Findings

### A.3.1 Contexts of use of SGs

Table A.1 systematises the findings addressing Q1. The reviewed studies were mainly conducted in European countries (n=6). Only two studies were carried out before 2014. Most of the studies (n=5) used games only in formal teacher education settings (classrooms). In most cases (n=3) studies were focused on science education. Most studies targeted exclusively prospective teachers (n=5). All studies reportedly involved young people (n=4). The SGs involved mainly role-playing (n=6) and simulation (n=7) gameplay and contextualisation aspects. In most cases (n=6) SGs were used aiming at improving specific teachers' skills and knowledge, with a particular focus on classroom management capabilities (n=3). In all these cases, players acted as virtual teachers in simulated formal education environments (e.g. virtual classrooms) and interacted with virtual students (AI agents).
	Description of the same based teacher	Purpose of the		Setting where	Cha	Characteristics of participan		pants
Study	Description of the game-based teacher education intervention	educational intervention	Country	the game was accessible	Age	Gender	Subject area	Status
Di Fuccio et al. (2019)	Role-playing SG simulating real-world classroom environment developed and used within the DoCENT (Digital Creativity Enhanced in Teacher education) training programme. Game facing players/teachers with realistic situations requiring them to interact with virtual students (AI agents).	Enhance teachers' ability to apply digital creativity strategies in schools.	Europe	not reported	not reported	not reported	Not reported	pre- service
Alyaz and Genc (2016)	Adventure SG "A Mysterious Mission" set in a fictional context used as a scenario- based training environment.	Improve teachers' language skills.	Turkey	formal (classroom); informal	22.33 average	88% female, 12% male	German language	pre- service
Aivelo and Uitto (2016)	Augmented reality (AR) simulation SG "Parasite Race" set in a fictional context, used as a scenario-based training environment.	Promote exploration of how AR may facilitate the learning of conceptual models.	Finland	formal (classroom)	mostly young	86% female, 14% male	Biology	in- service
Stavroulia et al. (2016)	Role-playing SG "SimSchool" simulating real-world classroom environment. Game facing players/teachers with realistic situations, requiring them to interact with virtual students (AI agents).	Enhance teachers' classroom management capabilities (strategies to support and motivate students).	Greece	formal (classroom)	97.3% 18 to 30	42.7% female, 57.3% male	Not reported	pre- service

### Table A.1: Context of use of SGs

	Description of the game-based teacher	Purpose of the		Setting where	e Characteristics of partici		s of particip	pants
Study	education intervention	educational intervention	Country	the game was accessible	Age	Gender	Subject area	Status
Lameras et al. (2014)	Role-playing SG "SimAULA" simulating real-world classroom environment. Game facing players/teachers with realistic situations, requiring them to interact with virtual students (AI agents). Training adopted an inquiry-based learning approach.	Enhance teachers' classroom management and lesson planning capabilities.	Europe	not reported	not reported	not reported	Science education	in- service and pre- service
Annetta et al. (2014)	Role-playing SG STIMULATE (Science Training Immersive Modules for University Learning Around Teacher Education) simulating science laboratory problem scenarios, used as a course for teacher preparation.	Advancing teachers' preparation and development in laboratory safety.	USA	formal (classroom)	not reported	35% female, 64% male	Science education	pre- service
Vrasidas and Solomou(2013)	Online role-playing gaming environment Quest Atlantis used to engage teachers in the evaluation and improvement of simulated classroom situations.	Promote the exploration of the potential of games for education, and reflection on practices from experience.	Cyprus	formal (classroom)	not reported	100% female	Not reported	in- service
Kennedy-Clark et al. (2011)	Teachers participated in a game-based inquiry learning process which included playing a role-playing SG and assessing other online SGs and virtual worlds.	Developing skills in using and evaluating game-based learning resources in secondary science education.	Austria	formal (classroom)	24.4 average	61% female, 39% male	Science education	pre- service

#### A.3.2 Key characteristics of SGs involved in the reviewed studies

Figure A.1 illustrates the key game features identified to address Q2.

Most SGs (n=6) adopted a 3rd-person view to show a "physical" representation of the player in the game space. Educational game contents involved in equal measure contents that trainee teachers would have to teach in the future (n=4), and educational methodologies/strategies/skillsets to enhance their practice (n=4). Studies were generally conducted using different approaches regarding the learning environment. Some directly used virtual game environments. For example, on Alyaz and Genc (2016) a detective story as a fantasy setting of learning was provided to teachers where people were kidnapped for a secret project and rescued. Others, real scenarios blended with SGs, which meant the settings of games simulated realistic approximations of the environment to support teachers' learning (n=7). Especially, these games tended to use simulated school as the setting (n=5). Most papers (n=5) reviewed reported respectively that players had to perform tasks simulating real-world educational activities and assumed an education-related role (e.g. teacher or student). Most (n=6) games required a PC to be run. Among these, three required internet connectivity. Only two games ran on mobile platforms. One of them (Aivelo and Uitto, 2016) adopted augmented-reality and location-based mechanics, thus aligning with current trends in mobile gaming.



Figure A.1 Key characteristics of SGs demonstrated effectiveness in mediating teacher education

#### A.3.3 Impacts of SGs

Table A.2 systematises the findings concerning impacts on teachers' professional development (Q3). Three key types of effects were identified: knowledge development; attitudes and others. The studies of Lameras et al. (2014) and Di Fuccio et al. (2019) did not report impacts. Hence, they were excluded from Table 2. Most studies reported quantitative measurements of impact. Vrasidas and Solomou (2013) reported qualitative outcomes. Only four studies reported knowledge development outcomes. However, mixed evidence was found regarding the impacts of SGs on teachers- knowledge development. Kennedy-Clark et al. (2011) showed that knowledge achievement did not change, via comparison of the pre-and post-tests within the game intervention. Alyaz and Genc (2016) and Annetta et al. (2014) observed significant effects on knowledge development through the intervention of SGs. From the perspective of attitudes, most studies reported positive impacts, such the perceiving SGs as useful tools for learning (Alyaz and Genc, 2016, Vrasidas and Solomou, 2013, Aivelo and Uitto, 2016), or positive shift of attitudes towards adopting technology in the classroom (Kennedy-Clark et al., 2011). Only one study reported other types of impacts (Kennedy-Clark et al., 2011).

## Table A.2: Impacts of SGs

Study	Types of Impact (measurement)					
	Knowledge development	Attitudes	Others			
Alyaz and Genc (2016)	<ol> <li>(1) The game expanded teachers' professional vocabulary (paired-samples t-test showed a pre-post increase in words, t (59) = 18.41, p&lt;0.01).</li> <li>(2) Learning achievement in the game is affected by age (ANOVA F-test, F=6.48, p&lt;0.05). (3) Gender, game preferences and language priming do not affect teachers' learning (ANOVA F-test, F=6.48, p&gt;0.05).</li> </ol>	Most teachers believed that digital games are useful for learning foreign languages (86.7%, by frequencies of survey responses).	Not reported			
Stavroulia et al. (2016)	not reported	<ol> <li>(1) Simulated classroom environment generates both positive and negative real-time emotions similar to those experienced in real-world settings.</li> <li>(2) Most frequently-felt positive emotions (by frequencies of survey responses in quartile four): interest (95.2%); excitement (90.4%), motivation (92.3%) and satisfaction (90.4%).</li> <li>(3) Most frequently negative emotions (by frequencies of survey responses in quartile four): fatigue (94.2%); inability to deal with situation (83.6%); nervousness (77.6%).</li> <li>(4) Feelings experienced were not affected by genre (x2= 2.652, df = 1, asymptotic p -value = 0.103).</li> </ol>	Not reported			
Aivelo and Uitto (2016)	not reported	(1) Teachers provided suggestions to enhance the game	Not reported			

Study	Types of Impact (measurement)					
	Knowledge development	Attitudes	Others			
		<ul> <li>(2) Key attitudes broadly supported by teachers</li> <li>(based on analysis of themes emerging from group interviews, and individual support to these):</li> <li>(2.1) Positive attitudes toward the game, including the outdoor setting and the educational game contents</li> <li>(2.2) Dissatisfaction with the technical issues</li> <li>(2.3) Need for guidance on how to play the game</li> </ul>				
Annetta et al. (2014)	(1) Self-perceived development of chemistry safety knowledge (pre-post questionnaire responses; Shapiro–Wilk W test, W= .907, p = .059, M = 5.06, SD = 1.84). (2) Self-perceived development of emergency response skills (pre- post questionnaire responses; Shapiro–Wilk W test: W= .94, p = .124, M = 5.81, SD = 1.93). (3) Gender affects self-perceived development of safety assessment skills (ANOVA F-Test, MFemales = 3.95, SDFemales = 1.46, MMales = 6.09, SDMales = 1.64, F(8, 30) = 11.54, p = .002, n2 = 0.27 (large)). (4) Factors not affecting self-perceived development of safety assessment skills: ethnicity (ANOVA F-Test, M = 4.68, SD = .94, F(4,30) = 0.356, p = 0.837); years in school (ANOVA F-Test, M = 1.29, SD = .46, F(1,30) = .048, p = .828); major (ANOVA F-Test, M = 1.54, .50, F(1,30) = .314, p = .581); and experience (ANOVA F-Test, M = 4.87, SD = 2.92, F(1,30) = 2.94, p = .101)	Most teachers endorsed mixed-mode video games integrating different game types (e.g. sport, first- person shooter) centred on social components, which similarly to what most SGs do (61.29%, by frequency of survey responses).	Not reported			

Study	Types of In	mpact (measurement)	
	Knowledge development	Attitudes	Others
Vrasidas and Solomon (2013)	<ul> <li>(1) Self-perceived development of knowledge on student motivation and strategies to enhance it.</li> <li>(2) Self-perceived improvement of readiness to deal with future motivation-related situations.</li> <li>(3) Self-perceived development of other knowledge and skills, such as navigation.</li> <li>(All the above stemming from qualitative content analysis of teachers' propositions)</li> </ul>	<ul> <li>(1) Identified learning affordances offered by online game environments: interaction, collaboration, authentic problem-solving, role-playing, immersion, learner-centred; and instructor support.</li> <li>(2) Identified learning hindrances generated by online environments: complexity in integrating game-acquired knowledge in the curriculum; understanding ability; time pressure; learning contents distributed across game space and other resources.</li> <li>(3) Knowledge acquisition enhanced by games, compared to other media (e.g. books, articles, web pages, etc.).</li> <li>(4) Appropriateness of games to promote situated learning and effective understanding of information. (All the above stemming from qualitative content analysis of teachers' propositions)</li> </ul>	Not reported
Kennedy- Clark et al. (2011)	<ol> <li>Participants were fairly consistent between the pre-and post- tests for the understanding of knowledge of serious games(e.g. defining a virtual world).</li> <li>Teachers could effectively choose and utilise games as a teaching tool in the classroom (pre-test 52.9%, post-test 83.3%, by frequency of survey responses).</li> </ol>	Teachers showed a positive shift towards using games in the classroom (pre-test 38.5%, post-test 86.7%, by frequency of survey responses)	There was no significant correlation between gender and frequency of playing game (the test: p = .021).

#### A.4 Discussion

#### A.4.1 Viable outcomes of serious game-based teacher education

The author proposes in Figure A.2 a model of teacher education outcomes that is viable to obtain through the use of SGs, based on the evidence that the author analysed and summarized in table 8.2.

This model can be useful to inform the planning and evaluation of teachers' education. For example, targeting specific impacts (e.g. developing specific attitudes, skills or knowledge), and involving teachers in user-centred design processes to anticipate likely impacts (e.g. identify possible issues and unforeseen learning benefits). By the same token, such a model could be used to plan the adoption of games in teacher education programmes. For example, by facilitating the identification of benefits that games could promote in a specific context, and then selecting and adopting games presenting features that could promote such benefits.

#### A.4.2 Trends and limitations emerging from the reviewed studies

The scarcity of empirical studies investigating the impacts of games on teachers' training through qualitative and quantitative methods is a key limitation that should be addressed. Based on this review, the author suggests that more empirical studies should be conducted to investigate the impacts of SGs in teacher training. Furthermore, there is a far larger number of empirical studies investigating the impacts of the use of SGs on other types of students according to these previous reviews (e.g. Hung et al., 2018, Perttula et al., 2017, Hwang et al.,

2012). The author proposed that this disparity should be pondered and addressed. After all, both teachers and students are key entities of pedagogical systems that require support and development.

Most studies (n=7) included in this review did not investigate associations between the effects of the use of SGs and the demographics of the participants. However, according to these findings, younger teacher trainees seem more likely to accept games as an effective tool to use. This is consistent with previous studies (Shaffer, 2006, Van Eck, 2006) which indicated that Serious Games and game-based learning may be more suitable to engage younger learners.



Figure A.2 A model of SGs-mediated teacher education outcomes

These findings suggest that future research may focus on the extent to which games should be leveraged to promote teachers' learning "anytime, anywhere". Most studies (n=6) implemented games within in a formal educational context, rather than promoting the voluntarily use of games outside classes. A good game might motivate people to use it - and consequently, learn through it - even if players are not "required" to do so (Crowston and Prestopnik, 2013). Hence, there may be interesting potentialities that might remain unexploited if SGs are not available outside formal training/educational sessions.

Furthermore, the platform of games can affect the learning experience (Klopfer and Squire, 2008, Squire et al., 2002). For example, a mobile game might be easier and more frequently used than a PC game. However, most projects (n=6) in this review reported using PC games. This should be further investigated in the future, exploring which platforms SGs should be designed in order to maximise the possibility of using and learning through them. This could be done, for example, involving teachers in participatory design processes. In addition, the author suggested that it would be good if researchers provided a rationale underpinning the choice of the technological platform, in relation to the characteristics of the target audience, the desired context of use, and the learning outcomes pursued.

#### A.4.3 Limitations of this review and trajectories for future studies

A limitation of this study is that it used Scopus as the only search database. Expanding searches to other databases, and/or adopting broader search criteria will be done in the future to further develop this review, hoping to increase the number of digital game-based interventions

reviewed. The author also plans to use more comprehensive strings in order to include interventions using non-digital games appropriately. For example, Petersen et al. (2011) used non-digital games for training pre-teachers to raise awareness of HIV/AIDS. Investigating these cases is crucial to achieve a more comprehensive analysis of broader analysis of game-mediated learning in the context of teacher education. After all, the notion of SGs was originally conceived for non-digital games (De Lope and Medina-Medina, 2017). Finally, the author plans to study associations between impacts, characteristics of the participants and game types/features, as this could more robustly inform the planning and evaluation of game-based teacher education, by allowing to identify these game features suitable to promote desirable effects in a specific target population.

#### A.5 Summary

This chapter is a work expended, the aim is to explore the viability of serious games as an effective tool for teacher training. For this, the author searched existing literature regarding serious games and teacher training between 2010 and 2020 in the Scopus database. Articles were included in the review if they reported any impacts on teachers, such as in their knowledge and skills, their motivations, perceptions and attitudes. Results revealed that: (1) most studies were conducted using different approaches regarding the learning environment (e.g. some use virtual game environments, others real scenarios blended with serious games) to support teachers' learning; (2) most studies report that the use of serious games could support teachers' preparation, especially the development of planning skills and strategies; (3) there are only a few studies that focus on teacher knowledge development; (4) research was mainly conducted in European countries; (5) most games are targeted to prospective teachers. In general, findings suggest that serious game can be good tools to promote teachers' professional development, but future research should explore which game characteristics are more suitable and effective for specific training aims and different teacher characteristics.

# Appendix B The questions of the five-cycle value

value creation	explanation	examples	questions
Immediate	The interaction and	Help each other; meetings;	1) What happened and experienced?
value	communication of	dialogues; advice; a story	2) What happened/were the important things?
(activities and	communities' members	be discussed; a project; a	3) How many people participate?
interactions)	can create value.	visit	4) What was the quality of mutual participation?
			5) Is it interesting/attractive?
			6) How were the activities related to me?
			7) Who did I contact with or have established with whom?
			8) Which relationships have the greatest impact on my own progress?
Potential value	Some values do not	Look at new perspectives	9) What happened to all these activities?
(knowledge	appear immediately. This	of the problem, new ways	10) How has my attendance changed me?
capital)	includes knowledge	to solve problems;	11) Was I getting new skills or knowledge?
	capital that is acquired	inspiration, care,	12) Was I aware of this knowledge or my opinion has changed?

## Table B.1: The value creation of the five-cycle (Wenger et al., 2011)

value creation	explanation	examples	questions
	and used when needed.	Confidence, status	13) Did you have more inspiration through your work?
	The form of knowledge	reputation, mutual	14) Was I confident in my practical ability?
	capital:	understanding, accompany,	15) How has my attendance changed my social relationships?
	human capital; social	learning challenges among	16) To new people what access I gain?
	capital; tangible capital;	members; access to	17) Do I know them enough that what they can do to my learning?
	intangible capital;	information, resources,	18) Do I believe they are enough to ask them for help?
	learning capital	tools; reputation of the	19) Am I not feeling so isolated?
		community; transformation	20) Did I gain a reputation from my engagement?
		of learning knowledge	21) What access to materials has my attendance given me?
			22) Do I have new instruments, means or processes?
			23) Can I have documents or sthisces of information that I do not have access to?
			24) What situation has the community gained?
			25) Is recognition of this expertise changed by the community?
			26) Did we get a new voice through group learning?
			27) How has my attendance transformed my idea of learning?
			28) Can I see learning opportunities I haven't seen before?
			29) Do I have the opportunity to convene a practice or online community to serve
			learning that I have never seen before?
Applied value changes in	The changes, or innovations of actions,	Use the knowledge learned in the community or the	30) How is it different from my practice/life/background?
practice)	practices, tools, methods,	resources provided to	31) Where do I use community/network products?
pructice)	system, are due to the	practice	32) Where do I use my acquired skills?
	application of intellectual	pruetiee	33) When do I use community/network connections with completing tasks?
	capital in different		34) Can I attract other people to pursue the cause that I care about?
	contexts.		35) When and how do I use documents or tools produced or accessed by the
			community?
			36) How to implement an idea? At what level - personality, unit, organization?

explanation	examples	questions
Members are not satisfied with applications, new	Achievements; comprehension; ability	37) Does it make any difference to my ability to achieve important functions for me or other stakeholders?
practices or tools and expect to practice new ideas or use community resources to improve performance	improvement	<ul> <li>38) What aspects of my behaviour influence my participation in the community/network?</li> <li>39) Am I saving time or implementing to gain some new?</li> <li>40) Am I more successful? How about it?</li> <li>41) What is the impact of the implementation of an idea?</li> <li>42) Depending for the implementation of an idea?</li> </ul>
		<ul><li>42) Does this affect some of the indicators used to evaluate activities?</li><li>43) Because of my attendance, what achievements can my team achieve in the community/network?</li></ul>
When social learning leads to a rethinking of the definition of success, reshaping will be realized	Definition of assessment criteria; suggestions for community development; self-reflection; new standards for learner evaluation	<ul> <li>44) If it has changed the understanding and definition of important things that I or other stakeholders have?</li> <li>45) Does the social learning process led to a reflection on important issues?</li> <li>46) If this has changed people understanding of important things?</li> <li>47) Does this mean that new standards and new indicators should be included in the assessment?</li> <li>48) How those people who have the right to decide standards of success are impacted by this new understanding?</li> <li>49) Can the new understanding change into institutional change?</li> <li>50) Due to this new understanding, is there a new framework or system developed or created?</li> </ul>
	Members are not satisfied with applications, new practices or tools and expect to practice new ideas or use community resources to improve performance When social learning leads to a rethinking of the definition of success,	Members are not satisfied with applications, new practices or tools and expect to practice new ideas or use community resources to improve performanceAchievements; comprehension; ability improvementWhen social learning leads to a rethinking of the definition of success, reshaping will be realizedDefinition of assessment criteria; suggestions for community development; self-reflection; new standards for learner

## **Appendix C Questionnaires**

## C.1 A questionnaire for testing the reliability of the model

### Table C.1: A form for testing the reliability of the model

Investigation introduction: This investigation aim was to exam game features influencing reflection to promote reframing value generated by the community of inquiry(COI). The investigation is based on a model of game features promoting reframing value. The model driven by value-making is based on integrating perspectives from reframing value theory and COI theory.

Criteria of reframing value and involved game features	Strongly Disagree	Disagree	Undecided	Agree	Strongly Agree
	1	2	3	4	5
Promoting reflection If players rethink the game content leads to a reflection on important issues or changed people understanding of important things:					
A real no-play character (NPC) image					
Game stories					
Multiple endings of the game					
The game design of particular style					
Tasks with emotional experience					
Game tools					
Generating new assessments for other people If players reframe the assessment standard for other people:					

Investigation introduction: This investigation aim was to exam game features influencing reflection to promote reframing value generated by the community of inquiry(COI). The investigation is based on a model of game features promoting reframing value. The model driven by value-making is based on integrating perspectives from reframing value theory and COI theory.

Criteria of reframing value and involved game features	Strongly Disagree	Disagree	Undecided	Agree	Strongly Agree
	1	2	3	4	5
Game avatars					
Suggestions for community development					
if players reshape the new expectation of the whole community development:					
Game stories					
Game avatars					
Game mode					
The game design of particular style					
Tasks for problem-solving					
Game tools					
The rank system of the game					
The change of evaluation					
If players change the process and pattern of evaluation via the game experience:					
The game design of particular style					
Tasks for problem-solving					

### C.2 The questionnaires for testing the validity of the model

 Table C.2: The questionnaires for the game experience test (Chinese Paladin: Sword and Fairy 3)

1.What is your status: teacher () or student () Multiple choice question 2. If the game covers the following game features: Multiple endings of the game (Y/N)Task with emotional experience (Y/N)A real no-play character (NPC) image (Y/N) Single choice question 3. I felt a sense of responsibility in controlling roles' destiny while working on the options of multiple- endings of the game. For example, I would consider the advantage and disadvantages of different game-endings then made a choice. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, \_\_\_\_\_ (0 point) 4. I felt that I could generate divergent ideas such as associating similar situations while working on tasks with emotional experience. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points)

D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)?\_\_\_\_\_\_ (0 point) 5. I felt that I could reflect an NPC's behaviours or personalities due to the NPC is closer to a real living person. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)?\_\_\_\_\_\_(0 point) 6. I felt that I could generate divergent ideas while exploring the game stories. For example, the game story would stimulate my curiosity to explore more details including the analysis of game plots, guessing the relationship of game characters, digging the hiding philosophic theory of life. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)? (0 point) 7. I thought that gave suggestions for improving the game story was interesting. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points)

## Table C.3: The questionnaires for the game experience test (*Limbo*)

1.What is your status: teacher () or student ()
Multiple choice question
2. If the game covers the following game features:
The game design of particular style (Y/N)
Game tools (Y/N)
Tasks for problem-solving (Y/N)
Single choice question
3. I found the game design of particular style (theme, details of design, music and voices) to be helpful in my rethinking such as an association of similar items. (
)
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)
E. Strongly agree (5 points)
F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)? (0 point)
4. I found the game tools to be helpful in my rethinking such as how similar items were applied. ( )
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)
E. Strongly agree (5 points)

F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the
change of evaluation and generating new assessments for other people)? (0 point)
5. I thought that gave suggestions for improving the game design of particular style was interesting, including revising theme, details of design, music and
voices. ( )
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)
E. Strongly agree (5 points)
F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the
change of evaluation and generating new assessments for other people)?(0 point)
6. I thought that gave suggestions for improving the game tools was interesting. For example, I suggested that enhance aid kits in order to win the battle easier. (
)
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)
E. Strongly agree (5 points)
F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the
change of evaluation and generating new assessments for other people)?(0 point)
7. I thought that gave suggestions for improving the tasks for question-solving was interesting. For example, I wanted to complete the task easier hence I gave
some suggestions to decrease the difficulty of tasks. ( )
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)

E. Strongly agree (5 points)
F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the
change of evaluation and generating new assessments for other people)? (0 point)
8. I felt that game tools allowed me to complete my task in several new strategies of matching. ( )
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)
E. Strongly agree (5 points)
F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the
change of evaluation and generating new assessments for other people)? (0 point)
9. I felt that tasks for question-solving allowed me to complete the task in several new ways. ( )
A. Strongly disagree (1 point)
B. Disagree (2 points)
C. Undecided (3 points)
D. Agree (4 points)
E. Strongly agree (5 points)
F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the
change of evaluation and generating new assessments for other people)? (0 point)

Table C.4: The questionnaires for the game experience test (World of Warcraft)

1.What is your status: teacher () or student () Multiple choice question 2. If the game covers the following game features: Game avatars (Y/N) The rank system of the game (Y/N)Game mode (Y/N)Single choice question 3. I thought that gave suggestions for improving the game avatars was interesting. For example, the skills and personalities of game avatars sometimes directly impacted the quality of the game entertainment and art, which inspired me to suggest developers changing or revising the skills or the avatar's personality. () A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)? (0 point) 4. I thought that the rank system of the game was able to support multiple players at the same time. Hence, the fairness of the rank system such as the pairing system would be suggested to optimize. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points)

F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)? (0 point) 5. I thought that gave suggestions for improving the game mode was interesting. For example, I would present advice such as adding a checking function for browsing members' details in order to know teammates better. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)?\_\_\_\_\_ (0 point) 6. I felt that game avatar allowed me to complete my task in several new strategies such as the skills matching could be appropriately utilised to enhance avatars' ability. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points) E. Strongly agree (5 points) F. Other (0 point). How could this feature be improved in order to support refreshing value (promoting reflection, suggestions for community development, the change of evaluation and generating new assessments for other people)? (0 point) 7. I believed that I was the game avatar I was controlling to show my emotions or behaviours. This would trigger my rethink for this avatar or inspire other players as enemy or teammate to assess for avatars' behaviours. ( ) A. Strongly disagree (1 point) B. Disagree (2 points) C. Undecided (3 points) D. Agree (4 points)